SECURITIES AND EXCHANGE COMMISSION

# **FORM 10-Q**

Quarterly report pursuant to sections 13 or 15(d)

Filing Date: 2004-04-30 | Period of Report: 2004-03-31 SEC Accession No. 0000950129-04-002615

(HTML Version on secdatabase.com)

## **FILER**

## INTEGRATED ELECTRICAL SERVICES INC

CIK:1048268| IRS No.: 760542208 | State of Incorp.:DE | Fiscal Year End: 0930 Type: 10-Q | Act: 34 | File No.: 001-13783 | Film No.: 04767815 SIC: 1731 Electrical work Mailing Address 1800 WEST LOOP SOUTH SUITE 500 HOUSTON TX 77027

Business Address 1800 WEST LOOP SOUTH SUITE 500 HOUSTON TX 77027 7138601500

#### UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-0

(Mark One)

(X) QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended March 31, 2004

OR

( ) TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from \_\_\_\_\_to \_\_\_\_.

Commission File No. 1-13783

INTEGRATED ELECTRICAL SERVICES, INC.

(Exact name of registrant as specified in its charter)

DELAWARE 76-0542208

(State or other jurisdiction of incorporation or organization) (I.R.S. Employer Identification No.)

1800 West Loop South Suite 500 Houston, Texas 77027-3233 (Address of principal executive offices) (zip code)

Registrant's telephone number, including area code: (713) 860-1500

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes [X] No []

The number of shares outstanding as of April 29, 2004 of the issuer's common stock was 36,151,909 and of the issuer's restricted voting common stock was 2,605,709.

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

INDEX

<TABLE> <CAPTION>

Page <S>  $\langle C \rangle$ PART I. FINANCIAL INFORMATION Ttem 1. Financial Statements Consolidated Balance Sheets as of September 30, 2003 and March 31, 2004..... 2 Consolidated Statements of Operations for the six months ended March 31, 2003 and 2004..... 3 Consolidated Statements of Operations for the three months ended March 31, 2003 and 2004..... 4 Consolidated Statement of Stockholders' Equity for the six months ended March 31, 2004..... 5 Consolidated Statements of Cash Flows for the six months ended March 31, 2003 and 2004..... 6 Consolidated Statements of Cash Flows for the three months ended March 31, 2003 and 2004..... 7 Condensed Notes to Consolidated Financial Statements..... 8 Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations..... 19 Item 3. Quantitative and Qualitative Disclosures about Market Risk..... 32 Item 4. Controls and Procedures..... 33

### PART II. OTHER INFORMATION

Item 4.	Submission of Matters to a Vote of Security Holders	34
Item 6.	Exhibits and Reports on Form 8-K	35
Signatures .		36

</TABLE>

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

#### CONSOLIDATED BALANCE SHEETS (IN THOUSANDS, EXCEPT SHARE INFORMATION)

<TABLE> <CAPTION>

<caption></caption>	September 30, 2003	March 31, 2004
<\$>	(Audited) <c></c>	(Unaudited) <c></c>
ASSETS		
CURRENT ASSETS:		
Cash and cash equivalentsAccounts receivable:	\$ 40,201	\$ 19,043
Trade, net of allowance of \$5,425 and \$3,764 respectively	245,618	227,343
Retainage	68,789	68,974
Related party Costs and estimated earnings in excess of billings on	67	36
uncompleted contracts	48,256	52,601
Inventories	20,473	23,817
Prepaid expenses and other current assets	23,319	26,526
Total current assets	446,723	418,340
PROPERTY AND EQUIPMENT, net	52,697	48,734
GOODWILL, net	197,884	197,884
OTHER NON-CURRENT ASSETS	28,870	31,530
Total assets	\$ 726,174	\$ 696,488
LIABILITIES AND STOCKHOLDERS' EQUITY		
CURRENT LIABILITIES:		
Current maturities of long-term debt	\$ 256	\$7,286
Accounts payable and accrued expenses Billings in excess of costs and estimated earnings on	138,143	121,748
uncompleted contracts	41,913	37,095
Total current liabilities	180,312	166,129
LONG-TERM BANK DEBT, net of current maturities	-	42,857
OTHER LONG-TERM DEBT, net of current maturities	195	110
SENIOR SUBORDINATED NOTES, net	247,927	173,244
OTHER NON-CURRENT LIABILITIES	30,183	33,081
Total liabilities	458,617	415,421
COMMITMENTS AND CONTINGENCIES		
STOCKHOLDERS' EQUITY:		
Preferred stock, \$.01 par value, 10,000,000 shares authorized,		
none issued and outstanding Common stock, \$.01 par value, 100,000,000 shares authorized,	-	-
38,439,984 shares issued Restricted voting common stock, \$.01 par value, 2,605,709 shares	385	385
issued, authorized and outstanding	26	26
Treasury stock, at cost, 2,725,793 and 2,299,076 shares, respectively	(16,361)	(14,562)
Unearned restricted stock	-	(1,660)
Additional paid-in capital	427,709	429,307
Retained deficit	(144,202)	(132,429)
Total stockholders' equity	267,557	281,067
Total liabilities and stockholders' equity	\$ 726,174	\$ 696,488

## Total liabilities and stockholders' equity .....

</TABLE>

The accompanying condensed notes to consolidated financial statements are an integral part of these financial statements.

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

## CONSOLIDATED STATEMENTS OF OPERATIONS (IN THOUSANDS, EXCEPT SHARE INFORMATION)

<TABLE> <CAPTION>

<capiion></capiion>	Six Months Ended March 31,					
	2003	2004				
	(Unaud					
<\$>	<c></c>	<c></c>				
Revenues Cost of services	\$ 691,712 591,251	\$ 703,692 605,196				
Gross profit	100,461	98,496				
Selling, general and administrative expenses	76,079	71,359				
Income from operations	24,382	27,137				
Other (income)/expense:						
Interest expense	12,799	13,015				
(Gain)/loss on sale of assets	(30)	147				
Other expense (income), net	(55)	4,865				
	12,714	18,027				
Income before income taxes	11,668	9,110				
Provision/(benefit) for income taxes	4,492	(2,663)				
Net income	\$ 7,176	\$ 11,773				
Basic earnings per share	\$ 0.18	\$ 0.31				
Diluted earnings per share	\$ 0.18	\$    0.30				
Observation the computation of country non above (1995)						
Shares used in the computation of earnings per share (Note 5): Basic	39,388,158	38,408,067				
Diluted	39,423,220	39,013,887				
	=================	==========				

</TABLE>

## The accompanying condensed notes to consolidated financial statements are an integral part of these financial statements.

3

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS (IN THOUSANDS, EXCEPT SHARE INFORMATION)

		Ended March 31,
	2003	2004
	(Unauc	dited)
<s></s>	<c></c>	<c></c>
Revenues	\$ 343,135	\$ 343,849
Cost of services	294,030	295,964
Gross profit	49,105	47,885
Selling, general and administrative expenses	37,460	35,081
Income from operations	11,645	12,804
Other (income)/expense:		
Interest expense	6,343	6 <b>,</b> 555

(Gain)/loss on sale of assets Other expense (income), net	(89) (86)			137 4,966
		6,168		11,658
Income before income taxes		5,477		1,146
Provision/(benefit) for income taxes		2,108		(4,399)
Net income		3,369	\$ ====	
Basic earnings per share	\$ =====	0.09		0.14
Diluted earnings per share	т	0.09	т	0.14
Shares used in the computation of earnings per share (Note 5):				
Basic		327,623		,544,198
Diluted		372,367	39	,242,461

</TABLE>

The accompanying condensed notes to consolidated financial statements are an integral part of these financial statements.

4

### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENT OF STOCKHOLDERS' EQUITY (IN THOUSANDS, EXCEPT SHARE INFORMATION)

<TABLE> <CAPTION>

	Common Stock			Restricted Voting Common Stock			Treasury Stock		
	Shares	Amount		Shares	Amount		Shares	Amount	
<s></s>	<c></c>	<c></c>		<c></c>	<c></c>		<c></c>	<c></c>	
BALANCE, September 30, 2003 Issuance of stock	38,439,984	\$	385	2,605,709	\$	26	(2,725,793)	\$ (16,361)	
(unaudited) Issuance of restricted	-		-	-		-	4,926	30	
stock (unaudited) Purchase of treasury	-		-	-		-	-	-	
stock (unaudited) Issuance of stock under employee stock	-		-	-		-	(549,200)	(4,340)	
purchase plan (unaudited) Exercise of stock							199,438	1,290	
options (unaudited) Non-cash compensation	-		-	-		_	771,553	4,819	
(unaudited)	-		-	-		-	-	-	
Net income (unaudited)	-		-	-		-			
BALANCE, March 31, 2004 (unaudited)	38,439,984 =======	\$ ====	385	2,605,709		26	(2,299,076)	\$ (14,562) ========	

<CAPTION>

	Unea: Restr: Sto		Additional Earnings Capital	Retained Stockholders' (Deficit)	Total Paid-In Equity
<\$>	<c></c>		<c></c>	<c></c>	<c></c>
BALANCE, September 30,					
2003	\$	-	\$427,709	\$ (144,202)	\$ 267,557
Issuance of stock					
(unaudited)		-	9	-	39
Issuance of restricted					
stock (unaudited)	(1	1,992)	1,992	-	-
Purchase of treasury					
stock (unaudited)		-	-	-	(4,340)
Issuance of stock under					
employee stock					
purchase plan					
(unaudited)		-	(638)	-	652

Exercise of stock options (unaudited) Non-cash compensation	-	235	-	5,054
(unaudited)	332	-	-	332
Net income (unaudited)	-	-	11,773	11,773
BALANCE, March 31,				
2004 (unaudited)	\$ (1,660)	\$ 429,307	\$ (132,429)	\$ 281,067

</TABLE>

The accompanying condensed notes to consolidated financial statements are an integral part of these financial statements.

5

### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (IN THOUSANDS)

<caption></caption>	Six Months En	
	2003	2004
	Unaud)	
	<c></c>	<c></c>
CASH FLOWS FROM OPERATING ACTIVITIES: Net income	\$ 7,176	11,773
Adjustments to reconcile net income to net cash provided by operating activities	\$ /,1/0	11,775
Provision for allowance for doubtful accounts	661	(8)
Deferred income taxes	-	(6,262)
Depreciation and amortization	7,341	6,732
Loss (gain) on sale of property and equipment	(30)	147
Non-cash compensation expense	-	332
Gain on divestiture Changes in operating assets and liabilities, net of acquisitions and	(26)	-
dispositions of businesses Accounts receivable	10,101	17,955
Inventories	2,075	(3,344)
Costs and estimated earnings in	_,	(-,,
excess of billings on uncompleted contracts	1,297	(4,345)
Prepaid expenses and other current assets	672	(1,675)
Other noncurrent assets	423	2,591
Accounts payable and accrued expensesBillings in excess of costs and estimated	(9,098)	(13,537)
earnings on uncompleted contracts	(5,982)	(4,818)
Other current liabilities	362	50
Other noncurrent liabilities	2,998	1,433
Net cash provided by operating activities	17,970	7,024
CASH FLOWS FROM INVESTING ACTIVITIES:		
Proceeds from sale of property and equipment	1,540	398
Purchases of property and equipment	(5,462)	(3,139)
Purchases of business, net of cash acquired	(2,723)	-
Sale of business	1,084	-
Investments in securities	(500)	(400)
Net cash used in investing activities	(6,061)	(3,141)
CASH FLOWS FROM FINANCING ACTIVITIES:		
Borrowings	27	50,040
Repayments of debt	(16,030)	(75,241)
Purchase of treasury stock Proceeds from exercise of stock options	(3,376)	(4,340) 5,054
Proceeds from exercise of stock	_	39
Proceeds from issuance of stock under employee stock purchase plan	821	652
Payments for debt issuance costs	-	(1,245)
-		
Net cash used in financing activities	(18,558)	(25,041)
NET INCREASE IN CASH AND CASH EQUIVALENTS	(6,649)	(21,158)
CASH AND CASH EQUIVALENTS, beginning of period	32,779	40,201
CASH AND CASH EQUIVALENTS, end of period	\$ 26,130 ======	19,043
SUPPLEMENTAL DISCLOSURE OF CASH FLOW INFORMATION:		
Cash paid for		
Interest	\$ 12,061	13,480

Income taxes .....

#### 6

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (IN THOUSANDS)

<TABLE> <CAPTION>

----

		···· · · · ,	
	2003	2004	
		dited)	
<s></s>	<c></c>	<c></c>	
CASH FLOWS FROM OPERATING ACTIVITIES:			
Net income	\$ 3,369	5,545	
Adjustments to reconcile net income to net cash			
provided by operating activities			
Provision for allowance for doubtful accounts	282	381	
Deferred income taxes	-	(4,832)	
Depreciation and amortization	3,691	3,279	
Loss (gain) on sale of property and equipment	(89)	137	
Non-cash compensation expense	-	249	
Changes in operating assets and liabilities, net of acquisitions and dispositions of businesses			
Accounts receivable	1,700	8,993	
Inventories	1,202	(1,170)	
Costs and estimated earnings in	1/202	(1/1/0)	
excess of billings on uncompleted contracts	1,192	(3,375)	
Prepaid expenses and other current assets	699	(1,798)	
Other noncurrent assets	(6)	1,918	
Accounts payable and accrued expenses	959	(5,207)	
Billings in excess of costs and estimated		(-, -,	
earnings on uncompleted contracts	119	(4,972)	
Other current liabilities	195	50	
Other noncurrent liabilities	1,386	1,399	
Net cash provided by operating activities	14,699	597	
CASH FLOWS FROM INVESTING ACTIVITIES:			
Proceeds from sale of property and equipment	484	175	
Purchases of property and equipment	(2,933)	(1,394)	
Purchases of business	(2,723)	-	
Investments in securities	(500)	-	
Net cash used in investing activities	(5,672)	(1,219)	
CASH FLOWS FROM FINANCING ACTIVITIES:	22	F0 000	
Borrowings	22	50,000	
Repayments of debt	(195) (2,607)	(75,102) (990)	
Purchase of treasury stock Proceeds from exercise of stock options	(2,007)	(990) 2,178	
Proceeds from issuance of stock	_	2,178	
Proceeds from issuance of stock under employee stock purchase plan	821	652	
Payments for debt issuance costs	-	(1,245)	
Regimento for acor robanice costo		(1/213)	
Net cash used in financing activities	(1,959)	(24,488)	
NET INCREASE IN CASH AND CASH EQUIVALENTS	7,068	(25,110)	
CASH AND CASH EQUIVALENTS, beginning of period	19,062	44,153	
CASH AND CASH EQUIVALENTS, end of period	\$ 26,130	19,043	
SUPPLEMENTAL DISCLOSURE OF CASH			
FLOW INFORMATION:			
Cash paid for			
Interest	\$ 11,784	13,270	
Income taxes	\$ -	326	

  |  ||  |  |  |
\$ -

700

Three Months Ended March 31,

The accompanying condensed notes to consolidated financial statements are an integral part of these financial statements.

## CONDENSED NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

#### 1. OVERVIEW

Integrated Electrical Services, Inc. (the "Company" or "IES"), a Delaware corporation, was founded in June 1997 to create a leading national provider of electrical services, focusing primarily on the commercial and industrial, residential, low voltage and service and maintenance markets.

The accompanying unaudited condensed historical financial statements of the Company have been prepared in accordance with accounting principles generally accepted in the United States and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and footnotes required for complete financial statements, and therefore should be reviewed in conjunction with the financial statements and related notes thereto contained in the Company's annual report for the year ended September 30, 2003, filed on Form 10-K with the Securities and Exchange Commission. In the opinion of management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Actual operating results for the six months ended March 31, 2004, are not necessarily indicative of the results that may be expected for the fiscal year ended September 30, 2004.

#### SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

For a description of these policies, refer to Note 2 of the Notes to Consolidated Financial Statements included in the Company's Annual Report on Form 10-K for the year ended September 30, 2003.

#### SUBSIDIARY GUARANTIES

All of the Company's operating income and cash flows are generated by its 100% owned subsidiaries, which are the subsidiary guarantors of the Company's outstanding 9 3/8% senior subordinated notes due 2009 (the "Senior Subordinated Notes"). The Company is structured as a holding company and substantially all of its assets and operations are held by its subsidiaries. There are currently no significant restrictions on the Company's ability to obtain funds from its subsidiaries by dividend or loan. The parent holding company's independent assets, revenues, income before taxes and operating cash flows are less than 3% of the consolidated total. The separate financial statements of the subsidiary guarantors are not included herein because (i) the subsidiary guarantors are all of the direct and indirect subsidiaries of the Company; (ii) the subsidiary guarantors have fully and unconditionally, jointly and severally guaranteed the Senior Subordinated Notes; and (iii) the aggregate assets, liabilities, earnings and equity of the subsidiary guarantors is substantially equivalent to the assets, liabilities, earnings and equity of the Company on a consolidated basis. As a result, the presentation of separate financial statements and other disclosures concerning the subsidiary guarantors is not deemed material.

8

#### USE OF ESTIMATES AND ASSUMPTIONS

The preparation of financial statements in conformity with accounting principles generally accepted in the United States requires the use of estimates and assumptions by management in determining the reported amounts of assets and liabilities, disclosures of contingent liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates. Estimates are primarily used in the Company's revenue recognition of construction in progress, fair value assumptions in analyzing goodwill impairment, allowance for doubtful accounts receivable, realizability of deferred tax assets and self-insured claims liability.

#### SEASONALITY AND QUARTERLY FLUCTUATIONS

The results of the Company's operations, particularly from residential construction, are seasonal, dependant upon weather trends, with typically higher revenues generated during the spring and summer and lower revenues during the fall and winter. The commercial and industrial aspect of its business is less subject to seasonal trends, as this work generally is performed inside structures protected from the weather. The Company's service business is generally not affected by seasonality. In addition, the construction industry has historically been highly cyclical. The Company's volume of business may be adversely affected by declines in construction projects resulting from adverse regional or national economic conditions. Quarterly results may also be materially affected by gross margins for both bid and negotiated projects, the timing of new construction projects and any acquisitions. Accordingly, operating results for any fiscal period are not necessarily indicative of results that may be achieved for any subsequent fiscal period.

#### NEW ACCOUNTING PRONOUNCEMENT

In January 2003, the Financial Accounting Standards Board issued Interpretation No. 46, "Consolidation of Variable Interest Entities," ("Interpretation 46"). The objective of Interpretation 46 is to improve the financial reporting by companies involved with variable interest entities. Until now, one company generally has included another entity in its consolidated financial statements only if it controlled the entity through voting interest. Interpretation  $4\, \rm 6$ changes that by requiring a variable interest entity to be consolidated by a company if that company is subject to a majority of the risk of loss from the variable interest entity's activities or entitled to receive a majority of the entity's residual returns or both. The consolidation requirements of Interpretation 46 apply immediately to variable interest entities created after January 31, 2003. The consolidation requirements apply to older entities in the first fiscal year or interim period ending after March 15, 2004. Certain disclosure requirements apply to all financial statements issued after January 31, 2003, regardless of when the variable interest entity was established. The Company has investments in two firms, EnerTech Capital Partners II, L.P. (EnerTech) and Energy Photovoltaics, Inc. (EPV) that might fall under this interpretation. The Company determined that EPV was excepted out of the provisions of Interpretation 46 and that it was not the primary beneficiary of EnerTech and as such, the adoption of this interpretation did not have a material impact on the Company's results of operations or its financial position.

9

#### STOCK BASED COMPENSATION

The Company accounts for its stock-based compensation arrangements using the intrinsic value method in accordance with the provisions of Accounting Principles Board Opinion No. 25, "Accounting for Stock Issued to Employees" ("APB 25") and related interpretations. Under APB 25, if the exercise price of employee stock options equals the market price of the underlying stock on the date of grant, no compensation expense is recognized. The Company's stock options have all been granted with exercise prices at fair value, therefore no compensation expense has been recognized under APB 25 (See Note 7). The following table illustrates the effect on net income and earnings per share assuming the compensation costs for IES' stock option and purchase plans had been determined using the fair value method at the grant dates amortized on a pro rata basis over the vesting period as required under SFAS No. 123, "Accounting for Stock-Based Compensation" for the three and six months ended March 31, 2003 and 2004 (in thousands, except for per share data):

<TABLE> <CAPTION>

		THREE MONTHS ENDED MARCH 31,							
	2003		2003 2004		2003			2004	
<\$>				>		>			
Net income, as reported Add: Stock-based employee compensation expense included in reported net income,	\$			5,545				11,773	
<pre>net of related tax effects Deduct: Total stock-based employee compensation expense determined under fair value based method for all awards,</pre>		-		151		-		201	
net of related tax effects		378		443		755		814	
Pro forma net income for SFAS									
No.123	\$	2,991	\$	5,253	\$	6,421	\$	11,160	
	===		==		==		==		
Earnings per share:									
Basic - as reported	\$	0.09	\$	0.14	\$	0.18	\$	0.31	
Basic - pro forma for SFAS No. 123	\$	0.08	\$	0.14	\$	0.16	\$	0.29	
Earnings per share:									
Diluted - as reported	\$	0.09	\$	0.14	\$	0.18	\$	0.30	
Diluted - pro forma for SFAS No. 123	\$	0.08	\$	0.13	\$	0.16	\$	0.29	

10

#### 2. BUSINESS COMBINATIONS

Acquisition

On February 27, 2003, the Company purchased the assets of Riviera Electric LLC,

an electrical contractor located in the state of Colorado, out of a bankruptcy auction of a prior competitor. The total consideration paid in this transaction was approximately \$2.7 million, comprised entirely of cash, net of cash acquired. The fair value of the tangible net assets acquired exceeded the total consideration paid. As a result, the long-term fixed assets of the acquisition were reduced to zero. The purchase price was allocated as follows (amounts in thousands):

<table></table>	
<\$>	<c></c>
Accounts receivable, net	\$ 11,643
Retention	3,884
Costs and estimated earnings in excess of billings	
on uncompleted projects and other	922
Less: Accounts payable and accrued expenses	(10,214)
Less: Billings in excess of costs and estimated	
earnings on uncompleted projects and other	(3,512)
Cash paid, net of cash acquired	\$ 2,723
	=======

#### </TABLE>

The unaudited pro forma data presented below reflect the results of operations of IES and the acquisition of Riviera Electric LLC assuming the transaction was completed on October 1, 2002 (in thousands):

#### <TABLE> <CAPTION>

	Six Months Ended March 31,	
		2003
<s></s>	<02	>
Revenues	\$	726,377
Net income	\$	8,060
Basic earnings per share	Ş	0.20
Diluted earnings per share	Ş	0.20

</TABLE>

The unaudited pro forma data summarized above also reflects pro forma adjustments primarily related to reductions in general and administrative expenses for contractually agreed reductions in compensation programs and additional income tax expense based on the Company's effective income tax rate. The unaudited pro forma financial data does not purport to represent what the Company's combined results of operations would actually have been if such transactions had in fact occurred on October 1, 2002, and are not necessarily representative of the Company's results of operations for any future period.

#### 11

#### Divestiture

On October 8, 2002, the Company sold all of the stock of one of its operating companies. The proceeds from the sale were \$1.1 million in cash and 70,330 shares of the Company's common stock. The Company recorded a pre-tax gain of less than \$0.1 million associated with this sale that is recorded in other income.

In connection with the disposition discussed above, the net pre-tax gain was determined as follows for the quarter ended December 31, 2002 (in thousands):

<TABLE>

<s></s>	<c></c>	
Book value of assets divested	\$	1,830
Liabilities divested		(502)
Net assets divested		1,328
Quality was a final d		1 004
Cash received		1,084
Common stock received		270
Total consideration received		1 254
Total consideration received		1,354
Due teu este	ŝ	2.6
Pre-tax gain	Ş	20
		;=======

<sup>&</sup>lt;/TABLE>

3. RESTRUCTURING CHARGES

In October 2001, the Company began implementation of a workforce reduction program. The purpose of this program was to cut costs by reducing the number of administrative staff both in the field and at the home office. The total number of terminated employees was approximately 450. As a result of the program implementation, the Company recorded pre-tax restructuring charges of \$5.6 million associated with 45 employees during the year ended September 30, 2002 and presented these charges as a separate component of the Company's results of operations for the period then ended. The charges were based on the costs of the workforce reduction program and include severance and other special termination benefits. The Company believes the reduction of these personnel resulted in annual savings of approximately \$4.1 million in salaries and benefits. During the six months ended March 31, 2004, the Company settled the remaining payments required under the restructuring agreements and reduced the restructuring accrual by \$0.4 million, which is included as a reduction in selling, general, and administrative expenses. As of March 31, 2004, there are no amounts accrued and unpaid as a result of the restructuring.

#### 4. DEBT

#### Credit Facility

On February 27, 2004, the Company amended and restated its \$150.0 million revolving credit facility to a \$125.0 million revolving credit facility and a \$50.0 million term loan led by Bank One, NA. The term loan is repayable in 15 quarterly installments of \$1.8 million beginning May 28, 2004, with a final installment equal to the then-unpaid principal balance of the term loan on February 27, 2008. The revolving credit facility matures February 27, 2008 (collectively, the "Credit Facility"). The proceeds from the term loan and our available cash were used to call

12

\$75.0 million of our long term bonds. Amounts borrowed under the Credit Facility bear interest at an annual rate equal to either (a) the Eurodollar rate plus 1.75 percent to 2.75 percent, as determined by the ratio of the Company's total funded debt to EBITDA (as defined in the Credit Facility) or (b) the higher of (i) the bank's prime rate or (ii) the sum of the Federal funds rate plus 0.50 percent plus an additional 0.00 percent to 0.50 percent, as determined by the ratio of the Company's total funded debt to EBITDA. Commitment fees of 0.375 percent to 0.50 percent are assessed on any unused borrowing capacity under the Credit Facility. The Company's direct and indirect subsidiaries guarantee the repayment of all amounts due under the facility, and the facility is secured by a first perfected security interest in all of the assets of the Company and those subsidiaries, including all of the outstanding shares of the capital stock of those subsidiaries. Among other restrictions, the financial covenants include a minimum net worth requirement, a maximum total consolidated funded debt to EBITDA ratio, a maximum senior consolidated debt to EBITDA ratio, a minimum fixed charge coverage ratio and a minimum asset coverage ratio. The Company was in compliance with the financial covenants of its Credit Facility at March 31, 2004. As of March 31, 2004, the Company had \$50.0 million outstanding under its Credit Facility, letters of credit outstanding under its Credit Facility of \$33.7 million, \$0.3 million of other borrowings and available borrowing capacity under its Credit Facility of \$91.3 million.

#### Senior Subordinated Notes

On January 25, 1999 and May 29, 2001, the Company completed offerings of \$150.0 million and \$125.0 million Senior Subordinated Notes, respectively. The Senior Subordinated Notes bear interest at 9 3/8% and mature on February 1, 2009. The Company pays interest on the Senior Subordinated Notes on February 1 and August 1 of each year. The Senior Subordinated Notes are unsecured obligations and are subordinated to all existing and future senior indebtedness. The Senior Subordinated Notes are guaranteed on a senior subordinated basis by all of the Company's subsidiaries. Under the terms of the Senior Subordinated Notes, the Company is required to comply with various affirmative and negative covenants including: (i) restrictions on additional indebtedness, and (ii) restrictions on liens, guarantees and dividends. During the year ended September 30, 2002, the Company retired approximately \$27.1 million of these Senior Subordinated Notes. During the three months ended March 31, 2004, the Company called approximately \$75.0 million of its Senior Subordinated Notes, paying a call premium of 4.688%, or \$3.5 million. This premium was recorded as a loss in other income and expense in accordance with SFAS No. 145, "Rescission of FASB Statements No. 4, 44 and 64, Amendment of FASB Statement No. 13, and Technical Corrections."

13

Debt consists of the following (in thousands):

	2003	2004
<\$>	<c></c>	 <c></c>
Secured credit facility with a group of lending institutions, due February 27, 2008, at a weighted average interest rate of 4.12% Senior Subordinated Notes, due February 1, 2009, bearing	\$ –	\$ 50,000
interest at 9.375% with an effective interest rate of 9.50%	137,885	62,885
interest at 9.375% with an effective interest rate of 10.00%	110,000	110,000
Other	451	253
Total debt Less - short-term debt and current maturities of long-term debt	248,336 (256)	223,138 (7,286)
Less - unamortized discount on Senior Subordinated Notes	(3,198)	(2,576)
Add - fair value of terminated interest rate hedge	3,240	2,935
Total long-term debt	\$ 248,122	\$ 216,211

#### </TABLE>

#### 5. EARNINGS PER SHARE

The following table reconciles the numerators and denominators of the basic and diluted earnings per share for the six months ended March 31, 2003 and 2004 (in thousands, except share information):

#### <TABLE> <CAPTION>

	Six Months Ended March 31,			
	2003		2004	
<s> Numerator:</s>	<c></c>	<(	c>	
Net income	\$ 7,176		11,773	
Denominator: Weighted average shares outstanding - basic Effect of dilutive stock options	39,388,3 35,0		38,408,067 605,820	
Weighted average shares outstanding - diluted	39,423,2		39,013,887 ======	
Earnings per share: Basic Diluted 				

  | .18 \$ .18 \$ | 0.31 0.30 |For the six months ended March 31, 2003 and 2004, stock options of 5.3 million and 2.2 million, respectively, were excluded from the computation of diluted earnings per share because the options exercise prices were greater than the average market price of the Company's common stock.

The following table reconciles the numerators and denominators of the basic and diluted earnings per share for the three months ended March 31, 2003 and 2004 (in thousands, except share information):

14

2003       2004 <s> <c>         Numerator:       \$ 3,369         Net income       \$ 3,369        </c></s>		Three Months Ended March 31,			:h 31,
Numerator:       \$ 3,369       \$ 5,545         Net income       \$ 3,369       \$ 5,545         Denominator:       \$ 39,327,623       38,544,198         Effect of dilutive stock options       \$ 44,744       698,263         Weighted average shares outstanding - diluted       \$ 39,372,367       \$ 39,242,461         Weighted average shares outstanding - diluted       \$ 0.09       \$ 0.14					
Net income       \$ 3,369       \$ 5,545         Denominator:       Weighted average shares outstanding - basic       39,327,623       38,544,198         Effect of dilutive stock options       44,744       698,263         Weighted average shares outstanding - diluted       39,372,367       39,242,461         Earnings per share:       \$ 0.09       \$ 0.14	<\$>	<c></c>		<c></c>	
Denominator:       Weighted average shares outstanding - basic       39,327,623       38,544,198         Effect of dilutive stock options       44,744       698,263         Weighted average shares outstanding - diluted       39,372,367       39,242,461         Earnings per share:       8 0.09       0.14	Numerator:				
Denominator:       39,327,623       38,544,198         Effect of dilutive stock options       44,744       698,263         Weighted average shares outstanding - diluted       39,372,367       39,242,461         Earnings per share:       80,009       0.14	Net income	\$	3,369	\$	5,545
Weighted average shares outstanding - basic       39,327,623       38,544,198         Effect of dilutive stock options       44,744       698,263         Weighted average shares outstanding - diluted       39,372,367       39,242,461         Earnings per share:       5       0.09       \$       0.14		====			
Effect of dilutive stock options       44,744       698,263         Weighted average shares outstanding - diluted       39,372,367       39,242,461         Earnings per share:       8       0.09       \$       0.14	Denominator:				
Weighted average shares outstanding - diluted       39,372,367       39,242,461         Earnings per share:       8       0.09       \$       0.14	Weighted average shares outstanding - basic	39,	327,623	38,	544,198
Earnings per share: Basic \$ 0.09 \$ 0.14	Effect of dilutive stock options		44,744		698,263
Earnings per share: Basic \$ 0.09 \$ 0.14	Weighted average shares outstanding - diluted	39,	372,367	39,	242,461
Basic \$ 0.09 \$ 0.14		====		====	
Basic \$ 0.09 \$ 0.14	Earnings per share:				
		\$	0.09	\$	0.14
Diluted S 0.09 S 0.14	Diluted	s	0.09	s	0.14

  |  |  |  |  |For the three months ended March 31, 2003 and 2004, stock options of 5.3 million and 2.2 million, respectively, were excluded from the computation of diluted earnings per share because the options exercise prices were greater than the average market price of the Company's common stock.

#### 6. OPERATING SEGMENTS

The Company follows SFAS No. 131, "Disclosures about Segments of an Enterprise and Related Information." Certain information is disclosed, per SFAS No. 131, based on the way management organizes financial information for making operating decisions and assessing performance.

The Company's reportable segments are strategic business units that offer products and services to two distinct customer groups. They are managed separately because each business requires different operating and marketing strategies. These segments, which contain different economic characteristics, are managed through geographically-based regions.

The accounting policies of the segments are the same as those described in the summary of significant accounting policies. The Company evaluates performance based on income from operations of the respective business units prior to home office expenses. Management allocates costs between segments for selling, general and administrative expenses, goodwill impairment, depreciation expense, capital expenditures and total assets.

15

Segment information for the six months ended March 31, 2003 and 2004 is as follows (in thousands):

<TABLE> <CAPTION>

	SIX HONING ENDED MARCH SI, 2005			
	COMMERCIAL/ INDUSTRIAL	RESIDENTIAL	OTHER	TOTAL
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$556 <b>,</b> 379	\$135,333	\$ <del>-</del>	\$691,712
Cost of services	485,129	106,122	-	591,251
Gross profit	71,250	29,211		100,461
Selling, general and administrative	50,122	16,859	9,098	76 <b>,</b> 079
Operating income	\$ 21,128	\$ 12,352	\$ (9,098)	\$ 24,382
Other data:				
Depreciation expense	\$ 5,896	\$ 503	\$ 942	\$ 7,341
Capital expenditures	3,253	360	1,849	5,462
Total assets	505,006	107,720	98,766	711,492

STX MONTHS ENDED MARCH 31, 2003

</TABLE>

<TABLE> <CAPTION>

(ALTION)	SIX MONTHS ENDED MARCH 31, 2004			
	COMMERCIAL/ INDUSTRIAL	RESIDENTIAL	OTHER	TOTAL
<s> Revenues</s>	<c> \$558,273</c>	<c> \$145,419</c>	<c> \$ -</c>	<c> \$703,692</c>
Cost of services	489,705	115,491	-	605,196
Gross profit	68,568	29,928	-	98,496
Selling, general and administrative	44,090	16,818	10,451	71,359
Operating income	\$ 24,478	\$ 13,110 ======	\$(10,451) ======	\$ 27,137 ======
Other data:				
Depreciation expense Capital expenditures	\$ 5,016 1,391	\$ 605 724	\$ 1,111 1,024	\$ 6,732 3,139
Total assets	503,900	102,777	89,811	696,488

Segment information for the three months ended March 31, 2003 and 2004 is as follows (in thousands):

	COMMERCIAL/ INDUSTRIAL	RESIDENTIAL	OTHER	TOTAL
<s> Revenues Cost of services</s>	<c> \$284,745 247,852</c>	<c> \$ 58,390 46,178</c>	<c> \$ - -</c>	<c> \$343,135 294,030</c>
Gross profit	36,893	12,212		49,105
Selling, general and administrative	24,864	8,160	4,436	37,460
Operating income	\$ 12,029	\$ 4,052	\$ (4,436)	\$ 11,645
Other data: Depreciation expense Capital expenditures Total assets 				

 \$ 2,976 1,973 505,006 | \$ 243 108 107,720 | \$ 472 852 98,766 | \$ 3,691 2,933 711,492 |\_\_\_\_\_

THREE MONTHS ENDED MARCH 31, 2003

16

<TABLE> <CAPTION>

	THREE MONTHS ENDED MARCH 31, 2004			
	COMMERCIAL/ INDUSTRIAL	RESIDENTIAL	OTHER	TOTAL
<s> Revenues Cost of services</s>	<c> \$270,078 237,134</c>	<c> \$ 73,771 58,830</c>	<c> \$ -</c>	<c> \$343,849 295,964</c>
Gross profit	32,944	14,941		47,885
Selling, general and administrative	21,371	8,486	5,224	35,081
Operating income	\$ 11,573	\$ 6,455 =======	\$ (5,224)	\$ 12,804
Other data: Depreciation expense Capital expenditures Total assets 				

 \$ 2,388 574 503,900 | \$ 327 422 102,777 | \$564 398 89,811 | \$ 3,279 1,394 696,488 |The Company does not have significant operations or long-lived assets in countries outside of the United States.

#### 7. 1999 INCENTIVE COMPENSATION PLAN

In November 1999, the Board of Directors adopted the 1999 Incentive Compensation Plan (the "1999 Plan"). The 1999 Plan authorizes the Compensation Committee of the Board of Directors or the Board of Directors to grant employees of the Company awards in the form of options, stock appreciation rights, restricted stock or other stock based awards. The Company has up to 5.5 million shares of common stock authorized for issuance under the 1999 Plan.

In December 2003, the Company granted a restricted stock award of 242,295 shares under its 1999 Plan to certain employees. This award will vest in equal installments on December 1, 2004 and 2005, provided the recipient is still employed by the Company. The market value of the stock on the date of grant for this award was \$2.0 million, which will be recognized as compensation expense over the related two year vesting period. During the six months ended March 31, 2004, the Company amortized \$0.3 million to expense in connection with this award.

#### 8. COMMITMENTS AND CONTINGENCIES

From time to time, the Company and its subsidiaries are made party to legal actions and claims, various governmental proceedings and private civil suits that arise in the ordinary course of business in the construction industry. Some of the legal proceedings include claims for punitive as well as compensatory damages. While the final outcome of these matters cannot be predicted with certainty, considering among other things the meritorious legal defense available, liabilities that have been recorded and applicable insurance, it is the opinion of the Company's management that none of these items will have a material adverse effect on the results of operations, financial position or liquidity of the Company. However, an unexpected adverse resolution of one or more of these items could have a material adverse effect on the results of operations in a particular quarter or fiscal year. The Company expenses routine legal costs related to such proceedings as incurred.

Some of the Company's customers require the Company to post letters of credit as a means of guaranteeing performance under its contracts and ensuring payment by the Company to subcontractors and vendors. If the customer has reasonable cause to effect payment under a letter of credit, the Company would be required to reimburse its creditor for the letter of credit. Depending on the circumstances surrounding a reimbursement to its creditor, the Company may have a charge to earnings in that period. To date, the Company has not had a situation where a customer has had reasonable cause to effect payment under a letter of credit. At March 31, 2004, \$2.1 million of the Company's outstanding letters of credit were to collateralize its customers.

17

Some of the underwriters of the Company's casualty insurance program require it to post letters of credit as collateral. This is common in the insurance industry. To date the Company has not had a situation where an underwriter has had reasonable cause to effect payment under a letter of credit. At March 31, 2004, \$31.6 million of the Company's outstanding letters of credit were to collateralize its insurance programs.

Many of the Company's customers require us to post performance and payment bonds issued by a surety. Those bonds guarantee the customer that the Company will perform under the terms of a contract and that it will pay its subcontractors and vendors. In the event that the Company fails to perform under a contract or pay subcontractors and vendors, the customer may demand the surety to pay or perform under the Company's bond. The Company's relationship with its sureties is such that it will indemnify the sureties for any expenses they incur in connection with any of the bonds they issue on the Company's behalf. To date, the Company has not incurred significant expenses to indemnify its sureties for expenses they incurred on the Company's behalf. As of March 31, 2004, the Company's cost to complete projects covered by surety bonds was approximately \$201.3 million.

The Company has committed to invest up to \$5.0 million in EnerTech Capital Partners II L.P. ("EnerTech"). EnerTech is a private equity firm specializing in investment opportunities emerging from the deregulation and resulting convergence of the energy, utility and telecommunications industries. Through March 31, 2004, the Company had invested \$3.1 million under its commitment to EnerTech.

At March 31, 2004, the Company had reserves of \$24.2 million recorded in other noncurrent liabilities for tax positions adopted that a taxing authority may view differently. The Company believes these reserves are adequate in the event the positions are not ultimately upheld. The timing of the payments of these reserves is not currently known and would be based on the outcome of a possible review by a taxing authority.

18

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

#### INTRODUCTION

The following should be read in conjunction with the response to Part I, Item 1 of this Report. Any capitalized terms used but not defined in this Item have the same meaning given to them in Part I, Item 1.

This report on Form 10-Q includes certain statements that may be deemed to be "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, Section 21E of the Securities Exchange Act of 1934, as amended. These statements are based on our expectations and involve risks and uncertainties that could cause our actual results to differ materially from those set forth in the statements. Such risks and uncertainties include, but are not limited to, the inherent uncertainties related to estimating future results, fluctuations in operating results because of downturns in levels of construction, incorrect estimates used in entering into fixed price contracts, difficulty in managing the operation and growth of existing and newly acquired businesses, the high level of competition in the construction industry and the effects of seasonality. The foregoing and other factors are discussed in our filings with the SEC, including our Annual Report on Form 10-K for the year ended September 30, 2003.

In response to the SEC's Release No. 33-8040, "Cautionary Advice Regarding Disclosure About Critical Accounting Policies," we have identified the accounting principles which we believe are most critical to our reported financial status by considering accounting policies that involve the most complex or subjective decisions or assessments. We identified our most critical accounting policies to be those related to revenue recognition, the assessment of goodwill impairment, our allowance for doubtful accounts receivable, the recording of our self-insurance liabilities and our estimation of the valuation

allowance for deferred tax assets. These accounting policies, as well as others, are described in the Note 2 of "Notes to Consolidated Financial Statements" of our Annual report on Form 10-K for the year ended September 30, 2003 and at relevant sections in this discussion and analysis.

We enter into contracts principally on the basis of competitive bids. We frequently negotiate the final terms and prices of those contracts with the customer. Although the terms of our contracts vary considerably, most are made on either a fixed price or unit price basis in which we agree to do the work for a fixed amount for the entire project (fixed price) or for units of work performed (unit price). We also perform services on a cost-plus or time and materials basis. We are generally able to achieve higher margins on fixed price and unit price than on cost-plus contracts. We currently generate, and expect to continue to generate, more than half of our revenues under fixed price contracts. Our most significant cost drivers are the cost of labor, the cost of materials and the cost of casualty and health insurance. These costs may vary from the costs we originally estimated. Variations from estimated contract costs along with other risks inherent in performing fixed price and unit price contracts may result in actual revenue and gross profits for a project differing from those we originally estimated and could result in losses on projects. Depending on the size of a particular project, variations from estimated project costs could have a significant

19

impact on our operating results for any fiscal quarter or year. We believe our exposure to losses on fixed price contracts is limited in aggregate by the high volume and relatively short duration of the fixed price contracts we undertake. Additionally, we derive a significant amount of our revenues from new construction and from the southern part of the United States. Downturns in new construction activity or in construction in the southern United States could affect our results.

We complete most projects within one year. We frequently provide service and maintenance work under open-ended, unit price master service agreements which are renewable annually. We recognize revenue on service and time and material work when services are performed. Work performed under a construction contract generally provides that the customers accept completion of progress to date and compensate us for services rendered measured in terms of units installed, hours expended or some other measure of progress. Revenues from construction contracts are recognized on the percentage-of-completion method in accordance with the American Institute of Certified Public Accountants Statement of Position 81-1 "Accounting for Performance of Construction-Type and Certain Production-Type Contracts." Percentage-of-completion for construction contracts is measured principally by the percentage of costs incurred and accrued to date for each contract to the estimated total costs for each contract at completion. We generally consider contracts to be substantially complete upon departure from the work site and acceptance by the customer. Contract costs include all direct material and labor costs and those indirect costs related to contract performance, such as indirect labor, supplies, tools, repairs and depreciation costs. Changes in job performance, job conditions, estimated contract costs and profitability and final contract settlements may result in revisions to costs and income, and the effects of these revisions are recognized in the period in which the revisions are determined. Provisions for total estimated losses on uncompleted contracts are made in the period in which such losses are determined.

We evaluate goodwill for potential impairment in accordance with Statement of Financial Accounting Standards (SFAS) No. 142, "Goodwill and Other Intangible Assets." Included in this evaluation are certain assumptions and estimates to determine the fair values of reporting units such as estimates of future cash flows, discount rates, as well as assumptions and estimates related to the valuation of other identified intangible assets. Changes in these assumptions and estimates or significant changes to the market value of our common stock could materially impact our results of operations or financial position.

We provide an allowance for doubtful accounts for unknown collection issues in addition to reserves for specific accounts receivable where collection is considered doubtful. Inherent in the assessment of the allowance for doubtful accounts are certain judgments and estimates including, among others, our customers' access to capital, our customers' willingness to pay, general economic conditions and the ongoing relationships with our customers.

We are insured for workers' compensation, automobile liability, general liability and employee-related health care claims, subject to large deductibles. Our general liability program provides coverage for bodily injury and property damage neither expected nor intended. Losses up to the deductible amounts are accrued based upon our estimates of the liability for claims incurred and an estimate of claims incurred but not reported. The accruals are derived from actuarial studies, known facts, historical trends and industry averages utilizing the assistance of an actuary to determine the best estimate of the ultimate expected loss. We believe such accruals to be adequate. However, insurance liabilities are difficult to assess and estimate due to unknown factors, including the severity of an injury, the determination of our liability in proportion to other parties, the number of incidents not reported and the effectiveness of our safety program. Therefore, if actual experience differs from than the assumptions used in the actuarial valuation, adjustments to the reserve may be required and would be recorded in the period that the experience becomes known.

We regularly evaluate valuation allowances established for deferred tax assets for which future realization is uncertain. We perform this evaluation at least annually at the end of each fiscal year at such time as events have occurred or are anticipated to occur that may change our most recent assessment. The estimation of required valuation allowances includes estimates of future taxable income. In assessing the realizability of deferred tax assets at March 31, 2004, we considered whether it was more likely than not that some portion or all of the deferred tax assets would not be realized. The ultimate realization of deferred tax assets is dependent upon the generation of future taxable income during the periods in which those temporary differences become deductible. We consider the scheduled reversal of deferred tax liabilities, projected future taxable income and tax planning strategies in making this assessment. If actual future taxable income differs from our estimates, our results could be affected.

RESULTS OF OPERATIONS FOR THE SIX MONTHS ENDED MARCH 31, 2003 COMPARED TO THE SIX MONTHS ENDED MARCH 31, 2004

The following table presents selected unaudited historical financial information for the six months ended March 31, 2003 and 2004.

<TABLE> <CAPTION>

CAPTION/

		Six Months	Ended March 31,	
	2003	%	2004	00
		(dollars	in millions)	
<s></s>	<c></c>	<c></c>	<c></c>	<c></c>
Revenues	\$691.7	100%	\$703.7	100%
Cost of services (including depreciation)	591.2	85%	605.2	86%
Gross profit	100.5	15%	98.5	14%
Selling, general & administrative expenses	76.1	11%	71.4	10%
Income from operations	24.4	4 %	27.1	4%
Interest and other expense, net	12.7	2%	18.0	3%
Income before income taxes	11.7	2%	9.1	1%
Provision for income taxes	4.5	1%	(2.7)	0%
Net income	\$ 7.2	1%	\$ 11.8	2%

</TABLE>

21

REVENUES

- -

<TABLE> <CAPTION>

	PERCENT OF TOTAL	REVENUES
	SIX MONTHS ENDER	) MARCH 31,
	2003	2004
<\$>	 <c></c>	 <c></c>
Commercial and Industrial	80%	79%
Residential	20%	21%
Total Company	100%	100%
	===	===

  |  |Total revenues increased \$12.0 million, or 1.7%, from \$691.7 million for the six months ended March 31, 2003, to \$703.7 million for the six months ended March 31, 2004. This increase in revenues is primarily the result of \$18.5 million of revenues included in the six months ended March 31, 2004 due to an acquisition in February 2003. Therefore, only one month of revenue was included in the six

months ended March 31, 2003. This increase was partially offset by weakness in a few select commercial markets and the reductions of communications revenues.

Commercial and industrial revenues increased \$1.9 million, or 0.3%, from \$556.4 million for the six months ended March 31, 2003, to \$558.3 million for the six months ended March 31, 2004. This increase in revenues is primarily the result of approximately \$18.5 million of commercial and industrial revenue due to the acquisition that was included in the results of operations for the six months ended March 31, 2004. This increase was of operations for the six months ended March 31, 2004 but was owned by us for one month during the six months ended March 31, 2003. This increase was offset by a decrease in commercial revenues of \$14.8 million, or 4.7%, primarily in the central region and a reduction in communications related revenues of \$7.5 million, or 16.5%, across the company due to decreased spending in these markets.

Residential revenues increased \$10.1 million, or 7.5%, from \$135.3 million for the six months ended March 31, 2003, to \$145.4 million for the six months ended March 31, 2004, primarily as a result of increased demand for new housing resulting from the continued low interest rate environment.

#### GROSS PROFIT

## <TABLE>

		9 PROFIT MARGINS 9 SEGMENT REVENUES
	SIX MONTHS	ENDED MARCH 31,
	2003	2004
<s></s>	 <c></c>	 <c></c>
Commercial and Industrial	13%	12%
Residential	22%	21%
Total Company	15%	14%
	===	===

#### </TABLE>

Gross profit decreased \$2.0 million, or 2.0%, from \$100.5 million for the six months ended March 31, 2003, to \$98.5 million for the six months ended March 31, 2004. Gross profit margin as a percentage of revenues decreased from 14.5% for the six months ended March 31, 2003, to 14.0% for the six months ended March 31, 2004. The decline in gross margin during the six months ended March 31, 2004 was primarily due to the significant increase in copper prices

22

which impacted IES' material costs. The fixed price nature of many of our contracts precluded us from passing these costs onto our customers. This accounted for 37 basis points impact to our gross margin.

Commercial and industrial gross profit decreased \$2.7 million, or 3.8%, from \$71.3 million for the six months ended March 31, 2003, to \$68.6 million for the six months ended March 31, 2004. Commercial and industrial gross profit margin as a percentage of revenues decreased from 12.8% for the six months ended March 31, 2003, to 12.3% for the six months ended March 31, 2004. The decrease in commercial and industrial gross profit was primarily due to the higher copper prices discussed above.

Residential gross profit increased \$0.7 million, or 2.4%, from \$29.2 million for the six months ended March 31, 2003, to \$29.9 million for the six months ended March 31, 2004. This increase results primarily from the increased revenues in the residential segment. Residential gross profit margin as a percentage of revenues decreased from 21.6% for the six months ended March 31, 2003, to 20.6% for the six months ended March 31, 2004. This decrease in gross profit margin as a percentage of revenues was primarily the result of increased costs incurred due to the increased price of copper wire during the six months ended March 31, 2004 and increased competition for available work.

#### SELLING, GENERAL AND ADMINISTRATIVE EXPENSES

Selling, general and administrative expenses decreased \$4.7 million, or 6.2%, from \$76.1 million for the six months ended March 31, 2003, to \$71.4 million for the six months ended March 31, 2004. Selling, general and administrative expenses as a percentage of revenues decreased 0.9% from 11.0% for the six months ended March 31, 2003 to 10.1% for the six months ended March 31, 2004. This decrease primarily results from an organizational restructuring that occurred during the last two years where we streamlined our administrative cost structure. The decrease also reflects a \$0.4 million release of accrued restructuring charges for one individual that settled during the six months ended March 31, 2004 and a \$0.8 million decrease in bad debt expense during the six months ended March 31, 2004. Lastly, health care costs were approximately

\$0.7 million lower in the six months ended March 31, 2004 because of plan design changes that took effect January 1, 2004 and reduced lives participating in the medical plan.

#### INCOME FROM OPERATIONS

Income from operations increased \$2.7 million, or 11.1%, from \$24.4 million for the six months ended March 31, 2003, to  $\$27.1\ {\rm million}$  for the six months ended March 31, 2004. This increase in income from operations was primarily attributed to a \$12.0 million increase in revenues earned year over year, a \$4.7 million decrease in selling, general and administrative expenses, offset by a 2.0% decrease in gross profit earned on those revenues primarily as a result of increased copper prices.

2.3

#### NET INTEREST AND OTHER EXPENSE

Interest and other expense, net, increased \$5.3 million, from \$12.7 million for the six months ended March 31, 2003, to \$18.0 million for the six months ended March 31, 2004. This increase in net interest and other expense was primarily the result of a \$5.2 million loss associated with retiring \$75 million of senior subordinated notes during the six months ended March 31, 2004 and additional interest expense related to the \$50 million term loan issued on February 27, 2004.

#### PROVISION FOR INCOME TAXES

Our effective tax rate decreased from 38.5% for the six months ended March 31, 2003 to negative 29.2% for the six months ended March 31, 2004. This decrease is attributable to the release of \$6.3 million of deferred tax valuation allowances during the six months ended March 31, 2004. We released these valuation allowances because we believe that we will now realize the deferred tax assets for which they were established. Without the impact of these valuation allowance releases, our effective tax rate was 39.5% for the six months ended March 31, 2004. This increase in our effective tax rate from 38.5% for the six months ended March 31, 2003 to 39.5% for the six months ended March 31, 2004 is attributable to a change in the mix of states where taxable income is generated, an increase in state net operating loss valuation allowances and additional reserves for potential tax liability related to tax return filing positions taken by the Company.

RESULTS OF OPERATIONS FOR THE THREE MONTHS ENDED MARCH 31, 2003 COMPARED TO THE THREE MONTHS ENDED MARCH 31, 2004

The following table presents selected unaudited historical financial information for the three months ended March 31, 2003 and 2004.

<TABLE> <CAPTION>

<s> Revenues Cost of services (including depreciation)</s>
Gross profit Selling, general & administrative expenses
Income from operations Interest and other expense, net
Income before income taxes Provision for income taxes
Net income

2003	8	2004	8
	(dollars in	n millions)	
<c></c>	<c></c>	<c></c>	<c></c>
\$343.1	100%	\$343.9	100%
294.0	86%	296.0	86%
49.1	14%	47.9	 14%
37.5	11%	35.1	10%
11.6	3%	12.8	 4%
6.1	2%	11.7	3%
5.5	2%	1.1	 0%
2.1	1%	(4.4)	(1%)
\$ 3.4	1%	\$ 5.5	2%

Three Months Ended March 31,

\_\_\_\_\_

/TABLE>

24

REVENUES

<TABLE> <CAPTION>

> PERCENT OF TOTAL REVENUES THREE MONTHS ENDED MARCH 31,

	2003	2004
<s></s>	<c></c>	<c></c>
Commercial and Industrial	83%	79%
Residential	17%	21%
Total Company	100%	100%
	====	====

#### </TABLE>

Total revenues increased \$0.8 million, or 0.2%, from \$343.1 million for the three months ended March 31, 2003, to \$343.9 million for the three months ended March 31, 2004. This increase in revenues is primarily the result of \$6.1 million of revenues due to an acquisition that was included in the results of operations for the three months ended March 31, 2004 but was owned by us for one month during the three months ended March 31, 2003. This increase was offset by weaknesses in a few select commercial markets and reductions in communications revenues.

Commercial and industrial revenues decreased \$14.6 million, or 5.1%, from \$284.7 million for the three months ended March 31, 2003, to \$270.1 million for the three months ended March 31, 2004. The acquisition revenues discussed above were more than offset by a decrease in commercial revenues of \$24.5 million, or 14.4%, primarily in the central region and reductions of communications revenues of \$2.1 million, or 9.6% across the company due to decreased spending in these markets.

Residential revenues increased \$15.4 million, or 26.4%, from \$58.4 million for the three months ended March 31, 2003, to \$73.8 million for the three months ended March 31, 2004, primarily as a result of increased demand for housing due to the low interest rate environment.

GROSS PROFIT

<TABLE> <CAPTION>

	SEGMENT GROSS P AS A PERCENT OF S	
	THREE MONTHS EN	
	2003	2004
<s></s>	<c></c>	<c></c>
Commercial and Industrial	13%	12%
Residential	21%	20%
Total Company	14%	14%
	===	===

#### </TABLE>

Gross profit decreased \$1.2 million, or 2.4%, from \$49.1 million for the three months ended March 31, 2003, to \$47.9 million for the three months ended March 31, 2004. Gross profit margin as a percentage of revenues decreased from 14.3% to 13.9% for the three months ended March 31, 2003 compared to three months ended March 31, 2004. The decline in gross margin during the three months ended March 31, 2004 was primarily due to the significant increase in copper prices which impacted IES' material costs. The fixed price nature of many of our contracts precluded us from passing these costs onto our customers. This accounted for 69 basis points impact to our gross margin.

25

Commercial and industrial gross profit decreased \$4.0 million, or 10.8%, from \$36.9 million for the three months ended March 31, 2003, to \$32.9 million for the three months ended March 31, 2004. Commercial and industrial gross profit margin as a percentage of revenues decreased from 13.0% for the three months ended March 31, 2003, to 12.2% for the three months ended March 31, 2004. This decrease in gross profit margin as a percentage of revenues was primarily the result of the dramatic increase in copper prices discussed above.

Residential gross profit increased \$2.7 million, or 22.1%, from \$12.2 million for the three months ended March 31, 2003, to \$14.9 million for the three months ended March 31, 2004. Residential gross profit margin as a percentage of revenues decreased from 20.9% for the three months ended March 31, 2003, to 20.3% for the three months ended March 31, 2004. This decrease in gross profit margin as a percentage of revenues was primarily the result of the increase in copper prices during the three months ended March 31, 2004 and increased competition for work.

#### SELLING, GENERAL AND ADMINISTRATIVE EXPENSES

Selling, general and administrative expenses decreased \$2.4 million, or 6.4%, from \$37.5 million for the three months ended March 31, 2003, to \$35.1 million for the three months ended March 31, 2004. Selling, general and administrative expenses as a percentage of revenues decreased 0.7% from 10.9% for the three months ended March 31, 2003 to 10.2% for the three months ended March 31, 2004. This decrease primarily results from an organizational restructuring that occurred during the last two years where we streamlined our administrative cost structure. The decrease also includes a \$0.2 million decrease in bad debt expense during the three months ended March 31, 2004. Health care costs were approximately \$0.7 million lower in the three months ended March 31, 2004 and reduced lives participating in the medical plan.

#### INCOME FROM OPERATIONS

Income from operations increased \$1.2 million, or 10.3%, from \$11.6 million for the three months ended March 31, 2003, to \$12.8 million for the three months ended March 31, 2004. This increase in income from operations was attributed to a \$2.4 million decrease in selling, general and administrative expenses year over year offset by an increase in cost of goods sold resulting from the increase in copper prices during the three months ended March 31, 2004.

#### NET INTEREST AND OTHER EXPENSE

Interest and other expense, net increased from \$6.1 million for the three months ended March 31, 2003, to \$11.7 million for the three months ended March 31, 2004. This increase in net interest and other expense was primarily the result of a \$5.2 million loss associated with retiring \$75 million of senior subordinated notes on March 31, 2004 and additional interest expense related to the \$50 million term loan issued on February 27, 2004.

26

#### PROVISION FOR INCOME TAXES

Our effective tax rate decreased from 38.5% for the three months ended March 31, 2003 to negative 383.9% for the three months ended March 31, 2004. This decrease is attributable to the release of \$4.8 million of deferred tax valuation allowances during the three months ended March 31, 2004. We released these valuation allowances because we believe that we will now realize the deferred tax assets for which they were established. Without the impact of these valuation allowance releases, our effective tax rate was 37.8% for the three months ended March 31, 2003 to 37.8% for the three months ended March 31, 2004 is attributable to a change in the projected tax differences for the year ended September 30, 2004, a change in the mix of states where taxable income is generated, an increase in state net operating loss valuation allowances and additional reserves for potential tax liability related to tax return filing positions taken by the Company.

WORKING CAPITAL

	SEPTEMBER 30, 2003	MARCH 31, 2004
<s></s>	<c></c>	<c></c>
CURRENT ASSETS:		
Cash and cash equivalents Accounts receivable:	\$ 40,201	\$ 19,043
Trade, net of allowance of \$5,425 and \$3,764 respectively	245,618	227,343
Retainage	68,789	68,974
Related party	67	36
Costs and estimated earnings in excess of billings on		
uncompleted contracts	48,256	52,601
Inventories	20,473	23,817
Prepaid expenses and other current assets	23,319	26,526
Total current assets	\$446,723	\$418,340
CURRENT LIABILITIES:	<b>A AFC</b>	7 006
Current maturities of long-term debt	\$ 256	7,286
Accounts payable and accrued expensesBillings in excess of costs and estimated earnings on	138,143	121,748
uncompleted contracts	41,913	37,095
Total current liabilities	\$180,312	\$166,129

Working ca	apital					•
------------	--------	--	--	--	--	---

</TABLE>

Total current assets decreased \$28.4 million, or 6.4%, from \$446.7 million as of September 30, 2003 to \$418.3 million as of March 31, 2004. This decrease is primarily the result of a \$18.3 million decrease in trade accounts receivable, net, due to a company-wide focus on collections and the timing of billings on projects in progress. Working capital was decreased by a \$21.2 million decrease in cash and cash equivalents that was impacted by \$50.0 million in term loan borrowings offset by cash used to retire \$75.0 million of senior subordinated debt included in

27

cash used in financing activities. Additionally, \$3.1 million of cash was used in investing activities. See "Liquidity and Capital Resources" below for further information. These decreases in current assets were offset by an increase in costs and estimated earnings in excess of billings on completed contracts of \$4.3 million due to a timing of billings and a \$1.7 million increase in prepaid and other current assets.

Total current liabilities decreased \$14.2 million, or 7.9%, from \$180.3 million as of September 30, 2003 to \$166.1 million as of March 31, 2004. This decrease is primarily the result of a \$16.4 million decrease in accounts payable and accrued expenses due to the timing of payments made and a \$4.8 million decrease in billings in excess of costs and estimated earnings on uncompleted contracts, offset by a \$7.0 million increase in current maturities of long-term debt.

LIQUIDITY AND CAPITAL RESOURCES

As of March 31, 2004, we had cash and cash equivalents of \$19.0 million, working capital of \$252.2 million, outstanding borrowings of \$50.0 million under our credit facility, \$33.7 million of letters of credit outstanding, other debt of \$0.3 million and available capacity under our credit facility of \$91.3 million. The amount outstanding under our senior subordinated notes was \$172.9 million. All debt obligations are on our balance sheet.

During the six months ended March 31, 2004, we generated \$7.0 million of net cash from operating activities. This net cash provided by operating activities was comprised of net income of \$11.8 million, increased by \$6.7 million of non-cash charges related primarily to depreciation expense, offset by the reversal of \$6.3 million in tax valuation allowances, and further increased by changes in working capital. Working capital changes consisted of an \$18.0 million decrease in trade accounts receivable, net, due to the timing of collections. Working capital changes also included a \$3.3 million increase in inventory and a \$13.5 million decrease in accounts payable and accrued expenses as a result of the timing of payments made, with the balance of the change due to other working capital changes. Net cash used in investing activities was \$3.1 million, consisting primarily of \$3.1 million used for capital expenditures and \$0.4 million invested in securities, offset by \$0.4 million in proceeds from the sale of fixed assets. Net cash used in financing activities was \$25.0 million, resulting primarily from repayments of debt of \$75.2 million and \$4.3 million used in the acquisition of treasury stock, offset by borrowings of \$50.0 million and \$5.1 million received from the exercise of stock options.

On February 27, 2004, we amended and restated our \$150.0 million revolving credit facility to a \$125.0 million revolving credit facility and a \$50.0 million term loan led by Bank One, NA. The term loan is repayable in 15 quarterly installments of \$1.8 million beginning May 28, 2004, with a final installment equal to the then-unpaid principal balance of the term loan on February 27, 2008. The revolving credit facility are collectively referred to as credit facility. The proceeds from the term loan and our available cash were used to call \$75.0 million of our long term bonds. Amounts borrowed under the credit facility bear interest at an annual rate equal to either (a) the Eurodollar rate plus 1.75 percent to 2.75 percent, as determined by the ratio of the higher of (i) the bank's prime rate or (ii) the sum of the Federal funds rate plus 0.50 percent plus an additional 0.00 percent to 0.50 percent, as

28

determined by the ratio of the our total funded debt to EBITDA. Commitment fees of 0.375 percent to 0.50 percent are assessed on any unused borrowing capacity under the credit facility. Our direct and indirect subsidiaries guarantee the repayment of all amounts due under the facility, and the facility is secured by a first perfected security interest in all of the assets of the company and those subsidiaries, including all of the outstanding shares of the capital stock of those subsidiaries. Among other restrictions, the financial covenants include a minimum net worth requirement, a maximum total consolidated funded debt to EBITDA ratio, a maximum senior consolidated debt to EBITDA ratio, a minimum fixed charge coverage ratio and a minimum asset coverage ratio. We were in compliance with the financial covenants of our credit facility at March 31, 2004. As of March 31, 2004, we had \$50.0 million outstanding under our credit facility, letters of credit outstanding under our credit facility of \$33.7 million, \$0.3 million of other borrowings and available borrowing capacity under our credit facility of \$91.3 million.

On January 25, 1999 and May 29, 2001, we completed our offerings of \$150.0 million and \$125.0 million senior subordinated notes, respectively. The notes bear interest at 9 3/8% and will mature on February 1, 2009. We pay interest on the notes on February 1 and August 1 of each year. The notes are unsecured senior subordinated obligations and are subordinated to all of our existing and future senior indebtedness. The notes are guaranteed on a senior subordinated basis by all of our subsidiaries. Under the terms of the notes, we are required to comply with various affirmative and negative covenants including (1) restrictions on additional indebtedness, and (2) restrictions on liens, guarantees and dividends. During the year ended September 30, 2002, we retired approximately \$27.1 million of these senior subordinated notes. During the three months ended March 31, 2004, we called approximately \$75.0 million of our senior subordinated notes, paying a call premium of 4.688%, or \$3.5 million. This premium was recorded as a loss in other income and expense in accordance with SFAS No. 145, "Rescission of FASB Statements No. 4, 44 and 64, Amendment of FASB Statement No. 13, and Technical Corrections."

We utilized approximately \$2.7 million cash, net of cash acquired, to purchase Riviera Electric LLC in Denver, Colorado on February 27, 2003.

All of our operating income and cash flows are generated by our 100% owned subsidiaries, which are the subsidiary guarantors of our outstanding senior subordinated notes. We are structured as a holding company and substantially all of our assets and operations are held by our subsidiaries. There are currently no significant restrictions on our ability to obtain funds from our subsidiaries by dividend or loan. Our parent holding company's independent assets, revenues, income before taxes and operating cash flows are less than 3% of the consolidated total. The separate financial statements of the subsidiary guarantors are not included herein because (i) the subsidiary guarantors are all of the direct and indirect subsidiaries of the company; (ii) the subsidiary guarantors have fully and unconditionally, jointly and severally guaranteed the senior subordinated notes; and (iii) the aggregate assets, liabilities, earnings and equity of the subsidiary guarantors is substantially equivalent to the assets, liabilities, earnings and equity of the company on a consolidated basis. As a result, the presentation of separate financial statements and other disclosures concerning the subsidiary guarantors is not deemed material.

29

Other Commitments. As is common in our industry, we have entered into certain off balance sheet arrangements that expose us to increased risk. Our significant off balance sheet transactions include liabilities associated with noncancelable operating leases, letter of credit obligations and surety guarantees.

We enter into noncancelable operating leases for many of our vehicle and equipment needs. These leases allow us to retain our cash when we do not own the vehicles or equipment and we pay a monthly lease rental fee. At the end of the lease, we have no further obligation to the lessor. We may determine to cancel or terminate a lease before the end of its term. Typically, we are liable to the lessor for various lease cancellation or termination costs and the difference between the then fair market value of the leased asset and the implied book value of the leased asset as calculated in accordance with the lease agreement.

Some of our customers require us to post letters of credit as a means of guaranteeing performance under our contracts and ensuring payment by us to subcontractors and vendors. If our customer has reasonable cause to effect payment under a letter of credit, we would be required to reimburse our creditor for the letter of credit. Depending on the circumstances surrounding a reimbursement to our creditor, we may have a charge to earnings in that period. To date, we have not had a situation where a customer has had reasonable cause to effect payment under a letter of credit. At March 31, 2004, \$2.1 million of our outstanding letters of credit were to collateralize our customers.

Some of the underwriters of our casualty insurance programs require us to post letters of credit as collateral. This is common in the insurance industry. To date we have not had a situation where an underwriter has had reasonable cause to effect payment under a letter of credit. At March 31, 2004, \$31.6 million of our outstanding letters of credit were to collateralize our insurance programs.

Many of our customers require us to post performance and payment bonds issued by a surety. Those bonds guarantee the customer that we will perform under the terms of a contract and that we will pay subcontractors and vendors. In the event that we fail to perform under a contract or pay subcontractors and vendors, the customer may demand the surety to pay or perform under our bond. Our relationship with our surety is such that we will indemnify the surety for any expenses it incurs in connection with any of the bonds it issues on our behalf. To date, we have not incurred significant expenses to indemnify our surety for expenses it incurred on our behalf. As of March 31, 2004, our cost to complete projects covered by surety bonds was approximately \$201.3 million.

We have committed to invest up to \$5.0 million in EnerTech Capital Partners II, L.P. ("EnerTech"). EnerTech is a private equity firm specializing in investment opportunities emerging from the deregulation and resulting convergence of the energy, utility and telecommunications industries. Through March 31, 2004, we had invested \$3.1 million under our commitment to EnerTech.

30

Our future contractual obligations include (in thousands):

<TABLE> <CAPTION>

		THAN YEAR		2005	2	2006	2	2007		2008	TH 	IEREAFTER	T 	OTAL
<s></s>	<c></c>		<c< th=""><th>&gt;</th><th colspan="2"><c></c></th><th colspan="2"><c></c></th><th colspan="2"><c></c></th><th colspan="2"><c></c></th><th colspan="2"><c></c></th></c<>	>	<c></c>		<c></c>		<c></c>		<c></c>		<c></c>	
Debt and capital lease obligations(1)	\$	3,655	\$	7,270	\$	7,163	\$	7,151	\$	25,014	\$	172,885	\$	223,138
Operating lease obligations	\$	5,202	\$	11,961	\$	8,004	\$	5,510	\$	3,722	\$	2,529	\$	36,928

(1) The tabular amounts exclude the interest obligations that will be created if the debt and capital lease obligations are outstanding for the periods presented.

Our other commercial commitments expire as follows (in thousands):

<TABLE>

<CAPTION>

		THAN YEAR	2005		20	06	20	07	2008		THEREAFTER		TOTAL	
<\$>	<c></c>		<c></c>		<c></c>		<c></c>		<c></c>		<c></c>		<c></c>	
Standby letters of credit	\$3	3,709	\$		\$		\$		\$		\$		\$ 33,709	
Other commercial commitments	\$		\$		\$		\$		\$		\$	1,900(2)	\$ 1,900	

  |  |  |  |  |  |  |  |  |  |  |  |  |(2) Balance of investment commitment in EnerTech.

Outlook. The following statements are based on current expectations. These statements are forward-looking and actual results may differ materially. Economic conditions across the country are challenging. We continue to focus on collecting receivables and reducing days sales outstanding. To improve our position for continued success, we continue to take steps to reduce costs. We have made significant cuts in administrative overhead at the home office and in the field. Although we have seen signs of improvement in six months ended March 31, 2004, the economic outlook for the remainder of fiscal 2004 is still somewhat uncertain. We expect earnings per share in the third quarter of fiscal 2004 to range between \$0.10 and \$0.17 per share. For the year ended September 30, 2004, we expect earnings to range between \$0.55 and \$0.75 per share.

We expect to generate cash flow from operations. Our cash flows from operations tend to track with the seasonality of our business and historically have improved in the latter part of our fiscal year. We anticipate that our cash flow from operations will provide sufficient cash to enable us to meet our working capital needs, debt service requirements and capital expenditures for property and equipment through the next twelve months. We expect capital expenditures of approximately \$12 million for the fiscal year ended September 30, 2004. Our ability to generate cash flow from operations is dependent on many factors, including demand for our products and services, the availability of work at margins acceptable to us and the ultimate collectibility of our receivables. See "Disclosure Regarding Forward-Looking Statements" contained in our annual report for the year ended September 30, 2003, filed on Form 10-K with the Securities and Exchange Commission.

#### SEASONALITY AND QUARTERLY FLUCTUATIONS

Our results of operations from residential construction are seasonal, depending on weather trends, with typically higher revenues generated during spring and summer and lower revenues during fall and winter. The commercial and industrial aspect of our business is less subject to seasonal trends, as this work generally is performed inside structures protected from the weather.

31

Our service and maintenance business is generally not affected by seasonality. In addition, the construction industry has historically been highly cyclical.

Our volume of business may be adversely affected by declines in construction projects resulting from adverse regional or national economic conditions. Quarterly results may also be materially affected by the timing of new construction projects, acquisitions and the timing and magnitude of acquisition assimilation costs. Accordingly, operating results for any fiscal period are not necessarily indicative of results that may be achieved for any subsequent fiscal period.

#### NEW ACCOUNTING PRONOUNCEMENT

In January 2003, the Financial Accounting Standards Board issued Interpretation No. 46, "Consolidation of Variable Interest Entities," ("Interpretation 46"). The objective of Interpretation 46 is to improve the financial reporting by companies involved with variable interest entities. Until now, one company generally has included another entity in its consolidated financial statements only if it controlled the entity through voting interest. Interpretation 46changes that by requiring a variable interest entity to be consolidated by a company if that company is subject to a majority of the risk of loss from the variable interest entity's activities or entitled to receive a majority of the entity's residual returns or both. The consolidation requirements of Interpretation 46 apply immediately to variable interest entities created after January 31, 2003. The consolidation requirements apply to older entities in the first fiscal year or interim period ending after March 15, 2004. Certain disclosure requirements apply to all financial statements issued after January 31, 2003, regardless of when the variable interest entity was established. We have investments in two firms, EnerTech Capital Partners II, L.P. (EnerTech) and Energy Photovoltaics, Inc. (EPV) that might fall under this interpretation. We determined that EPV was excepted out of the provisions of Interpretation 46 and that it was not the primary beneficiary of EnerTech and as such, the adoption of this interpretation did not have a material impact on our results of operations or our financial position.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Management is actively involved in monitoring exposure to market risk and continues to develop and utilize appropriate risk management techniques. We are not exposed to any significant market risks from commodity price risk or foreign currency exchange risk. Our exposure to significant market risks includes outstanding borrowings under our floating rate credit facility. Management does not use derivative financial instruments for trading purposes or to speculate on changes in interest rates or commodity prices.

32

As a result, our exposure to changes in interest rates results from our short-term and long-term debt with both fixed and floating interest rates. The following table presents principal or notional amounts (stated in thousands) and related interest rates by year of maturity for our debt obligations and their indicated fair market value at March 31, 2004:

<TABLE> <CAPTION>

		2004		2005		2006		2007		2008	Т	HEREAFTER		TOTAL
<s> Liabilities - Debt:</s>	<c></c>		<c></c>	>	<c:< th=""><th>&gt;</th><th><c:< th=""><th>&gt;</th><th>&lt;0</th><th>:&gt;</th><th><c></c></th><th></th><th><c< th=""><th>:&gt;</th></c<></th></c:<></th></c:<>	>	<c:< th=""><th>&gt;</th><th>&lt;0</th><th>:&gt;</th><th><c></c></th><th></th><th><c< th=""><th>:&gt;</th></c<></th></c:<>	>	<0	:>	<c></c>		<c< th=""><th>:&gt;</th></c<>	:>
Variable Rate (term loan) Variable Interest Rate Fixed Rate (senior subordinated		3,571 4.12%	\$	7,143 4.12%	\$	7,143 4.12%	\$	7,143 4.12%	\$	25,000 4.12%	\$		Ş	50,000 4.12%
notes) Fixed Interest Rate	\$	 9.375%	\$	 9.375%	\$	 9.375%	\$	 9.375%	\$	 9.375%	\$	172,885 9.375%	Ş	172,885 9.375%

50,000

181,097

\$

Fair Value of Debt:

Variable Rate Fixed Rate </TABLE>

#### ITEM 4. CONTROLS AND PROCEDURES

As of March 31, 2004, an evaluation was performed under the supervision and with the participation of the Company's management, including the CEO, of the effectiveness of the design and operation of the Company's disclosure controls and procedures. Based on that evaluation, the Company's management, including the CEO, concluded that the Company's disclosure controls and procedures were effective as of March 31, 2004 in providing reasonable assurances that material information required to be disclosed is included on a timely basis in the reports it files with the Securities and Exchange Commission.

Since the date of the evaluation, there have been no significant changes in the Company's internal controls or in other factors that could significantly affect these controls.

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

#### PART II. OTHER INFORMATION

33

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

- (A) The Company held its annual meeting of stockholders in Houston, Texas on January 22, 2004. The following sets forth matters submitted to a vote of the stockholders:
- (B) The following individuals were elected to the Board of Directors as stated in the Company's Proxy Statement dated December 12, 2003, for terms expiring at the 2007 annual stockholders' meeting or until their successors have been elected and qualified - Class III Directors: Donald P. Hodel, Donald C. Trauscht and James D. Woods.

Mr. Hodel was elected by a vote of 31,025,428 shares, being more than a majority of the common stock of the Company, and 610,925 shares withheld. Mr. Trauscht was elected by a vote of 30,972,545 shares, being more than a majority of common stock of the Company, and 663,808 shares withheld. Mr. Woods was elected by a vote of 30,971,679 shares, being more than a majority of the common stock of the Company, and 664,674 shares withheld.

(C) The stockholders approved the Company's Employee Stock Purchase Plan, as amended and restated, including the issuance of an additional 1,000,000 shares of the Company's common stock to be issued pursuant thereto, by a vote of 23,515,898 shares, being a majority of the common stock and restricted voting common stock of the Company, with 2,263,469 shares of common stock voted against and 8,911 shares of common stock abstaining.

The stockholders ratified the appointment of Ernst & Young LLP to audit the financial statements of the Company and its subsidiaries, by a vote of 32,852,378 shares, being more than a majority of the common stock and restricted voting common stock of the Company, with 83,153 of common stock voted against, and 3,677 shares of common stock abstaining.

34

#### INTEGRATED ELECTRICAL SERVICES, INC. AND SUBSIDIARIES

#### PART II. OTHER INFORMATION

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

#### A. EXHIBITS

- 10.1 Credit Facility dated February 27, 2004, among the Company, as borrower, the Financial Institutions named therein, as banks, U.S. Bank National Association as syndication agent, Bank of Scotland as managing agent, LaSalle Bank National Association as Documentation agent and Bank One, NA as administrative agent.
- 31.1 Certification of Herbert R. Allen, Chief Executive Officer and Chief Financial Offer, pursuant to Section 302 of The Sarbanes-Oxley Act of 2002.
- 32.1 Certification of Herbert R. Allen, Chief Executive Officer and Chief Financial Offer, pursuant to Section 906 of The Sarbanes-Oxley Act of 2002.
- B. REPORTS ON FORM 8-K

On March 2, 2004, the Company filed a Current Report on Form 8-K in connection with the press release of its Company and Investment Profile dated March 2004.

#### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized, who has signed this report on behalf of the Registrant and as the principal financial officer of the Registrant.

INTEGRATED ELECTRICAL SERVICES, INC.

Date: April 29, 2004

By: /s/ Herbert R. Allen

Herbert R. Allen Chief Executive Officer and Interim Chief Financial Officer

36

#### EXHIBIT INDEX

- 10.1 Credit Facility dated February 27, 2004, among the Company, as borrower, the Financial Institutions named therein, as banks, U.S. Bank National Association as syndication agent, Bank of Scotland as managing agent, LaSalle Bank National Association as Documentation agent and Bank One, NA as administrative agent.
- 31.1 Certification of Herbert R. Allen, Chief Executive Officer and Chief Financial Offer, pursuant to Section 302 of The Sarbanes-Oxley Act of 2002.
- 32.1 Certification of Herbert R. Allen, Chief Executive Officer and Chief Financial Offer, pursuant to Section 906 of The Sarbanes-Oxley Act of 2002.

## CREDIT AGREEMENT

## among

INTEGRATED ELECTRICAL SERVICES, INC., as Borrower,

THE FINANCIAL INSTITUTIONS NAMED IN THIS CREDIT AGREEMENT, as Banks,

U. S. BANK NATIONAL ASSOCIATION, as Syndication Agent,

BANK OF SCOTLAND, as Managing Agent,

LaSALLE BANK NATIONAL ASSOCIATION, as Documentation Agent,

and

BANK ONE, NA, as Administrative Agent

Dated as of February 27, 2004

Sole Bookrunner and Lead Arranger:

Bank One Capital Markets, Inc.

<table></table>		
<s></s>		<c></c>
ARTICLE	1 DEFINITIONS AND ACCOUNTING TERMS	1
1.1	Certain Defined Terms	1
1.2	Computation of Time Periods	20
1.3	Accounting Terms; Preparation of Financials	20
1.4	Types	21
1.5	Interpretation	21
ARTICLE	2 CREDIT FACILITIES	23
2.1	Term Loan Facility	23
2.2	Revolving Loan Facility	25
2.3	Letter of Credit Facility	28
2.4	Swing Line Facility	32
2.5	Fees	34
2.6	Interest	35

2.7	Breakage Costs	38
2.8	Increased Costs	39
2.9	Illegality	40
2.10	Market Failure	40
2.11	Payment Procedures and Computations	41
2.12	Taxes	43
2.13	Change of Lending Office	44
ARTICLE	3 CONDITIONS PRECEDENT	45
3.1	Conditions Precedent to Initial Extension of Credit	45
3.2	Conditions Precedent to Each Extension of Credit	46
ARTICLE	4 REPRESENTATIONS AND WARRANTIES	46
4.1	Organization	47
4.2	Authorization	47
4.3	Enforceability	48
4.4	Absence of Conflicts and Approvals	48
4.5	Investment Companies	48
4.6	Public Utilities	48
4.7	Financial Condition	48
4.8	Condition of Assets	49
4.9	Litigation	49
4.10	Subsidiaries	49
4.11	Laws and Regulations	49
4.12	Environmental Compliance	50
4.13	ERISA	50
4.14	Taxes	50
4.15	True and Complete Disclosure	51
4.16	Senior Indebtedness	51
4.17	Certain Documents	51
ARTICLE	5 COVENANTS	51
5.1	Organization	51
5.2	Reporting	52
5.3	Inspection	54

 > |  |

### <TABLE>

	<c></c>
Use of Proceeds	54
Financial Covenants	54
Debt	56
Liens	56
Other Obligations	56
Corporate Transactions and Asset Sales	57
Distributions	58
Transactions with Affiliates	58
Insurance	59
Investments	59
Lines of Business	59
Compliance with Laws	60
Environmental Compliance	60
ERISA Compliance	60
Payment of Certain Claims	60
	Financial Covenants. Debt. Liens. Other Obligations. Corporate Transactions and Asset Sales. Distributions. Transactions with Affiliates. Insurance. Investments. Lines of Business. Compliance with Laws. Environmental Compliance.

5.19	Subsidiaries	61
5.20	Changes in Fiscal Periods	61
5.21	Negative Pledge Clauses	61
5.22	Subordinated Debt Payments	61
ARTICLE	6 DEFAULT AND REMEDIES	61
6.1	Events of Default	61
6.2	Termination of Commitments	63
6.3	Acceleration of Credit Obligations	64
6.4	Cash Collateralization of Letters of Credit	64
6.5	Default Interest	64
6.6	Right of Setoff; Adjustments	64
6.7	Actions Under Credit Documents	65
6.8	Remedies Cumulative	65
6.9	Application of Payments	65
ARTICLE	7 THE ADMINISTRATIVE AGENT	66
7.1	Appointment; Nature of Relationship	66
7.2	Powers	67
7.3	General Immunity	67
7.4	No Responsibility for Loans, Recitals, etc	67
7.5	Acting on Instructions of Banks	67
7.6	Employment of Agents and Counsel	68
7.7	Reliance on Documents; Counsel	68
7.8	Administrative Agent's Reimbursement and Indemnification	68
7.9	Notice of Default	69
7.10	Rights as a Bank	69
7.11	Bank Credit Decision	69
7.12	Successor Administrative Agent	70
7.13	Administrative Agent and Arranger Fees	70
7.14	Delegation to Affiliates	71
7.15	Execution of Collateral Documents	71
7.16	Collateral Releases	71

 > |  |

## .

## <TABLE>

<s></s>		<c></c>
7.17	Syndication and Documentation Agents	71
ARTICLE	8 BENEFIT OF AGREEMENT; ASSIGNMENT; PARTICIPATIONS	71
8.1	Successors and Assigns	71
8.2	Participations	72
8.3	Assignments	73
8.4	Dissemination of Information	74
8.5	Tax Treatment	74
ARTICLE	9 MISCELLANEOUS	74
9.1	Expenses	74
9.2	Indemnification	75
9.3	Modifications, Waivers, and Consents	75
9.4	Survival of Agreements	76
9.5	Notice	76
9.6	CHOICE OF LAW	76
9.7	Submission To Jurisdiction; Waivers	76
9.8	WAIVER OF JURY TRIAL	77

9.9	Counterparts	77
9.10	No Further Agreements	77
9.11	2001 Credit Agreement	77
<td>&gt;</td> <td></td>	>	

<table></table>	
<s></s>	
EXHIBITS:	

<C>

Exhibit	A	Form of	Compliance Certificate
Exhibit	B-1	Form of	Revolving Loan Borrowing Request
Exhibit	в-2	Form of	Term Loan Borrowing Request
Exhibit	С	Form of	Continuation/Conversion Request
Exhibit	D-1	Form of	Revolving Loan Note
Exhibit	D-2	Form of	Term Loan Note
Exhibit	D-3	Form of	Swing Line Note
Exhibit	E	Form of	Assignment and Acceptance
Exhibit	F	Closing	Documents List
Exhibit	G	Form of	Joinder Agreement
Exhibit	Н	Form of	Acquisition Certificate
Exhibit	I	Joinder	Deliveries

## SCHEDULES:

Schedule	I-A	Commitments
Schedule	I-B	Administrative Information (Borrower; Administrative Agent; Banks)
Schedule	II	Disclosures (Existing Subsidiaries)
Schedule	III	Restricted Payment Terms (Qualified Preferred Stock)
Schedule	IV	Good Standing Exceptions

  |  |

### CREDIT AGREEMENT

This Credit Agreement dated as of February 27, 2004, is among INTEGRATED ELECTRICAL SERVICES, INC., a Delaware corporation, as Borrower, the financial institutions named herein, as Banks, U. S. BANK NATIONAL ASSOCIATION, as syndication agent, BANK OF SCOTLAND, as managing agent, LaSALLE BANK NATIONAL ASSOCIATION, as documentation agent, and BANK ONE, NA, with its main office in Chicago, Illinois, as administrative agent.

The Borrower, the financial institutions (collectively, the "Prior Banks") party thereto, Credit Lyonnais and The Bank of Nova Scotia, as syndication agents, Toronto Dominion (Texas), Inc., as documentation agent, and JPMorgan Chase Bank, as administrative agent (in such capacity, the "Prior Administrative Agent") are parties to that certain Credit Agreement (as heretofore amended and supplemented, the "2001 Credit Agreement") dated as of May 22, 2001. Pursuant to the terms of the 2001 Credit Agreement, the Prior Administrative Agent has resigned in favor of the Administrative Agent, and by their execution of this Agreement the Banks have appointed the Administrative Agent as the successor to the Prior Administrative Agent, the Administrative Agent has accepted such appointment and the Borrower has consented thereto; the Administrative Agent therefore has succeeded to all of the rights, powers and obligations of the Prior Administrative Agent under the 2001 Credit Agreement. Further, the Borrower, the Banks and the Administrative Agent have agreed to (a) amend and restate the 2001 Credit Agreement in its entirety and (b) release, on the terms set forth herein, each of the Prior Banks (other than any Prior Bank which is also a Bank) and the Prior Administrative Agent from its obligations under the 2001 Credit Agreement; provided that the Prior Administrative Agent shall continue to enjoy all of its rights (including indemnities) under the 2001 Credit Agreement with respect to its actions in such capacity before the date hereof.

The parties hereto agree to amend and restate the 2001 Credit Agreement in its entirety as follows:

## ARTICLE I

### DEFINITIONS AND ACCOUNTING TERMS

1.1 Certain Defined Terms. As used in this Agreement, the following terms shall have the following meanings (unless otherwise indicated, such meanings to be equally applicable to both the singular and plural forms of the terms defined):

"Acquisition" means the direct or indirect purchase or acquisition, whether in one or more related transactions, of any Person or group of Persons or any related group of assets, liabilities, or securities of any Person or group of Persons.

"Acquisition Certificate" means an acquisition certificate executed by a Responsible Officer of the Borrower in substantially the form of Exhibit H.

"Administrative Agent" means Bank One, NA, together with its affiliates, as the arranger of the Commitments and as the administrative agent for the Banks under this Agreement and the other Credit Documents, together with any of its successors.

"Affiliate" means, as to any Person, any other Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person or any Subsidiary of such Person. The term "control" (including the terms "controlled by" or "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership, by contract, or otherwise.

"Agreement" means this Credit Agreement.

"Applicable Lending Office" means, with respect to each Bank and for any particular type of transaction, the office of such Bank set forth in Schedule I-B to this Agreement (or in the applicable Assignment and Acceptance by which such Bank joined this Agreement) as its applicable lending office for such type of transaction or such other office of such Bank as such Bank may from time to time specify in writing to the Borrower and the Administrative Agent for such particular type of transaction.

"Applicable Margin" means, with respect to interest rates and letter of credit fees and as of any date of its determination, an amount equal to the percentage amount set forth in the table below opposite the applicable Total Debt to EBITDA ratio ("X" in the table below), measured as provided in Section 5.5(b)(ii), of the Borrower for the four fiscal quarters then most recently ended:

<TABLE> <CAPTION>

Total Debt to EBITDA	Applicable Margin LIBOR Tranches and Letter of Credit Fee	Applicable Margin Prime Rate Tranche
<s></s>	<c></c>	<c></c>
X < or = 2.25	1.75%	0%
2.25 < X < or = 2.75	2.00%	0%
2.75 < X < or = 3.25	2.25%	0%
3.25 < X < or = 3.75	2.50%	.25%
3.75 < X	2.75%	.50%

  |  |The foregoing ratio and resulting Applicable Margin shall be based upon Schedule C of the most recent Compliance Certificate delivered to the Administrative Agent pursuant to Section 5.2(a) or Section 5.2(b) (provided that for the period from the date of this Agreement until the date six months after the Closing Date, the Total Debt to EBITDA ratio shall be deemed to be greater than 3.75 and the Applicable Margin shall be set accordingly).

Any adjustments to the Applicable Margin shall become effective on the 5th Business Day following the Administrative Agent's receipt of a Compliance Certificate; provided, however, that if any such Compliance Certificate is not delivered when required hereunder, the Applicable Margin shall be deemed to be the maximum percentage amount in each table until such Compliance Certificate is received by the Administrative Agent.

Upon any change in the Applicable Margin, the Administrative Agent shall promptly notify the Borrower and the Banks of the new Applicable Margin.

"Approved Fund" means any Fund that is administered or managed by (a) a Bank, (b) an Affiliate of a Bank or (c) an entity or an Affiliate of an entity that administers or manages a Bank.

"Arranger" means Banc One Capital Markets, Inc.

"Asset Sale" means, with respect to any Person, any sale, transfer or other disposition (including any loss, destruction or condemnation) of any property, other than (a) the sale of inventory in the ordinary course of business; (b) the transfer of property between Restricted Entities; (c) any disposition of property in connection with the acquisition of property of a similar kind and character, provided that the fair market value of such acquired property shall be at least equal to the fair market value of such disposed property, and (d) the sale of assets (collectively, the "Tesla Assets") by Tesla Power and Automation, L.P. and Tesla Power Properties, L.P.; provided that the Banks hereby direct the Administrative Agent to release its Liens on that part of the Tesla Assets covered by the Security Documents upon the sale thereof if and only if no Default has then occurred and is continuing or would result therefrom.

"Assignment and Acceptance" means an Assignment and Acceptance in substantially the form of Exhibit E executed by an assignor Bank, an assignee Bank, and (if applicable) the Administrative Agent and/or the Borrower, in accordance with Section 8.3.

"Available Revolving Loan Commitments" means an amount equal to the excess, if any, of (a) the Revolving Loan Commitments then in effect over (b) the aggregate outstanding principal amount of the Revolving Loan Borrowings plus the aggregate outstanding principal amount of the Swing Line Loans plus the Letter of Credit Exposure.

"Bankruptcy Event of Default" has the meaning set forth in Section 6.1(h).

"Banks" means the lenders listed as Banks on the signature pages of this Agreement and each Purchaser that shall become a party to this Agreement pursuant to Section 8.3.

"Base Rate" means, for any day, the fluctuating rate per annum of interest equal to the greater of (a) the Prime Rate in effect on such day or (b) the Federal Funds Rate in effect on such day plus 0.50%.

"Board" means the Board of Governors of the Federal Reserve System of the United States (or any successor).

"Borrower" means Integrated Electrical Services, Inc., a Delaware corporation.

"Borrowings" means the collective reference to the Term Loan Borrowings and the Revolving Loan Borrowings.

"Business Day" means any Monday through Friday during which commercial banks are open for business in Houston, Texas, and, if the applicable Business Day relates to any LIBOR Tranche, on which dealings are carried on in the London interbank market.

"Capital Expenditures" means, with respect to any Person and any period of its determination, the consolidated expenditures of such Person during such period that are required to be included in or are reflected by the consolidated property, plant, or equipment accounts of such Person, or any similar fixed asset or long term capitalized asset accounts of such Person, on the consolidated balance sheet of such Person in conformity with generally accepted accounting principles.

"Capital Lease" means, with respect to any Person, any lease of any property by such Person which would, in accordance with generally accepted accounting principles, be required to be classified and accounted for as a capital lease on the balance sheet of such Person.

"Capital Lease Obligations" means, as to any Person, the obligations of such Person to pay rent or other amounts under any Capital Lease.

"Change of Control" means, with respect to the Borrower, (a) the direct or indirect acquisition after the date hereof by any Person or related Persons constituting a group of (i) beneficial ownership of issued and outstanding shares of Voting Securities of the Borrower, the result of which acquisition is that such Person or such group possesses 35% or more of the combined voting power of all then issued and outstanding Voting Securities of the Borrower or (ii) the power to elect, appoint, or cause the election or appointment of at least a majority of the members of the board of directors of the Borrower, or (b) the individuals who, at the beginning of any period of 12 consecutive months, constitute the Borrower's board of directors (together with any new director whose election by the Borrower's board of directors or whose nomination for election by the Borrower's stockholders entitled to vote thereon was approved by a vote of at least a majority of the directors then still in office who either were directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason (other than death or disability) to constitute a majority of the Borrower's board of directors then in office.

"Closing Date" means the date on which the conditions precedent set forth in Section 3.1 shall have been satisfied, which date is February 27, 2004.

"Collateral" means any property (whether real or personal, tangible or intangible, or mixed) subject to a Lien in favor of the Administrative Agent pursuant to any Security Document.

"Code" means the Internal Revenue Code of 1986, as amended, or any successor statute.

"Commitment" means, as to any Bank, the sum of the Term Loan Commitment and the Revolving Loan Commitment of such Bank.

"Commitment Fee Rate" means, as of any date of its determination, an amount equal to the percentage amount set forth in the table below opposite the applicable Total Debt to EBITDA ratio ("X" in the table below), measured as provided in Section 5.5(b)(ii), of the Borrower for the four fiscal quarters then most recently ended:

<TABLE> <CAPTION> Total Debt

Commitment

to EBITDA						Fee Rate	
<s></s>							<c></c>
		Х	<	or	=	2.25	0.375%
2.25	<	Х	<	or	=	2.75	0.375%
2.75	<	Х	<	or	=	3.25	0.500%
3.25	<	Х	<	or	=	3.75	0.500%
3.75	<	Х					0.500%

The foregoing ratio and resulting Commitment Fee Rate shall be based upon Schedule C of the most recent Compliance Certificate delivered to the Administrative Agent pursuant to Section 5.2(a) or Section 5.2(b) (provided that for the period from the date of this Agreement until the date six months after the Closing Date, the Total Debt to EBITDA ratio shall be deemed to be greater than 3.75 and the Commitment Fee Rate shall be set accordingly).

Any adjustments to the Commitment Fee Rate shall become effective on the 5th Business Day following the Administrative Agent's receipt of a Compliance Certificate; provided, however, that if any such Compliance Certificate is not delivered when required hereunder, the Commitment Fee Rate shall be deemed to be the maximum percentage amount in the table until such Compliance Certificate is received by the Administrative Agent.

Upon any change in the Commitment Fee Rate, the Administrative Agent shall promptly notify the Borrower and the Banks of the new Commitment Fee Rate.

"Commonly Controlled Entity" means, with respect to any Person, any other Person which is under common control with such Person within the meaning of Section 414 of the Code.

"Compliance Certificate" means a compliance certificate executed by a Responsible Officer of the Borrower in substantially the form of Exhibit A, including the following attached Schedules:

> Schedule A: The applicable financial reports provided under Section 5.2(a) or 5.2(b) ending on the date of the computation of the financial covenants.

Schedule B: A schedule of any adjustments to the financial reports in Schedule A to the Compliance Certificate, listed on a company-by-company basis, that are requested by the Borrower to reflect the financial results of Acquisitions made prior to the end of the applicable period, together with the supporting financial reports of the Acquisitions from which the Borrower prepared such adjustments, prepared in accordance with Section 1.3(c) and otherwise in a form acceptable to the Administrative Agent.

Schedule C: The computation of the financial covenants under this Agreement that are based upon the financial reports in Schedule A or Schedule B, as applicable, to the Compliance Certificate, in a form acceptable to the Administrative Agent. "Contemplated Senior Subordinated Note Retirement" means the purchase, redemption, defeasance or retirement of Senior Subordinated Notes (a) in an aggregate outstanding principal amount of no more than \$100,000,000 and (b) to be consummated by no later than 180 days after the Closing Date.

"Continuation/Conversion Request" means a Continuation/Conversion Request in substantially the form of Exhibit C executed by a Responsible Officer of the Borrower and delivered to the Administrative Agent.

"Contract Status Report" means a report, in form and substance acceptable to the Administrative Agent, detailing the status of each contract of any Restricted Entity which contract has a value equal to or greater than \$7,500,000.

"Credit Documents" means this Agreement, the Term Loan Notes, the Revolving Loan Notes, the Swing Line Note, the Letter of Credit Documents, the Guaranty, the Security Documents, the Interest Hedge Agreements, and each other agreement, instrument, or document executed at any time in connection with this Agreement.

"Credit Obligations" means all principal and interest (including interest accruing after the maturity of the loans made hereunder and interest accruing after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding, relating to the Borrower, whether or not a claim for post-filing or post-petition interest is allowed in such proceeding) and fees, reimbursements, indemnifications, and other amounts now or hereafter owed by the Borrower to the Administrative Agent and the Banks (or with respect to the Interest Hedge Agreements, any Affiliates of the Banks) under this Agreement, the Term Loan Notes, the Revolving Loan Notes, the Swing Line Note, the Letter of Credit Documents, and the other Credit Documents and any increases, extensions, and rearrangements of those obligations under any amendments, supplements, and other modifications of the documents and agreements creating those obligations.

"Credit Parties" means the Borrower and the Guarantors.

"Debt" means, with respect to any Person, without duplication, (a) indebtedness of such Person for borrowed money, (b) obligations of such Person evidenced by bonds, debentures, notes, or other similar instruments, (c) obligations of such Person to pay the deferred purchase price of property or services (other than trade debt and normal operating liabilities incurred in the ordinary course of business), (d) obligations of such Person as lessee under Capital Leases, (e) obligations of such Person under or relating to letters of credit, (f) obligations of such Person under or relating to guaranties, purchase agreements, or other assurances in favor of obligees of the kinds of indebtedness and obligations referred to in clauses (a) through (e) of this definition, (g) nonrecourse indebtedness or obligations of others of the kinds referred to in clauses (a) through (f) of this definition secured by any Lien on or in respect of any property of such Person, and (h) obligations of such Person evidenced by preferred stock or other equity interests in such Person which provide for mandatory redemption, mandatory payment of dividends, or similar rights to the payment of money; provided, however, that in each instance such term shall not

mean or include any Debt in respect of which monies sufficient to pay and discharge the same in full (either on the expressed date of maturity thereof or on such earlier date as such Debt may be duly

called for redemption and payment) shall be deposited with the trustee under the appropriate Senior Subordinated Note Indenture, if relevant, or otherwise a depository, agency or trustee acceptable to the Administrative Agent, in each case in trust for the payment thereof. For the purposes of determining the amount of any Debt, the amount of any Debt described in clauses (e) and (f) of this definition shall be valued at the maximum amount of the contingent liability thereunder, the amount of any Debt described in clause (g) that is not covered by clause (e) or (f) shall be valued at the lesser of the amount of the Debt secured or the book value of the property securing such Debt, and the amount of any Debt described in clause (h) shall be valued at the stated redemption value of such Debt as of the date of determination.

"Default" means (a) an Event of Default or (b) any event or condition which with notice or lapse of time or both would, unless cured or waived, become an Event of Default.

"Default Rate" means, with respect to any amount due hereunder, a per annum interest rate equal to (a) if such amount is either outstanding principal accruing interest based upon a rate established elsewhere in this Agreement or accrued but unpaid interest thereon, the sum of (i) the interest rate established elsewhere in this Agreement from time to time for such principal amount, including any Applicable Margin, plus (ii) 2.00% per annum or (b) in all other cases, the Base Rate in effect from time to time plus the Applicable Margin for the Prime Rate Tranche in effect from time to time plus 2.00% per annum.

"Derivatives" means any swap, hedge, cap, collar, or similar arrangement providing for the exchange of risks related to price changes in any commodity, including money.

"Dollars" or "\$" means lawful money of the United States of America.

"EBITDA" means, with respect to any Person and for any period of its determination, the consolidated net income (excluding any extraordinary gains or losses and all non-cash items, including depreciation, amortization and non-recurring charges) of such Person for such period, plus the Interest Expense and income taxes of such Person for such period, provided that, notwithstanding anything herein to the contrary, for the purposes of calculating EBITDA, the following shall be disregarded: (a) impairment to goodwill calculated in accordance with FASB Statement No. 142, Goodwill and Other Intangibles; (b) the prepayment premium due under the Senior Subordinated Note Indentures in connection with the Contemplated Senior Subordinated Note Retirement, and (c) losses from the write-down or write-off of the value of future (to the extent made pursuant to current commitments heretofore disclosed to the Banks in writing) and current investments in Energy Photovoltaics, Inc. and Entertech Capital Partners II L.P. "Eligible Junior Interests Sales" means all Junior Interests Sales except (a) to the extent that the proceeds thereof are used to purchase, redeem, retire or defease Senior Subordinated Notes; (b) the issuance or sale of the Borrower's capital stock (including treasury stock) in connection with the exercise of an option by an ESOP or by a beneficiary of an ESOP pursuant to the terms of such ESOP, and (c) the issuance or sale of the Borrower's capital stock (including treasury stock) pursuant to an ESOP.

"Environmental Law" means all federal, state, and local laws, rules, regulations, ordinances, orders, decisions, agreements, and other requirements now or hereafter in effect

relating to the pollution, destruction, loss, or injury of the environment, the presence of any contaminant in the environment, the protection, cleanup, remediation, or restoration of the environment, the creation, handling, transportation, use, or disposal of any waste product in the environment, exposure of persons to any contaminant, waste, or hazardous substance in the environment, and the health and safety of employees in relation to their environment.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ESOP" means (a) the IES, Inc. 1997 Director's Stock Plan; (b) the IES, Inc. 1997 Incentive Compensation Plan; (c) the IES, Inc. 1997 Stock Plan; (d) the IES, Inc. Employee Stock Repurchase Plan, and (e) similar successor employee benefit stock plans.

"Eurocurrency Reserve Requirements" means for any day as applied to a LIBOR Tranche, the aggregate (without duplication) of the maximum rates (expressed as a decimal fraction) of reserve requirements in effect on such day (including basic, supplemental, marginal and emergency reserves) under any regulations of the Board or other governmental authority having jurisdiction with respect thereto dealing with reserve requirements prescribed for eurocurrency funding (currently referred to as "Eurocurrency Liabilities" in Regulation D of the Board) maintained by a member bank of the Federal Reserve System.

"Eurodollar Base Rate" means, for any LIBOR Tranche for any Interest Period therefor, the applicable British Bankers' Association LIBOR rate for deposits in Dollars as reported by any generally recognized financial information service as of 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, and having a maturity equal to such Interest Period, provided that, if no such British Bankers' Association LIBOR rate is available to the Administrative Agent, the applicable LIBOR for the relevant Interest Period shall instead be the rate determined by the Administrative Agent to be the rate at which the Administrative Agent (in its individual capacity) or one of its Affiliate banks offers to place deposits in Dollars with first class banks in the interbank market at approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period, in the approximate amount of the Administrative Agent's (acting in its individual capacity) ratable share of such LIBOR Tranche and having a maturity equal to such Interest Period.

"Event of Default" has the meaning specified in Section 6.1.

"Facilities" means each of (a) the Term Loan Commitments and the Term Loan made thereunder and (b) the Revolving Loan Commitments and the extensions of credit made thereunder.

"Federal Funds Rate" means, for any period, a fluctuating per annum interest rate equal for each day during such period to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations at approximately 10:00 a.m., Houston time, for any such day on such transactions received by the Administrative Agent from three different Federal funds brokers of recognized standing selected by it in its sole discretion.

"Fee Letter" means the letter agreement dated January 22, 2004 among the Borrower, Bank One, NA and the Arranger.

"Financial Statements" means the audited annual financial statements of the Borrower for the period ending September 30, 2003, including the consolidated balance sheets of the Borrower as of the end of such fiscal year and the consolidated statements of income, stockholders' equity, and cash flows for such fiscal year, as contained in the Form 10 K filed by the Borrower with the Securities and Exchange Commission for the period ending on such date.

"Fixed Charges" means, with respect to any Person and as of the last day of any fiscal quarter, the sum of (a) the current maturities of Debt (other than the Revolving Loan, if relevant) of such Person as of such date plus (b) principal payments on the Senior Subordinated Notes (not including (x) principal payments made with the proceeds of Junior Interests Sales and (y) principal payments in connection with the Contemplated Senior Subordinated Note Retirement) and the Subordinated Debt for the preceding four fiscal quarters then ending plus (c) Interest Expense of such Person for the preceding four fiscal quarters then ending.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"Guarantors" means (a) the Subsidiaries of the Borrower that have executed or joined in the Guaranty and (b) any future Subsidiaries of the Borrower that join the Guaranty pursuant to Section 5.19.

"Guaranty" means the Guaranty dated as of May 22, 2001, made by the Subsidiaries of the Borrower in favor of the Administrative Agent guaranteeing the Credit Obligations. "Hazardous Materials" means any substance or material identified as a hazardous substance pursuant to the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended and as now or hereafter in effect; any substance or material regulated as a hazardous waste pursuant to the Resource Conservation and Recovery Act of 1976, as amended and as now or hereafter in effect; and any substance or material designated as a hazardous substance or hazardous waste pursuant to any other Environmental Law.

"Highest Lawful Rate" means the maximum lawful interest rate, if any, that at any time or from time to time may be contracted for, charged, or received under the laws applicable to the relevant Bank which are presently in effect or, to the extent allowed by law, under such applicable laws which may hereafter be in effect and which allow a higher maximum nonusurious interest rate than applicable laws now allow. On each day, if any, that Texas law establishes the Highest Lawful Rate, the Highest Lawful Rate shall be the "weekly ceiling" (as defined in the Texas Credit Code) for that day.

"Interest Expense" means, with respect to any Person and for any period of its determination, the consolidated interest expense of such Person during such period excluding the amortization of debt issuance costs.

"Interest Hedge Agreements" means any swap, hedge, cap, collar, or similar arrangement between the Borrower and any Bank (or any Affiliate of any Bank).

"Interest Period" means, with respect to each LIBOR Tranche, the period commencing on the date of such LIBOR Tranche and ending on the last day of the period selected by the Borrower pursuant to the provisions below. The duration of each such Interest Period shall be one, two, three, or six months, in each case as the Borrower may select in the applicable Term Loan Borrowing Request, Revolving Loan Borrowing Request or Continuation/Conversion Request (unless there shall exist any Default or Event of Default, in which case the Borrower may only select one month Interest Periods); provided, however, that:

a. whenever the last day of any Interest Period would otherwise occur on a day other than a Business Day, the last day of such Interest Period shall be extended to occur on the next succeeding Business Day; provided that if such extension would cause the last day of such Interest Period to occur in the next following calendar month, the last day of such Interest Period shall occur on the next preceding Business Day;

b. any Interest Period which begins on the last Business Day of the calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month in which it would have ended if there were a numerically corresponding day in such calendar month; and

c. the Borrower may not select an Interest Period for any LIBOR Tranche which ends after the Revolving Loan Maturity Date or after the date final payment is due on the Term Loan, as the case may be. "Interim Financial Statements" means the consolidated financial statements of the Borrower dated as of December 31, 2003, including the consolidated balance sheets of the Borrower as of the end of such fiscal quarter and the consolidated statements of income and cash flows for such fiscal quarter and for the fiscal year to date period ending on the last day of such fiscal quarter.

"Issuing Bank" means Bank One, NA.

"Joinder Agreement" has the meaning set forth in Section 5.19.

"Junior Interests Sale" means, with respect to any Person, (a) the issuance of capital stock (including the sale of treasury stock), partnership interests, membership interests in a limited liability company, beneficial interests in a trust or other equity ownership interests in a Person; (b) the issuance of any warrants, options or other rights entitling the holder thereof to purchase or acquire any such equity interest, and (c) the issuance of any Senior Subordinated Notes or Subordinated Debt, in each case except where both the issuer and the recipient are Restricted Entities.

"Letter of Credit" means any commercial or standby letter of credit issued by the Issuing Bank for the account of the Borrower pursuant to the terms of this Agreement.

"Letter of Credit Application" means the Issuing Bank's standard form letter of credit application for either a commercial or standby letter of credit, as the case may be, which has been executed by the Borrower and accepted by the Issuing Bank in connection with the issuance of a Letter of Credit.

"Letter of Credit Application Amendment" means the Issuing Bank's standard form application to amend a letter of credit for either a commercial or standby letter of credit, as the case may be, which has been executed by the Borrower and accepted by the Issuing Bank in connection with the increase or extension of a Letter of Credit.

"Letter of Credit Collateral Account" means a special cash collateral account pledged to the Administrative Agent containing cash deposited pursuant to Section 2.3(d) or 6.4 to be maintained with the Administrative Agent in accordance with Section 2.3(g).

"Letter of Credit Documents" means all Letters of Credit, Letter of Credit Applications, Letter of Credit Application Amendments, and agreements, documents, and instruments entered into in connection with or relating thereto.

"Letter of Credit Exposure" means, as of any date of its determination, the aggregate outstanding undrawn amount of all then outstanding Letters of Credit plus the aggregate then outstanding amount of the reimbursement obligations of the Borrower under the Letter of Credit Applications and this Agreement. "LIBOR" means, for any LIBOR Tranche for any Interest Period therefor, a rate per annum determined in accordance with the following formula (rounded upward to the nearest 1/100th of 1%):

> Eurodollar Base Rate ------1.00 - Eurocurrency Reserve Requirements

"LIBOR Tranche" means any Tranche which bears interest based upon the LIBOR, as determined in accordance with Section 2.6.

"Lien" means any mortgage, lien, pledge, charge, deed of trust, security interest, encumbrance, or other type of preferential arrangement to secure or provide for the payment of any obligation of any Person, whether arising by contract, operation of law, or otherwise (including any title retention for such purposes under any conditional sale agreement, any Capital Lease, or any other title transfer or retention agreement).

"Loan Advances" means the collective reference to the Term Loan Advances and the Revolving Loan Advances.

"Majority Banks" means, at any time, Banks holding more than 50% of the sum of (a) the aggregate then-unpaid principal amount of the Term Loan Notes held by the Banks (or, if no such principal amount is then outstanding and the Term Loan Borrowing has not been made, the aggregate amount of the Term Loan Commitments at such time) and (b) the aggregate amount of the Revolving Loan Commitments at such time (or, if the Revolving Loan Commitments have

been terminated, the aggregate then-unpaid principal amount of the Revolving Loan Notes held by the Banks and the Letter of Credit Exposure of the Banks at such time).

"Material Adverse Change" means any material adverse change in the business, assets, liabilities (actual or contingent), operations or condition (financial or otherwise) of the Borrower and its Subsidiaries on a consolidated basis.

"Mortgages" means a mortgage, deed of trust, security deed or other instrument, in form and substance satisfactory to the Administrative Agent, creating a Lien on certain real property of the Borrower and its Subsidiaries located in Maryland, South Carolina, Tennessee and Texas to secure the Credit Obligations.

"Net EBITDA" means, with respect to any Person and for any period of its determination, EBITDA minus cash tax expense minus \$12,000,000 minus Restricted Payments, in each case of such Person for such period and determined on a consolidated basis.

"Net Worth" means, with respect to any Person and as of any date of its determination, the excess of (a) the assets of such Person over (b) the liabilities of such Person, provided that, notwithstanding anything herein to

the contrary, for the purposes of calculating Net Worth, (x) impairments to goodwill calculated in accordance with FASB Statement No. 142, Goodwill and Other Intangibles, and (y) losses from the write-down or write-off of the value of future (to the extent made pursuant to current commitments heretofore disclosed to the Banks in writing) and current investments in Energy Photovolaics, Inc. and Entertech Capital Partners II L.P. shall be disregarded.

"Other Taxes" has the meaning set forth in Section 2.12(b).

"Partnership Pledge Agreement" means the Pledge Agreement dated as of May 22, 2001, made by the respective limited and general partners of each limited partnership Subsidiary of the Borrower, in favor of the Administrative Agent granting the Administrative Agent a security interest in the partnership interests of such partners in such limited partnerships, to secure the Credit Obligations.

"Permitted Debt" means all of the following Debt:

a. Debt in the form of the Credit Obligations;

b. Debt in the form of indebtedness for borrowed money and letters of credit owed by any Subsidiary of the Borrower prior to the acquisition of such Subsidiary by the Borrower in an Acquisition transaction, or owed by any Person that is the subject of any Acquisition assumed by the Borrower or any Subsidiary of the Borrower in connection with such Acquisition, provided that with respect to any such indebtedness, arrangements satisfactory to the Administrative Agent for the repayment of such indebtedness within 90 days following the closing of the Acquisition are made prior to the closing of the Acquisition and such arrangements are executed;

c. Debt in the form of (i) purchase money indebtedness and Capital Leases and (ii) other Debt, which Debt under clauses (i) and (ii), together are in an aggregate outstanding amount not to exceed \$3,500,000;

d. Debt in the form of Subordinated Debt and the Senior Subordinated Notes;

e. Debt in the form of Qualified Preferred Stock;

f. Debt in the form of reimbursement obligations for Surety Bonds procured in the ordinary course of business; and

g. renewals, extensions and refinancings of the Debt described above in the same or lesser principal amount on similar terms and conditions and in any event no less favorable to the interests of the Banks.

"Permitted Investments" means all of the following investments:

a. investments (including investments in the form of loans) in wholly owned Subsidiaries of the Borrower;

b. investments in the form of loans, guaranties, open accounts, and other extensions of trade credit in the ordinary course of business;

c. investments in commercial paper, bankers' acceptances, loan participation agreements, and other similar investments, in each case, maturing in twelve months or less from the date of issuance and which, at the time of acquisition are rated A 2 or better by Standard & Poor's Ratings Services and P 2 or better by Moody's Investors Service, Inc.;

d. investments in direct obligations of the United States, or investments in any Person which investments are guaranteed by the full faith and credit of the United States, in either case maturing in twelve months or less from the date of acquisition thereof and repurchase agreements having a term of less than one year and fully collateralized by such obligations which are entered into with banks or trust companies described in clause (e) below or brokerage companies having net worth in excess of \$250,000,000;

e. investments in time deposits or certificates of deposit maturing within one year from the date such investment is made, issued by a bank or trust company organized under the laws of the United States or any state thereof having capital, surplus, and undivided profits aggregating at least \$250,000,000 or a foreign branch thereof and whose long term certificates of deposit are, at the time of acquisition thereof, rated A 2 by Standard & Poor's Ratings Services or Prime 2 by Moody's Investors Service, Inc.;

f. investments in money market funds which invest solely in the types of investments described in paragraphs (c) through (e) above;

g. loans and advances to directors, officers, and employees of the Credit Parties made in the ordinary course of business in an aggregate outstanding amount not to exceed \$1,000,000;

h. acquisitions permitted by Section 5.9(c)(ii);

i. purchases of the Borrower's capital stock permitted by Section 5.10; and

j. other investments (including loans and advances), not to exceed \$10,000,000 in the aggregate in any fiscal year of the Borrower.

"Permitted Liens" means all of the following Liens:

a. Liens securing the Credit Obligations;

b. Liens securing purchase money debt, Capital Leases, and assumed or acquired indebtedness for borrowed money and letters of credit permitted under clause (b) of the definition of Permitted Debt, provided that no such Lien is spread to cover any property not (i) purchased in connection with the incurrence of such Debt, in the case of purchase money debt, or (ii) covered by such Lien at the time of the assumption or acquisition of the indebtedness secured thereby, in the case of assumed or acquired indebtedness for borrowed money and letters of credit; and

c. Liens arising in the ordinary course of business which are not incurred in connection with the borrowing of money, the obtaining of advances or credit, or payment of legal judgments and which do not materially detract from the value of any Restricted Entity's assets or materially interfere with any Restricted Entity's business, including such (i) Liens for taxes, assessments, or other governmental charges or levies not yet due or that are being contested in good faith by appropriate proceedings, provided that adequate reserves with respect thereto are maintained on the books of the Borrower or its Subsidiaries; (ii) Liens in connection with worker's compensation, unemployment insurance, or other social security, old age pension, or public liability obligations not yet due or that are being contested in good faith by appropriate proceedings; (iii) Liens in the form of legal or equitable encumbrances deemed to exist by reason of negative pledge covenants and other covenants or undertakings of like nature; (iv) Liens in the form of vendors', carriers', warehousemen's, repairmen's, mechanics', workmen's, materialmen's, construction, or other like Liens arising by operation of law in the ordinary course of business or incident to the construction or improvement of any property, including liens on such property securing reimbursement obligations for Surety Bonds procured in the ordinary course of business; and (v) Liens in the form of zoning restrictions, easements, licenses, and other restrictions on the use of real property or minor irregularities in title thereto which do not materially impair the use of such property in the operation of the business of the applicable Restricted Entity or the value of such property.

"Person" means an individual, partnership, corporation (including a business trust), limited liability company, joint stock company, trust, unincorporated association, joint venture, or other entity, or a government or any political subdivision or agency thereof, or any trustee, receiver, custodian, or similar official.

"Plan" means any (a) employee welfare benefit plan under Section 3(1) of ERISA, (b) employee pension benefit plan under Section 3(2) of ERISA, (c) multiemployer plan under Section 4001(a)(3) of ERISA, and (d) employee individual account benefit plan under Section 3(2) of ERISA, in each case, in which each Restricted Entity or any of their respective

Commonly Controlled Entities is or would be deemed to be an "employer" as defined in Section 3(5) of ERISA.

"Pledge Agreements" means, collectively (i) the Stock Pledge Agreement, (ii) the Partnership Pledge Agreement, and (iii) any future agreements granting the Administrative Agent a security interest in capital stock, partnership interests, or other membership interests of any Subsidiary (direct or indirect) of the Borrower, in each case, to secure the Credit Obligations.

"Prepayment Proceeds" means (a) all net cash proceeds (including, if applicable, cash proceeds received over time as and when the same are received) from an Asset Sale to the extent such proceeds, plus the net cash proceeds from all other Asset Sales received during the 12 months before such receipt of proceeds from such Asset Sale, exceeds \$5,000,000 and (b) 75% of all net cash proceeds (including, if applicable, cash proceeds received over time as and when the same are received) from Eligible Junior Interest Sales.

"Prime Rate" means, for any day, the fluctuating per annum interest rate in effect on such day equal to the rate of interest publicly announced by the Administrative Agent as its prime rate, whether or not the Borrower has notice thereof.

"Prime Rate Borrowing" shall mean that portion of any Borrowing which bears interest based upon the Base Rate as determined in accordance with Section 2.6.

"Prime Rate Tranche" shall mean the Tranche which bears interest based upon the Base Rate, as determined in accordance with Section 2.6.

"Prohibited Transaction" means any transaction set forth in Section 406 of ERISA or Section 4975 of the Code.

"Projections" has the meaning set forth in Section 5.2(f).

"Purchaser" has the meaning set forth in Section 8.3.

"Qualified Preferred Stock" means, with respect to the Borrower and as of any date of its issuance, any shares of preferred stock of the Borrower that (a) are issued after the date of this Agreement, (b) provide for their mandatory redemption on a date, if at all, that is on or after the first anniversary of the latest scheduled maturity of any of the Credit Obligations at the time issued, and (c) provide in the applicable certificate of designation for the redemption of such shares and for the blockage of Restricted Payments in respect of such shares during the existence of a Default or an Event of Default (i) on the terms and conditions set forth on Schedule III, or (ii) on terms approved by the Administrative Agent and the Majority Banks in their sole discretion, all such provisions to be in form and content satisfactory to the Administrative Agent and the Majority Banks in their sole discretion.

"ratable share" or "pro rata share" means, with respect to any Bank and as of any date of its determination, either (a) until the Closing Date, the ratio of such Bank's Commitment at such time to the aggregate Commitments at such time or (b) thereafter, the ratio of such Bank's aggregate outstanding Term Loan Advance and/or such Bank's Revolving Loan Commitment at such time (or, if the Revolving Loan Commitments have been terminated, Revolving Loan

Advances and share of the Letter of Credit Exposure at such time) to the aggregate outstanding Term Loan Advances and/or Revolving Loan Commitments at such time (or, if the Revolving Loan Commitments have been terminated, Revolving Loan Advances and Letter of Credit Exposure at such time), in each case as the context may require.

"Refunded Swing Line Loans" has the meaning set forth in Section

2.4(c)(i).

"Refunding Date" has the meaning set forth in Section 2.4(c)(ii).

"Related Parties" means, with respect to any Person, such Person's stockholders, directors, officers, employees, agents, Affiliates, successors, and assigns, and their respective stockholders, directors, officers, employees, and agents, and, with respect to any Person that is an individual, such Person's family relations and heirs.

"Reportable Event" means any of the events set forth in Section 4043 of ERISA.

"Responsible Officer" means, with respect to any Person, such Person's Chief Executive Officer, President, Chief Financial Officer, Secretary, Chief Accounting Officer, or any other officer of such Person designated by any of the foregoing in writing from time to time.

"Restricted Entities" means the Borrower and each Subsidiary of the Borrower.

"Restricted Payment" means the declaration or making by any Person of any (a) dividends; (b) purchase, redemption, retirement, or other acquisition for value any of its capital stock now or hereafter outstanding, or any distribution of assets to its stockholders as such, whether in cash, assets, or in obligations of it; (c) allocation or other setting apart of any sum for the payment of any dividend or distribution on, or for the purchase, redemption, or retirement of, any shares of its capital stock; or (d) making of any other distribution by reduction of capital or otherwise in respect of any shares of its capital stock; in each case, other than any such dividends, distributions, and payments payable solely in such Person's common stock.

"Revolving Loan" means the aggregate outstanding principal amount of the Revolving Loan Borrowings.

"Revolving Loan Advance" means the outstanding principal from a Bank which represents such Bank's ratable share of a Revolving Loan Borrowing.

"Revolving Loan Borrowing" means any aggregate amount of principal advanced on the same day and pursuant to the same Revolving Loan Borrowing Request under the revolving loan facility created in Section 2.2.

"Revolving Loan Borrowing Request" means a Revolving Loan Borrowing Request in substantially the form of Exhibit B-1 executed by a Responsible Officer of the Borrower and delivered to the Administrative Agent.

"Revolving Loan Commitment" means, for any Bank, the amount set forth opposite such Bank's name on Schedule I-A as its Revolving Loan Commitment, or if such Bank has entered into any Assignment and Acceptance since the date of this Agreement, as set forth for such Bank as its Revolving Loan Commitment in the Register maintained by the Administrative Agent pursuant to Section 8.3(d), in each case as such amount may be reduced pursuant to Section 21.2(e) or terminated pursuant to Section 6.2.

"Revolving Loan Maturity Date" means the date four years after the Closing Date.

"Revolving Loan Note" means a promissory note of the Borrower payable to the order of a Bank, in substantially the form of Exhibit D-1, evidencing the indebtedness of the Borrower to such Bank resulting from Revolving Loan Advances made by such Bank to the Borrower.

"Revolving Loan Percentage" means, as to any Bank at any time, the percentage which such Bank's Revolving Loan Commitment then constitutes of the aggregate Revolving Loan Commitments or, at any time after the Revolving Loan Commitments shall have expired or terminated, the percentage which the aggregate principal amount of such Bank's Revolving Loan Advances then outstanding constitutes of the aggregate principal amount of the Revolving Loan Advances then outstanding.

"Security Agreement" means the Security Agreement dated as of May 22, 2001, made by the Borrower and the Subsidiaries of the Borrower in favor of the Administrative Agent granting the Administrative Agent a security interest in the receivables, equipment, inventory and other personal property of each such Credit Party to secure the Credit Obligations.

"Security Documents" means the Security Agreement, the Pledge Agreements, the Mortgages and any other document creating or consenting to Liens in favor of the Administrative Agent securing Credit Obligations.

"Senior Debt" means all Debt of the Borrower and the Subsidiaries of the Borrower other than Debt in the form of Subordinated Debt and the Senior Subordinated Notes.

"Senior Subordinated Note Indentures" means (i) the Indenture, dated as of January 28, 1999, among the Borrower, certain of its Subsidiaries and U.S. Bank Corporate Trust, successor in interest to State Street Bank and Trust Company, as trustee, together with all instruments and other agreements entered into by the Borrower or such Subsidiaries in connection therewith, and (ii) the indenture(s) entered into by the Borrower and others in connection with the issuance of other Senior Subordinated Notes, together with all instruments and other agreements entered into by the Borrower and/or such Subsidiaries in connection therewith.

"Senior Subordinated Notes" means the outstanding 9 3/8% Senior Subordinated Notes due 2009 issued by the Borrower, together with the subsidiary guarantees thereof.

"Stock Pledge Agreement" means the Pledge Agreement dated as of May 22, 2001, as the same has been supplemented by Joinder Agreements prior to the date hereof, made by Borrower and certain Subsidiaries of Borrower in favor of the Administrative Agent granting the Administrative Agent a security interest in the capital stock or membership interests, as applicable, of each Subsidiary (whether direct or indirect) of Borrower, to secure the Credit Obligations.

"Subordinated Debt" means, with respect to the Borrower and as of any date of its issuance, any unsecured indebtedness for borrowed money, other than the Senior Subordinated Notes, for which the Borrower is directly and primarily obligated that (a) arises after the date of this Agreement, (b) does not have any stated maturity before the date six months following the latest maturity of any of the Credit Obligations at the time incurred, (c) has terms that are no more restrictive than the terms of the Credit Documents, which terms shall in any event (i) prohibit the acceleration of such indebtedness without 5 days' prior written notice to the Administrative Agent, (ii) not include a cross-default provision to other indebtedness (other than to a principal payment default of such other indebtedness), (iii) not include any financial covenants other than the type described in Section 5.5 and (iv) with respect any such financial covenants, be at least 20% less restrictive on the Borrower and its Subsidiaries than those contained herein, and (d) is expressly subordinated to the Credit Obligations (i) on the terms and conditions no less restrictive on the holders thereof than the subordination terms applicable to the Senior Subordinated Notes, or (ii) on terms approved by the Administrative Agent and the Majority Banks in their sole discretion, including payment subordination, remedy subordination, and related terms satisfactory to the Administrative Agent and the Majority Banks in their sole discretion.

"Subordinated Debt Indenture" means any indenture or like agreement entered into by the Borrower in connection with the issuance of Subordinated Debt, together with all instruments and other agreements entered into by the Borrower in connection therewith.

"Subsidiary" means, with respect to any Person, any other Person, a majority of whose outstanding Voting Securities (other than directors' qualifying shares) shall at any time be owned by such Person or one or more Subsidiaries of such person.

"Surety Bond" means any surety bond, insurance policy, indemnity agreement, guaranty, letter of credit or other instrument provided by a third party (that is, excluding an Affiliate of the obligor) to an obligee to assure the payment by and/or performance of an obligor.

"Swing Line Borrowing" means any aggregate amount of principal advanced on the same day and pursuant to the same notice under the swing line facility created in Section 2.4.

"Swing Line Commitment" means the obligation of the Swing Line Lender to make Swing Line Loans pursuant to Section 2.4 in an aggregate principal amount at any one time outstanding not to exceed \$10,000,000.

"Swing Line Lender" means Bank One, NA.

"Swing Line Loan" has the meaning set forth in Section 2.4(a).

"Swing Line Note" means the promissory note of the Borrower payable to the order of the Swing Line Lender, in substantially the form of Exhibit D-3, evidencing the indebtedness of the Borrower to the Swing Line Lender resulting from Swing Line Loans made by the Swing Line Lender to the Borrower.

"Swing Line Participation Amount" has the meaning set forth in Section 2.4(c)(ii).

"Term Loan" means the aggregate outstanding principal amount of the Term Loan Borrowing.

"Term Loan Advance" means the outstanding principal from a Bank which represents such Bank's ratable share of the Term Loan Borrowing.

"Term Loan Borrowing" means the aggregate amount of principal advanced pursuant to the Term Loan Borrowing Request under the term loan facility created in Section 2.1.

"Term Loan Borrowing Request" means the Term Loan Borrowing Request in substantially the form of Exhibit B-2 executed by a Responsible Officer of the Borrower and delivered to the Administrative Agent.

"Term Loan Commitment" means, for any Bank, the amount set forth opposite such Bank's name on Schedule I-A as its Term Loan Commitment, or if such Bank has entered into any Assignment and Acceptance since the date of this Agreement, as set forth for such Bank as its Term Loan Commitment in the Register maintained by the Administrative Agent pursuant to Section 8.3(d), in each case as such amount may be terminated pursuant to Section 6.2.

"Term Loan Note" means a promissory note of the Borrower payable to the order of a Bank, in substantially the form of Exhibit D-2, evidencing the indebtedness of the Borrower to such Bank resulting from the Term Loan Advance made by such Bank to the Borrower.

"Term Loan Percentage" means, as to any Bank at any time, the percentage which such Bank's Term Loan Commitment then constitutes of the aggregate Term Loan Commitments (or, at any time after the Closing Date, the percentage which the aggregate principal amount of such Bank's Term Loan Advance then outstanding constitutes of the aggregate principal amount of the Term Loan Advances then outstanding).

"Total Debt" means all Debt of the Borrower and the Subsidiaries of the Borrower.

"Tranche" means any tranche of principal outstanding under the Term Loan or the Revolving Loan accruing interest on the same basis, whether created in connection with new advances of principal under the Revolving Loan pursuant to Section 2.6(a)(i) or by the continuation or conversion of existing tranches of principal under the Term Loan or the Revolving Loan pursuant to Section 2.6(a)(ii), and shall include the Prime Rate Tranche or any LIBOR Tranche. "Type" has the meaning set forth in Section 1.4.

"Unbonded Accounts Receivable" means, with respect to any Person, the accounts receivable of such Person, determined on a consolidated basis and net of allowances, excluding accounts receivable arising under contracts where the account obligee has caused to be provided to the account debtor a Surety Bond.

"Voting Securities" means (a) with respect to any corporation, any capital stock of the corporation having general voting power under ordinary circumstances to elect directors of such corporation, (b) with respect to any partnership, any partnership interest having general voting

power under ordinary circumstances to elect the general partner or other management of the partnership, and (c) with respect to any other Person, such ownership interests in such Person having general voting power under ordinary circumstances to elect the management of such Person, in each case irrespective of whether at the time any other class of stock, partnership interests, or other ownership interest might have special voting power or rights by reason of the happening of any contingency.

1.2 Computation of Time Periods.

In this Agreement in the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each means "to but excluding."

1.3 Accounting Terms; Preparation of Financials.

a. All accounting terms, definitions, ratios, and other tests described herein shall be construed in accordance with United States generally accepted accounting principles applied on a consistent basis with those applied in the preparation of the Financial Statements, except as expressly set forth in this Agreement. In the event that any Accounting Change (as defined below) shall occur and such change results in a change in the method of calculation of financial covenants, standards or terms in this Agreement, then the Borrower and the Administrative Agent agree to enter into negotiations in order to amend such provisions of this Agreement so as to reflect equitably such Accounting Changes with the desired result that the criteria for evaluating the Borrower's financial condition shall be the same after such Accounting Changes as if such Accounting Changes had not been made. Until such time as such an amendment shall have been executed and delivered by the Borrower, the Administrative Agent and the Majority Banks, all financial covenants, standards and terms in this Agreement shall continue to be calculated or construed as if such Accounting Changes had not occurred. "Accounting Changes" refers to changes in accounting principles required by the promulgation of any rule, regulation, pronouncement or opinion by the Financial Accounting Standards Board of the American Institute of Certified Public Accountants or, if applicable, the Securities and Exchange Commission.

b. The Restricted Entities shall prepare their financial statements in accordance with United States generally accepted accounting principles applied

on a consistent basis with those applied in the preparation of the Financial Statements, unless otherwise approved in writing by the Administrative Agent.

c. The Borrower shall calculate the Applicable Margin and compliance with the financial covenants under this Agreement in Schedule C of the Compliance Certificate. Where expressly permitted in this Agreement, the Borrower may elect to base such calculations upon the financial reports in Schedule B of such Compliance Certificate. In such case the accounting terms, definitions, ratios, and other tests used in making such calculation shall be construed as required by paragraph (a) above except that the consolidated financial results of the Borrower shall be deemed to be the adjusted consolidated financial results of the Borrower set forth in Schedule B of the Compliance Certificate. Unless otherwise expressly permitted in this Agreement, the Borrower shall calculate the Applicable Margin or compliance with a financial covenant under

this Agreement based upon the financial reports in Schedule A of a Compliance Certificate. In such case the accounting terms, definitions, ratios, and other tests used in making such calculation shall be construed as required by paragraph (a) above. When calculating the Applicable Margin or compliance with a financial covenant under this Agreement based upon the financial reports in Schedule B of the Compliance Certificate, the Borrower shall prepare such calculations in accordance with the following provisions:

> i. The Borrower may select one or more Acquisitions for inclusion in the adjustments permitted in Schedule B or C of the Compliance Certificate; provided that if the inclusion of any Acquisition results in an increase in the consolidated EBITDA of the Borrower over the consolidated EBITDA of the Borrower required to be reported in Schedule A of the Compliance Certificate, then the Borrower must include all Acquisitions which would cause a decrease in the consolidated EBITDA of the Borrower in Schedule B and C of the Compliance Certificate.

ii. No Acquisition may be included in Schedule B or C of the Compliance Certificate at the request of the Borrower unless the financial reports of the acquired Person or assets from which Schedule B and C are prepared are (A) audited financial reports prepared by an independent certified public accounting firm, or (B) with respect to unaudited financial reports, are otherwise approved by the Majority Banks (provided that no Bank may condition such approval upon its receipt of a fee, however denominated).

### 1.4 Types.

The "Type" of a Tranche refers to the determination whether such tranche is a LIBOR Tranche or the Prime Rate Tranche.

## 1.5 Interpretation.

Article, Section, Schedule, and Exhibit references are to this Agreement, unless otherwise specified. All references to instruments, documents, contracts, and

agreements are references to such instruments, documents, contracts, and agreements as the same may be amended, supplemented, and otherwise modified from time to time, unless otherwise specified. The word "including" shall mean "including but not limited to." The word "or" shall mean "and/or" wherever necessary to prevent interpretation of any provision against the Administrative Agent or the Banks. Whenever the Borrower has an obligation under this Agreement and the other Credit Documents the expense of complying with that obligation shall be an expense of the Borrower unless otherwise specified. Whenever any determination is to be made by the Administrative Agent or any Bank, such determination shall be in such Person's sole discretion unless otherwise specified in this Agreement. If any provision in this Agreement or the other Credit Documents is held to be illegal, invalid, not binding, or unenforceable, such provision shall be fully severable and this Agreement and the other Credit Documents shall be construed and enforced as if such illegal, invalid, not binding, or unenforceable provision had never comprised a part of this Agreement or the other Credit Documents, and the remaining provisions shall remain in full force and effect. This Agreement and the other Credit Documents have been reviewed and negotiated by sophisticated parties with access to legal counsel and shall not be

construed against the drafter. In the event of a conflict between this Agreement and any other Credit Documents, this Agreement shall control.

# ARTICLE 2 CREDIT FACILITIES

## 2.1 Term Loan Facility

a. Term Loan Commitments. Each Bank severally agrees, on the terms and conditions set forth in this Agreement and for the purposes set forth in Section 5.4, to make a Term Loan Advance to the Borrower as such Bank's ratable share of the Term Loan Borrowing requested by the Borrower on the Closing Date, provided that the aggregate outstanding principal amount of the Term Loan Advance made by such Bank shall not exceed such Bank's Term Loan Commitment. The Term Loan Borrowing made on the Closing Date shall initially be a Prime Rate Borrowing, but may be immediately converted to a LIBOR Tranche in accordance with Section 2.6(a). The indebtedness of the Borrower to the Banks resulting from the Term Loan Advances made by the Banks shall be evidenced by Term Loan Notes made by the Borrower.

b. Method of Advancing.

i. The Term Loan Borrowing shall be made pursuant to a Term Loan Borrowing Request given by the Borrower to the Administrative Agent in writing or by telecopy not later than 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the anticipated Closing Date. The Term Loan Borrowing Request shall be fully completed and shall specify the information required therein, and shall be irrevocable and binding on the Borrower. Upon receipt of the Term Loan Borrowing Request by the Administrative Agent, the Administrative Agent shall promptly forward notice of the Term Loan Borrowing to the Banks. Each Bank shall, before 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the anticipated Closing Date, make available from its Applicable Lending Office to the Administrative Agent at the Administrative Agent's Applicable Lending Office, in immediately available funds, such Bank's ratable share of the Term Loan Borrowing. Subject to the satisfaction of all applicable conditions precedent, after receipt by the Administrative Agent of such funds, the Administrative Agent shall, by 4:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent), on the Closing Date make the Term Loan Borrowing available to the Borrower in immediately available funds at any account of Borrower which is designated in writing by the Borrower to the Administrative Agent.

ii. Unless the Administrative Agent shall have received notice from a Bank before the anticipated Closing Date that such Bank shall not make available to the Administrative Agent such Bank's ratable share of the Term Loan Borrowing, the Administrative Agent may assume that such Bank has made its ratable share of the Term Loan Borrowing available to the Administrative Agent on the Closing Date in accordance with paragraph (i) above, and the Administrative Agent may, in reliance upon such assumption, make available to the Borrower on such date a corresponding amount. If and to the extent that such Bank shall not have so made its ratable share of the Term Loan Borrowing available to the Administrative Agent, such Bank agrees that it shall pay interest on such amount for each day from the date such amount is made available to the Borrower by the Administrative Agent until the date such amount is paid to the Administrative Agent by such Bank at the Federal Funds Rate in effect from time to time, provided that with respect to such Bank if such amount is not paid by such Bank by the end of the second day after the Administrative Agent makes such amount available to the Borrower, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the

third day and shall remain at such increased rate thereafter. Interest on such amount shall be due and payable by such Bank upon demand by the Administrative Agent. If such Bank shall pay to the Administrative Agent such amount and interest as provided above, such amount so paid shall constitute such Bank's Term Loan Advance as part of the Term Loan Borrowing for all purposes of this Agreement even though not made on the same day as the other Term Loan Advances comprising the Term Loan Borrowing. In the event that such Bank has not repaid such amount by the end of the fifth day after such amount was made available to the Borrower, the Borrower agrees to repay to the Administrative Agent on demand such amount, together with interest on such amount for each day from the date such amount was made available to the Borrower until the date such amount is repaid to the Administrative Agent at the interest rate charged to the Borrower for the Term Loan Borrowing under the terms of this Agreement.

iii. The failure of any Bank to make available its ratable share of the Term Loan Borrowing shall not relieve any other Bank of its obligation, if any, to make available its ratable share of the Term Loan Borrowing. No Bank shall be responsible for the failure of any other Bank to honor such other Bank's obligations hereunder, including any failure to make available any funds as part of the Term Loan Borrowing.

#### c. Prepayment.

i. The Borrower may prepay the outstanding principal amount of the Term Loan pursuant to written notice given by the Borrower to the Administrative Agent in writing or by telecopy not later than (A) 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed prepayment, in the case of the prepayment of any portion of the Term Loan which is comprised of LIBOR Tranches, or (B) 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the same Business Day of the proposed prepayment, in the case of the prepayment of any portion of the Term Loan comprised solely of the Prime Rate Tranche. Each such notice shall specify the principal amount and Tranche or Tranches of the Term Loan which shall be prepaid, the date of the prepayment, and shall be irrevocable and binding on the Borrower. Prepayments of the Term Loan shall be made in integral multiples of \$500,000, in the case of prepayments of any LIBOR Tranches, or \$100,000 in the case of prepayments of the Prime Rate Tranche. If the prepayment would cause the aggregate outstanding principal amount of any LIBOR Tranche comprising all or any part of the Term Loan or the aggregate outstanding principal amount of the Prime Rate Tranche comprising all or any part of the Term Loan, to be less than \$1,000,000, in the case of any such LIBOR Tranche, or \$1,000,000, in the case of the Prime Rate Tranche, the prepayment must be in an amount equal to the entire outstanding principal amount of such LIBOR Tranche under the Term Loan or the entire outstanding principal amount of the Prime Rate Tranche under the Term Loan, as the case may be. Upon receipt of any notice of prepayment, the Administrative Agent shall give prompt notice of the intended prepayment to the Banks. For each such notice given by the Borrower, the Borrower shall prepay the Term Loan in the specified amount on the specified date as set forth in such notice. The Borrower shall have no right to prepay any principal amount of the Term Loan except as provided in this Section 2.1(c)(i); provided that the Borrower shall be obligated to prepay the Term Loan as provided in Section 2.1(c)(ii).

ii. Upon any Asset Sale or Junior Interests Sale by any Restricted Entity consummated before the payment in full of the Term Loan, the Borrower shall use the Prepayment Proceeds to prepay, on the date of the receipt of such proceeds, (A) first, the principal of the Prime Rate Tranche with respect to the Term Loan; (B) second, the principal of and unpaid accrued interest on the LIBOR Tranches of the Term Loan, selected in such order as to minimize the costs to be paid by the Borrower pursuant to Section 2.7 (and such costs shall be in addition to the mandatory prepayment due under this Section 2.1(c)(ii)); (C) third, the unpaid accrued interest on the Prime Rate Tranche with respect to the Term Loan, and (D), fourth, the unpaid principal of and accrued interest on the Revolving Loan as required by Section 2.2(c)(ii); none of the requirements of Section 2.1(c)(i) with respect to the prepayment of the Term Loan shall be applicable to mandatory prepayments under this Section 2.1(c)(ii).

iii. Each voluntary prepayment of principal under the Term Loan pursuant to Section 2.1(c)(i) shall be accompanied by payment of all accrued but unpaid interest on the principal amount prepaid, and each voluntary or mandatory prepayment of principal under the Term Loan pursuant to this Section 2.1(c) shall be accompanied by payment of and any amounts required to be paid pursuant to Section 2.7 as a result of such prepayment. The amount of each voluntary or mandatory principal prepayment under the Term Loan pursuant to this Section 2.1(c) shall be applied to the then-remaining principal installments of the Term Loan in inverse order of maturity. Amounts prepaid on account of the Term Loan may not be reborrowed.

#### d. Repayment.

The Term Loan Advance of each Bank shall mature in (1) 16 quarterly installments, each of which shall be in an amount equal to such Bank's Term Loan Percentage multiplied by \$1,785,714.29, the first of such quarterly installments to be due and payable on May 28, 2004 and the remaining quarterly installments to be due and payable on the last Business Day of each August, November, February and May thereafter, and (2) a final installment equal to the then-unpaid principal balance of such Bank's Term Loan Advance, which shall be due and payable on the fourth anniversary of the Closing Date.

## 2.2 Revolving Loan Facility.

a. Revolving Loan Commitments. Each Bank severally agrees, on the terms and conditions set forth in this Agreement and for the purposes set forth in Section 5.4, to make Revolving Loan Advances to the Borrower as such Bank's ratable share of Revolving Loan Borrowings requested by the Borrower from time to time on any Business Day during the period from the date of this Agreement until the Revolving Loan Maturity Date; provided that the aggregate outstanding principal amount of Revolving Loan Advances made by such Bank plus such Bank's ratable share of the aggregate outstanding principal amount of the Swing Line Loans plus such Bank's ratable share of the Letter of Credit Exposure shall not exceed such Bank's Revolving Loan Commitment. Revolving Loan Borrowings must be made in an amount equal to or greater than \$1,000,000, in the case of any Revolving Loan Borrowing comprised of a LIBOR Tranche, or \$1,000,000, in the case of any Prime Rate Borrowing, and be made in multiples of \$500,000, in the case of any Revolving Loan Borrowing comprised of a LIBOR Tranche, or \$100,000, in the case of any Prime Rate Borrowing. Any Revolving Loan Borrowing made on the Closing Date shall initially be a Prime Rate Borrowing, but may be immediately converted to a LIBOR Tranche in accordance with Section 2.6(a). Within the limits expressed in this Agreement, the

Borrower may from time to time borrow, prepay, and reborrow Revolving Loan Borrowings. The indebtedness of the Borrower to the Banks resulting from the Revolving Loan Advances made by the Banks shall be evidenced by Revolving Loan Notes made by the Borrower.

## b. Method of Advancing.

i. Each Revolving Loan Borrowing shall be made pursuant to a Revolving Loan Borrowing Request given by the Borrower to the Administrative Agent in writing or by telecopy not later than the time required pursuant to Section 2.6(a)(i) to select the interest rate basis for the Revolving Loan Borrowing. Each Revolving Loan Borrowing Request shall be fully completed and shall specify the information required therein, and shall be irrevocable and binding on the Borrower. Upon receipt of the Revolving Loan Borrowing Request by the Administrative Agent, the Administrative Agent shall promptly forward notice of the Revolving Loan Borrowing to the Banks. Each Bank shall, before 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the date of the requested Revolving Loan Borrowing, make available from its Applicable Lending Office to the Administrative Agent at the Administrative Agent's Applicable Lending Office, in immediately available funds, such Bank's ratable share of such Revolving Loan Borrowing. Subject to the satisfaction of all applicable conditions precedent, after receipt by the Administrative Agent of such funds, the Administrative Agent shall, by 4:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent), on the date requested for such Revolving Loan Borrowing make such Revolving Loan Borrowing available to the Borrower in immediately available funds at any account of Borrower which is designated in writing by the Borrower to the Administrative Agent.

ii. Unless the Administrative Agent shall have received notice from a Bank before the date of any Revolving Loan Borrowing that such Bank shall not make available to the Administrative Agent such Bank's ratable share of such Revolving Loan Borrowing, the Administrative Agent may assume that such Bank has made its ratable share of such Revolving Loan Borrowing available to the Administrative Agent on the date of such Revolving Loan Borrowing in accordance with paragraph (i) above and the Administrative Agent may, in reliance upon such assumption, make available to the Borrower on such date a corresponding amount. If and to the extent that such Bank shall not have so made its ratable share of such Revolving Loan Borrowing available to the Administrative Agent, such Bank agrees that it shall pay interest on such amount for each day from the date such amount is made available to the Borrower by the Administrative Agent until the date such amount is paid to the Administrative Agent by such Bank at the Federal Funds Rate in effect from time to time, provided that with respect to such Bank if such amount is not paid by such Bank by the end of the second day after the Administrative Agent makes such amount available to the Borrower, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the third day and shall remain at such increased rate thereafter. Interest on such amount shall be due and payable by such Bank upon demand by the Administrative Agent. If such Bank shall pay to the Administrative Agent such amount and interest as provided above, such amount so paid shall constitute such Bank's Revolving Loan Advance as part of such Revolving Loan Borrowing for all purposes of this Agreement even though not made on the same day as the other Revolving Loan Advances comprising such Revolving Loan Borrowing. In the event that such Bank has not repaid such amount by the end

of the fifth day after such amount was made available to the Borrower, the Borrower agrees to repay to the Administrative Agent on demand such amount, together with interest on such amount for each day from the date such amount was made available to the Borrower until the date such amount is repaid to the Administrative Agent at the interest rate charged to the Borrower for such Revolving Loan Borrowing under the terms of this Agreement.

iii. The failure of any Bank to make available its ratable share of any

Revolving Loan Borrowing shall not relieve any other Bank of its obligation, if any, to make available its ratable share of such Revolving Loan Borrowing. No Bank shall be responsible for the failure of any other Bank to honor such other Bank's obligations hereunder, including any failure to make available any funds as part of any Revolving Loan Borrowing.

c. Prepayment.

i. The Borrower may prepay the outstanding principal amount of the Revolving Loan pursuant to written notice given by the Borrower to the Administrative Agent in writing or by telecopy not later than (A) 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed prepayment, in the case of the prepayment of any portion of the Revolving Loan which is comprised of LIBOR Tranches, or (B) 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the same Business Day of the proposed prepayment, in the case of the prepayment of any portion of the Revolving Loan comprised solely of the Prime Rate Tranche. Each such notice shall specify the principal amount and Tranche or Tranches of the Revolving Loan which shall be prepaid, the date of the prepayment, and shall be irrevocable and binding on the Borrower. Prepayments of the Revolving Loan shall be made in integral multiples of \$500,000, in the case of prepayments of any LIBOR Tranches, or \$100,000 in the case of prepayments of the Prime Rate Tranche. If the prepayment would cause the aggregate outstanding principal amount of any LIBOR Tranche comprising all or any part of the Revolving Loan or the aggregate outstanding principal amount of the Prime Rate Tranche comprising all or any part of the Revolving Loan, to be less than \$1,000,000, in the case of any such LIBOR Tranche, or \$1,000,000, in the case of the Prime Rate Tranche, the prepayment must be in an amount equal to the entire outstanding principal amount of such LIBOR Tranche under the Revolving Loan or the entire outstanding principal amount of the Prime Rate Tranche under the Revolving Loan, as the case may be. Upon receipt of any notice of prepayment, the Administrative Agent shall give prompt notice of the intended prepayment to the Banks. For each such notice given by the Borrower, the Borrower shall prepay the Revolving Loan in the specified amount on the specified date as set forth in such notice. The Borrower shall have no right to prepay any principal amount of the Revolving Loan except as provided in this Section 2.2(c)(i).

ii. Upon any Asset Sale or Junior Interests Sale by any Restricted Entity, the Borrower shall use the Prepayment Proceeds remaining after the payment in full of the Term Loan (if required by Section 2.1(c)(ii)) to prepay, on the date of the receipt of such proceeds, (A) first, the principal of the Prime Rate Tranche with respect to the Revolving Loan; (B) second, the principal of and unpaid accrued interest on the LIBOR Tranches of the Revolving Loan, selected in such order as to minimize the costs to be paid by the Borrower pursuant to Section 2.7 (and such costs shall be in addition to the mandatory prepayment due under this Section 2.1(c)(ii)),

and (C) third, the unpaid accrued interest on the Prime Rate Tranche with respect to the Revolving Loan, with any proceeds remaining thereafter to be retained by the Restricted Entities; none of the requirements of Section 2.2(c)(i) with respect to the prepayment of the Revolving Loan shall be

applicable to mandatory prepayments under this Section 2.2(c)(ii).

iii. Each prepayment of principal of any LIBOR Tranche under the Revolving Loan pursuant to this Section 2.2(c) shall be accompanied by payment of all accrued but unpaid interest on the principal amount prepaid and any amounts required to be paid pursuant to Section 2.7 as a result of such prepayment.

d. Repayment.

The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks the aggregate outstanding principal amount of the Revolving Loan on the Revolving Loan Maturity Date.

e. Reduction of Commitments.

The Borrower shall have the right, upon at least three Business Days' irrevocable notice to the Administrative Agent, to terminate in whole or reduce ratably in part the Available Revolving Loan Commitments; provided that each partial reduction shall be in the aggregate amount of \$5,000,000 or in integral multiples of \$5,000,000 in excess thereof. Any reduction or termination of the Revolving Loan Commitments pursuant to this Section 2.2(e) shall be permanent, with no obligation of the Banks to reinstate such Revolving Loan Commitments, and the commitment fees provided for in Section 2.5(a) shall thereafter be computed on the basis of the Revolving Loan Commitments, as so reduced.

2.3 Letter of Credit Facility.

a. Commitment for Letters of Credit.

The Issuing Bank shall, on the terms and conditions set forth in this Agreement and for the purposes set forth in Section 5.4, issue, increase, and extend Letters of Credit at the request of the Borrower from time to time on any Business Day during the period from the date of this Agreement until the Revolving Loan Maturity Date, provided that (1) the aggregate outstanding principal amount of Revolving Loan Borrowings plus the aggregate outstanding principal amount of the Swing Line Loans plus the Letter of Credit Exposure shall not exceed the aggregate amount of the Revolving Loan Commitments then in effect and (2) the Issuing Bank shall not issue any Letter of Credit if it has received unrevoked notice from the Majority Banks that a Default has occurred and the Issuing Bank should not issue Letters of Credit in the future (and if the Issuing Bank issues any letter of credit in contravention of such notice, such letter of credit shall not be a "Letter of Credit" for purposes of this Agreement). No Letter of Credit may have an expiration date later than 12 months after its issuance date, and each Letter of Credit which is self-extending beyond its expiration date must be cancelable upon no more than 30 days notice prior to each extension period given by the Issuing Bank to the beneficiary of such Letter of Credit. No Letter of Credit may have an expiration date later than the date five Business Days before the then-scheduled Revolving Loan Maturity Date. Each Letter of Credit must be in form and substance acceptable to the Issuing Bank. The indebtedness of the Borrower to the Issuing

Bank resulting from Letters of Credit requested by the Borrower shall be evidenced by the Letter of Credit Applications made by the Borrower.

b. Requesting Letters of Credit.

Each Letter of Credit shall be issued, increased, or extended pursuant to a Letter of Credit Application or Letter of Credit Application Amendment, as applicable, given by the Borrower to the Issuing Bank in writing or by telecopy promptly confirmed in writing, such Letter of Credit Application or Letter of Credit Application Amendment being given not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed issuance, increase, or extension of the Letter of Credit. Each Letter of Credit Application or Letter of Credit Application Amendment shall be fully completed and shall specify the information required therein (including the proposed form of the Letter of Credit or change thereto), and shall be irrevocable and binding on the Borrower. Upon receipt by the Issuing Bank of the Letter of Credit Application or Letter of Credit Application Amendment, the Issuing Bank shall give prompt notice thereof to the Administrative Agent, and the Administrative Agent shall promptly inform the Banks of the proposed Letter of Credit or change thereto. Subject to the satisfaction of all applicable conditions precedent, the Issuing Bank shall, by 4:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent), on the date requested by the Borrower for the issuance, increase, or extension of such Letter of Credit issue, increase, or extend such Letter of Credit to the specified beneficiary. Upon the date of the issuance, increase, or extension of a Letter of Credit, the Issuing Bank shall be deemed to have sold to each other Bank and each other Bank shall be deemed to have purchased from the Issuing Bank a ratable participation in the related Letter of Credit or change thereto. The Issuing Bank shall notify the Administrative Agent of each Letter of Credit issued, increased, or extended and the date and amount of each Bank's participation in such Letter of Credit, and the Administrative Agent shall in turn notify the Banks.

c. Reimbursements for Letters of Credit.

i. With respect to any Letter of Credit and in accordance with the related Letter of Credit Application, the Borrower agrees to pay to the Issuing Bank on demand fees due with respect to such Letter of Credit as specified in Section 2.5(b). If the Borrower does not pay upon demand of the Issuing Bank any amount due to the Issuing Bank under any Letter of Credit Application, in addition to any rights the Issuing Bank may have under such Letter of Credit Application, the Issuing Bank may upon written notice to the Administrative Agent request the satisfaction of such obligation by the making of a Revolving Loan Borrowing. Concurrently with such notice to the Administrative Agent, the Issuing Bank will use reasonable efforts to provide like notice to the Borrower, provided that failure to provide such notice to the Borrower at such time shall not invalidate the effectiveness of such request for a Revolving Loan Borrowing. Upon such request, the Borrower shall be deemed to have requested the making of a Revolving Loan Borrowing in the amount of such obligation and the transfer of the proceeds thereof to the Issuing Bank. Such Revolving Loan Borrowing shall be a Prime Rate Borrowing. The Administrative Agent shall promptly forward notice of such Revolving Loan Borrowing to the Borrower and the Banks, and each Bank

shall, in accordance with the procedures of Section 2.2(b), other than limitations on the size of Revolving Loan Borrowings, and notwithstanding the failure of any conditions precedent, make available such Bank's ratable share of such Revolving Loan Borrowing to the Administrative Agent, and the Administrative Agent shall promptly

deliver the proceeds thereof to the Issuing Bank for application to such Bank's ratable share of the obligations under such Letter of Credit. The Borrower hereby unconditionally and irrevocably authorizes, empowers, and directs the Issuing Bank to make such requests for Revolving Loan Borrowings on behalf of the Borrower, and the Banks to make Revolving Loan Advances to the Administrative Agent for the benefit of the Issuing Bank in satisfaction of such obligations. The Administrative Agent and each Bank may record and otherwise treat the making of such Revolving Loan Borrowings as the making of Revolving Loan Borrowings to the Borrower under this Agreement as if requested by the Borrower. Nothing herein is intended to release the Borrower's obligations under any Letter of Credit Application, but only to provide an additional method of payment therefor. The making of any Revolving Loan Borrowing under this Section 2.3(c) shall not constitute a cure or waiver of any Default or Event of Default, other than the payment Default or Event of Default which is satisfied by the application of the amounts deemed advanced hereunder, caused by the Borrower's failure to comply with the provisions of this Agreement or any Letter of Credit Application.

ii. If prior to the time a Revolving Loan Advance would have otherwise been made pursuant to Section 2.3(c)(i), a Bankruptcy Event of Default shall have occurred and be continuing with respect to the Borrower or if for any other reason, as determined by the Issuing Bank in its sole discretion, Revolving Loan Advances may not be made as contemplated by Section 2.3(c)(i), each Bank shall, on the date such Revolving Loan Advance was to have been made pursuant to the notice referred to in Section 2.3(c)(i), pay to the Issuing Bank such Bank's ratable share of the obligations under such Letter of Credit.

iii. Whenever, at any time after the Issuing Bank has made payment under any Letter of Credit and has received from any Bank its ratable share of such payment in accordance with this Section 2.3(c), the Issuing Bank receives any payment related to such Letter of Credit (whether directly from the Borrower or otherwise, including proceeds of Collateral applied thereto by the Issuing Bank), or any payment of interest on account thereof, the Issuing Bank will distribute to such Bank its ratable share thereof; provided, however, that in the event that any such payment received by the Issuing Bank shall be required to be returned by the Issuing Bank, such Bank shall return to the Issuing Bank the portion thereof previously distributed by the Issuing Bank to it.

iv. Each Bank's obligation to make the Revolving Loan Advances referred to in Section 2.3(c)(i) and to purchase participating interests pursuant to Section 2.3(b) shall be absolute and unconditional and shall not be affected by any circumstance, including (a) any setoff, counterclaim, recoupment, defense or other right that such Bank or the Borrower may have against the Issuing Bank, the Borrower or any other Person for any reason whatsoever; (b) the occurrence or continuance of a Default or an Event of Default or the failure to satisfy any of the other conditions specified in Section 3; (c) any adverse change in the condition (financial or otherwise) of the Borrower; (d) any breach of this Agreement or any other Credit Document by the Borrower, any other Credit Party or any other Bank; or (e) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing.

d. Prepayments of Letters of Credit.

In the event that any Letters of Credit shall be outstanding according to their terms after the Revolving Loan Maturity Date, the Borrower shall pay to the Administrative Agent an amount

equal to the Letter of Credit Exposure allocable to such Letters of Credit to be held in the Letter of Credit Collateral Account and applied in accordance with paragraph (g) below.

### e. Obligations Unconditional.

The obligations of the Borrower and each Bank under this Agreement and the Letter of Credit Applications to make payments as required to reimburse the Issuing Bank for draws under Letters of Credit and to make other payments due in respect of Letters of Credit shall be unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement and the Letter of Credit Applications under all circumstances, including: (i) any lack of validity or enforceability of any Letter of Credit Document; (ii) any amendment, waiver, or consent to departure from any Letter of Credit Document; (iii) the existence of any claim, set off, defense, or other right which the Borrower or any Bank may have at any time against any beneficiary or transferee of any Letter of Credit (or any Persons for whom any such beneficiary or any such transferee may be acting), the Issuing Bank, or any other person or entity, whether in connection with the transactions contemplated in this Agreement or any unrelated transaction; (iv) any statement or any other document presented under such Letter of Credit proving to be forged, fraudulent, invalid, or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or (v) payment by the Issuing Bank under any Letter of Credit against presentation of a draft or certificate which does not comply with the terms of such Letter of Credit; provided, however, that nothing contained in this paragraph (e) shall be deemed to constitute a waiver of any remedies of the Borrower or any Bank in connection with the Letters of Credit or the Borrower's or such Bank's rights under paragraph (f) below.

## f. Liability of Issuing Bank.

The Issuing Bank shall not be liable or responsible for: (i) the use which may be made of any Letter of Credit or any acts or omissions of any beneficiary or transferee in connection therewith; (ii) the validity, sufficiency, or genuineness of documents related to Letters of Credit, or of any endorsement thereon, even if such documents should prove to be in any or all respects invalid, insufficient, fraudulent, or forged; (iii) payment by the Issuing Bank against presentation of documents which do not strictly comply with the terms of a Letter of Credit, including failure of any documents to bear any reference or adequate reference to the relevant Letter of Credit; or (iv) any other circumstances whatsoever in making or failing to make payment under any Letter of Credit (INCLUDING THE ISSUING BANK'S OWN NEGLIGENCE); except that the Issuing Bank shall be liable to the Borrower or any Bank to the extent of any direct, as opposed to consequential, damages suffered by the Borrower or such Bank which the Borrower or such Bank proves were caused by (A) the Issuing Bank's gross negligence or willful misconduct in determining whether documents presented under a Letter of Credit comply with the terms of such Letter of Credit, (B) the Issuing Bank's willful failure to make or delay in making lawful payment under any Letter of Credit after the presentation to it of documentation strictly complying with the terms and conditions of such Letter of Credit or the Issuing Bank's payment of greater than the maximum amount permitted under any Letter of Credit, or (C) the Issuing Bank's negligence in the handling of money.

g. Letter of Credit Collateral Account.

i. If the Borrower is required to deposit funds in the Letter of Credit Collateral Account pursuant to Section 2.3(d) or 6.4, then the Borrower and the Administrative Agent shall

establish the Letter of Credit Collateral Account, and the Borrower shall execute any documents and agreements, including the Administrative Agent's standard form assignment of deposit accounts, that the Administrative Agent reasonably requests in connection therewith to establish the Letter of Credit Collateral Account and grant the Administrative Agent a first priority security interest in such account and the funds therein. The Borrower hereby pledges to the Administrative Agent and grants the Administrative Agent a security interest in the Letter of Credit Collateral Account, whenever established, all funds held in the Letter of Credit Collateral Account from time to time, and all proceeds thereof as security for the payment of the Credit Obligations.

ii. Funds held in the Letter of Credit Collateral Account shall be held as cash collateral for obligations with respect to Letters of Credit and promptly applied by the Administrative Agent at the request of the Issuing Bank to any reimbursement or other obligations under Letters of Credit that exist or occur. To the extent that any surplus funds are held in the Letter of Credit Collateral Account above the Letter of Credit Exposure, during the existence of an Event of Default the Administrative Agent may (A) hold such surplus funds in the Letter of Credit Collateral Account as cash collateral for the Credit Obligations or (B) apply such surplus funds to any Credit Obligations in accordance with Section 6.9. If no Default exists, the Administrative Agent shall release to the Borrower at the Borrower's written request any funds held in the Letter of Credit Collateral Account above the amounts required by Section 2.3(d).

iii. Funds held in the Letter of Credit Collateral Account shall be invested in money market funds of the Administrative Agent or in another investment if mutually agreed upon by the Borrower and the Administrative Agent, but the Administrative Agent shall have no other obligation to make any other investment of the funds therein. The Administrative Agent shall exercise reasonable care in the custody and preservation of any funds held in the Letter of Credit Collateral Account and shall be deemed to have exercised such care if such funds are accorded treatment substantially equivalent to that which the Administrative Agent accords its own property, it being understood that the Administrative Agent shall not have any responsibility for taking any necessary steps to preserve rights against any parties with respect to any such funds.

2.4 Swing Line Facility.

a. Commitment.

i. Subject to the terms and conditions hereof, the Swing Line Lender agrees to make a portion of the credit otherwise available to the Borrower under the Revolving Loan Commitments from time to time during the period from the date of this Agreement until the Revolving Loan Maturity Date by making swing line loans ("Swing Line Loans") to the Borrower; provided that (1) the aggregate principal amount of Swing Line Loans outstanding at any time shall not exceed the Swing Line Commitment then in effect (notwithstanding that the Swing Line Loans outstanding at any time, when aggregated with the Swing Line Lender's other outstanding Revolving Loan Advances, may exceed the Swing Line Commitment then in effect); (2) the Borrower shall not request, and the Swing Line Lender shall not make, any Swing Line Loan if, after giving effect to the making of such Swing Line Loan, the aggregate amount of the Available Revolving Loan Commitments would be less than zero, and (3) the Swing Line Lender shall not make any Swing Line Loan if it has received unrevoked notice from the Majority Banks

stating that a Default has occurred and the Swing Line Lender should not make future Swing Line Loans (and if the Swing Line Lender makes any loan in contravention of such notice, such loan shall not be a "Swing Line Loan" for purposes of this Agreement). Within the limits expressed in this Agreement, the Borrower may use the Swing Line Commitment by borrowing, repaying and reborrowing, all in accordance with the terms and conditions hereof. Each Swing Line Loan shall be a Prime Rate Borrowing. The indebtedness of the Borrower to the Swing Line Lender resulting from the Swing Line Loans made by the Swing Line Lender shall be evidenced by the Swing Line Note made by the Borrower.

ii. The Borrower shall repay the principal of and accrued interest on each Swing Line Loan ten Business Days after the making of such Swing Line Loan, and the Borrower shall repay all principal of and accrued interest on the outstanding Swing Line Loans on the Revolving Loan Maturity Date.

b. Procedure for Swing Line Borrowing.

Whenever the Borrower desires that the Swing Line Lender make Swing Line Loans it shall give the Swing Line Lender irrevocable telephonic notice confirmed promptly in writing (which telephonic notice must be received by the Swing Line Lender not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the proposed date of the Swing Line Borrowing), specifying (i) the amount to be borrowed and (ii) the requested date of the Swing Line Borrowing (which shall be a Business Day during the period from the date of this Agreement until the Revolving Loan Maturity Date). Each borrowing under the Swing Line Commitment shall be in an amount equal to \$500,000 or a whole multiple of \$100,000 in excess thereof. Not later than 3:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the date requested for a Swing Line Borrowing specified in a notice in respect of Swing Line Loans, the Swing Line Lender shall make available to the Administrative Agent at the Applicable Lending Office an amount in immediately available funds equal to the amount of the Swing Line Loan to be made by the Swing Line Lender. The Administrative Agent shall make the proceeds of such Swing Line Loan available to the Borrower on such date in immediately available funds at any account of Borrower which is designated in writing by the Borrower to the Administrative Agent.

c. Reimbursements for Swing Line Loan Obligations.

i. The Swing Line Lender, at any time and from time to time in its sole and absolute discretion may, on behalf of the Borrower (which hereby irrevocably directs the Swing Line Lender to act on its behalf), on one Business Day's notice given by the Swing Line Lender no later than 12:00 noon (local time at the Applicable Lending Office of the Administrative Agent) request each Bank to make, and each Bank hereby agrees to make, a Revolving Loan Advance, in an amount equal to such Bank's Revolving Loan Percentage of the aggregate amount of the Swing Line Loans (the "Refunded Swing Line Loans") outstanding on the date of such notice, to repay the Swing Line Lender. Upon such request, the Borrower shall be deemed to have requested the making of a Revolving Loan Borrowing in the amount of such request. Such Revolving Loan Borrowing shall be a Prime Rate Borrowing. Each Bank shall make the amount of its Revolving Loan Advance available to the Administrative Agent at the Administrative Agent's Applicable Lending Office in immediately available funds, not later than 10:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) one Business Day after the date of such notice. The proceeds of such Revolving Loan Borrowing shall be

immediately made available by the Administrative Agent to the Swing Line Lender for application by the Swing Line Lender to the repayment of the Refunded Swing Line Loans. The Borrower irrevocably authorizes the Swing Line Lender to charge the Borrower's accounts with the Administrative Agent (up to the amount available in each such account) in order to immediately pay the amount of such Refunded Swing Line Loans to the extent amounts received from the Banks are not sufficient to repay in full such Refunded Swing Line Loans.

ii. If prior to the time a Revolving Loan Advance would have otherwise been made pursuant to Section 2.4(c)(i), a Bankruptcy Event of Default shall have occurred and be continuing with respect to the Borrower or if for any other reason, as determined by the Swing Line Lender in its sole discretion, Revolving Loan Advances may not be made as contemplated by Section 2.4(c)(i), each Bank shall, on the date such Revolving Loan Advance was to have been made pursuant to the notice referred to in Section 2.4(c)(i) (the "Refunding Date"), purchase for cash an undivided participating interest in the then-outstanding Swing Line Loans by paying to the Swing Line Lender an amount (the "Swing Line Participation Amount") equal to (a) such Bank's Revolving Loan Percentage times (b) the sum of the aggregate principal amount of Swing Line Loans then outstanding that were to have been repaid with such Revolving Loan Borrowing.

iii. Whenever, at any time after the Swing Line Lender has received from any Bank such Bank's Swing Line Participation Amount, the Swing Line Lender receives any payment on account of the Swing Line Loans, the Swing Line Lender will distribute to such Bank its Swing Line Participation Amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Bank's participating interest was outstanding and funded and, in the case of principal and interest payments, to reflect such Bank's pro rata portion of such payment if such payment is not sufficient to pay the principal of and interest on all Swing Line Loans then due); provided, however, that in the event that such payment received by the Swing Line Lender is required to be returned, such Bank will return to the Swing Line Lender any portion thereof previously distributed to it by the Swing Line Lender.

iv. Each Bank's obligation to make the Revolving Loan Advances referred to in Section 2.4(c)(i) and to purchase participating interests pursuant to Section 2.4(c)(ii) shall be absolute and unconditional and shall not be affected by any circumstance, including (a) any setoff, counterclaim, recoupment, defense or other right that such Bank or the Borrower may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever; (b) the occurrence or continuance of a Default or an Event of Default or the failure to satisfy any of the other conditions specified in Section 3; (c) any adverse change in the condition (financial or otherwise) of the Borrower; (d) any breach of this Agreement or any other Credit Document by the Borrower, any other Credit Party or any other Bank; or (e) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing.

- 2.5 Fees.
- a. Commitment Fees.

The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks an unused commitment fee of the Commitment Fee Rate per annum on the average daily amount by which (i) the aggregate amount of the Revolving Loan Commitments exceeds (ii) the outstanding

principal amount of the Revolving Loan plus the Letter of Credit Exposure. The unused commitment fee shall accrue from the date of this Agreement and be due and payable in arrears on the last day of each calendar quarter and on the Revolving Loan Maturity Date.

b. Fees for Letters of Credit.

For each Letter of Credit issued by the Issuing Bank, the Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks a letter of credit fee equal to the Applicable Margin for letter of credit fees per annum on the face amount of such Letter of Credit for the stated term of such Letter of Credit, with a minimum fee of \$500. In addition, as a condition to the issuance of each Letter of Credit by the Issuing Bank, the Borrower shall pay to the Administrative Agent for the benefit of the Issuing Bank a fronting fee of 0.125% on the face amount of such Letter of Credit, with a minimum fee of \$500. Furthermore, the Borrower shall pay to the Administrative Agent for the benefit of the Issuing Bank any documentary and processing fees in accordance with the Issuing Bank's standard schedule from time to time in effect with respect to the issuance, amendment, cancellation, negotiation or transfer of each Letter of Credit and each drawing made thereunder. The Borrower shall pay each such letter of credit fee for each Letter of Credit quarterly in arrears within ten days after when billed therefor by the Issuing Bank.

2.6 Interest.

a. Election of Interest Rate Basis.

The Borrower may select the interest rate basis for the Term Loan and the Revolving Loan in accordance with the terms of this Section 2.6(a):

i. Under the Revolving Loan Borrowing Request provided to the Administrative Agent in connection with the making of each Revolving Loan Borrowing, the Borrower shall select the amount and the Type of the Tranches, and for each LIBOR Tranche selected, any permitted Interest Period for each such LIBOR Tranche, which will comprise such Revolving Loan Borrowing, provided that (A) at no time shall there be more than ten separate LIBOR Tranches outstanding and (B) each LIBOR Tranche must be in a principal amount equal to or greater than \$1,000,000 and be made in multiples of \$500,000, and the Prime Rate Tranche must be in a principal amount equal to or greater than \$1,000,000 and be made in multiples of \$100,000. Such interest rate elections must be provided to the Administrative Agent in writing or by telecopy not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of any proposed Revolving Loan Borrowing comprised of a LIBOR Tranche or 11:00 a.m. (local time at the Applicable Lending Office of the Administrative Agent) on the same day of any proposed Revolving Loan Borrowing which is a Prime Rate Borrowing. The Administrative Agent shall promptly forward copies of such interest rate elections to the Banks. In the case of any Revolving Loan Borrowing comprised of a LIBOR Tranche, upon determination by the Administrative Agent, the Administrative Agent shall promptly notify the Borrower and the Banks of the applicable interest rate for such Tranche.

ii. With respect to any Tranche, the Borrower may continue or convert any portion of any LIBOR Tranche or the Prime Rate Tranche to form new LIBOR Tranches or increase or decrease the amount of the Prime Rate Tranche in accordance with this paragraph. Each such

continuation or conversion shall be deemed to create a new LIBOR Tranche or increase or decrease the amount of the Prime Rate Tranche, as applicable, for all purposes of this Agreement. Each such continuation or conversion shall be made pursuant to a Continuation/Conversion Request given by the Borrower to the Administrative Agent in writing or by telecopy not later than 1:00 p.m. (local time at the Applicable Lending Office of the Administrative Agent) on the third Business Day before the date of the proposed continuation or conversion. Each Continuation/Conversion Request shall be fully completed and shall specify the

information required therein, and shall be irrevocable and binding on the Borrower. The Administrative Agent shall promptly forward notice of the continuation or conversion to the Banks. In the case of any continuation or conversion into LIBOR Tranches, upon determination by the Administrative Agent, the Administrative Agent shall notify the Borrower and the Banks of the applicable interest rate. Continuations and conversions of LIBOR Tranches shall be made in integral multiples of \$500,000, and continuations and conversions of the Prime Rate Tranche shall be made in integral multiples of \$100,000 (provided that the Prime Rate Tranche with respect to the Term Loan may be of any size). No continuation or conversion shall be permitted if such continuation or conversion would cause the aggregate outstanding principal amount of any LIBOR Tranche which would remain outstanding to be less than \$1,000,000, or the aggregate outstanding principal amount of the Prime Rate Tranche which would remain outstanding to be less than \$1,000,000 (provided that the Prime Rate Tranche with respect to the Term Loan may be of any size). At no time shall there be more than ten separate LIBOR Tranches outstanding. Any conversion of an existing LIBOR Tranche is subject to Section 2.7. Subject to the satisfaction of all applicable conditions precedent, the Administrative Agent and the Banks shall before close of business on the date requested by the Borrower for the continuation or conversion, make such continuation or conversion.

iii. At the end of the Interest Period for any LIBOR Tranche if the Borrower has not continued or converted such LIBOR Tranche into new Tranches as provided for in paragraph (ii) above, the Borrower shall be deemed to have continued such LIBOR Tranche as a new LIBOR Tranche with an Interest Period of one month. All of the Prime Rate Tranche shall continue as the Prime Rate Tranche unless the Borrower converts such Prime Rate Tranche as provided for in paragraph (ii) above.

### b. LIBOR Tranches.

Each LIBOR Tranche with respect to the Revolving Loan shall bear interest during its Interest Period at a per annum interest rate equal to the sum of the LIBOR for such Tranche plus the Applicable Margin for LIBOR Tranches in effect from time to time. Each LIBOR Tranche with respect to the Term Loan shall bear interest during its Interest Period at a per annum interest rate equal to the sum of the LIBOR for such Tranche plus 0.25% plus the Applicable Margin for LIBOR Tranches in effect from time to time. The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks all accrued but unpaid interest on each LIBOR Tranche on the last day of the applicable Interest Period for such LIBOR Tranche (and with respect to LIBOR Tranches with Interest Periods of greater than three months, on the date which is three months after the first day of the Interest Period for such LIBOR Tranche), when required upon prepayment as specified elsewhere in this Agreement, on any date when such LIBOR Tranche is prepaid or repaid in full, and on the Revolving Loan Maturity Date.

c. Prime Rate Tranche.

The Prime Rate Tranche shall bear interest at a per annum interest rate equal to the Base Rate in effect from time to time plus the Applicable Margin for the Prime Rate Tranche in effect from time to time. The Borrower shall pay to the Administrative Agent for the ratable benefit of the Banks all accrued but unpaid interest on the aggregate outstanding principal amount of the Prime Rate Tranche on the last day of each calendar quarter, when required upon prepayment as specified elsewhere in this Agreement, on any date the Prime Rate Tranche is prepaid or repaid in full, and on the Revolving Loan Maturity Date.

d. Usury Protection.

i. If, with respect to any Bank and the Borrower, the effective rate of interest contracted for by such Bank with the Borrower under the Credit Documents, including the stated rates of interest contracted for hereunder and any other amounts contracted for under the Credit Documents which are deemed to be interest, at any time exceeds the Highest Lawful Rate, then the outstanding principal amount of the loans made by such Bank to the Borrower hereunder shall bear interest at a rate which would make the effective rate of interest on the loans made by such Bank to the Borrower under the Credit Documents equal the Highest Lawful Rate until the difference between the amounts which would have been due by the Borrower to such Bank at the stated rates and the amounts which were due by the Borrower to such Bank at the Highest Lawful Rate (the "Lost Interest") has been recaptured by such Bank. If, when the loans made hereunder are repaid in full, the Lost Interest has not been fully recaptured by such Bank pursuant to the preceding sentence, then, to the extent permitted by law, the interest rates charged by such Bank to the Borrower hereunder shall be retroactively increased such that the effective rate of interest on the loans made by such Bank to the Borrower under the Credit Documents was at the Highest Lawful Rate since the effectiveness of this Agreement to the extent necessary to recapture the Lost Interest not recaptured pursuant to the preceding sentence and, to the extent allowed by law, the Borrower shall pay to such Bank the amount of the Lost Interest remaining to be recaptured by such Bank.

ii. In calculating all sums paid or agreed to be paid to any Bank by the Borrower for the use, forbearance, or detention of money under the Credit Documents, (x) such amounts shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread in equal parts throughout the term of the Credit Documents; (y) the parties shall, to the maximum extent

not prohibited by applicable law, characterize any nonprincipal payment as an expense, fee or premium rather than interest, and (z) the parties shall exclude voluntary prepayments and the effects thereof.

iii. The parties hereto expressly agree, pursuant to Chapter 346 ("Chapter 346") of the Texas Credit Code, that Chapter 346 (which relates to open-end line of credit revolving loan accounts) shall not apply to any Credit Document or any Credit Obligation, and that none of the same shall be governed by Chapter 346 or subject to its provisions in any manner whatsoever.

iv. NOTWITHSTANDING THE FOREGOING OR ANY OTHER TERM IN THIS AGREEMENT AND THE CREDIT DOCUMENTS TO THE CONTRARY, it is the intention of each Bank and the Borrower to conform strictly to any applicable usury laws. Accordingly, if any Bank contracts for, charges, or receives any consideration from the Borrower which constitutes interest in excess of the Highest Lawful Rate, then any such excess shall be canceled automatically and, if previously paid, shall at such Bank's option be applied to the outstanding amount of the loans made hereunder by such Bank to the Borrower or be refunded to the Borrower.

2.7 Breakage Costs.

If (i) any payment of principal on or any conversion of any LIBOR Tranche is made on any date other than the last day of the Interest Period for such LIBOR Tranche, whether as a result of any voluntary or mandatory prepayment (other than a prepayment upon the occurrence of any event subject to Section 2.9 or 2.10), any acceleration of maturity, or any other cause, (ii) any payment of principal on any LIBOR Tranche is not made when due, or (iii) any LIBOR Tranche is not borrowed, converted, or prepaid in accordance with the respective notice thereof provided by the Borrower to the Administrative Agent, whether as a result of any failure to meet any applicable conditions precedent for borrowing, conversion, or prepayment, the permitted cancellation of any request for borrowing, conversion, or prepayment, the failure of the Borrower to provide the respective notice of borrowing, conversion, or prepayment, or any other cause not specified above which is created by the Borrower, then the Borrower shall pay to each Bank upon demand any amounts required to compensate such Bank for any losses, costs, or expenses, including lost profits and administrative expenses, which are reasonably allocable to such action, including losses, costs, and expenses related to the liquidation or redeployment of funds acquired or designated by such Bank to fund or maintain such Bank's ratable share of such LIBOR Tranche or related to the reacquisition or redesignation of funds by such Bank to fund or maintain such Bank's ratable share of such LIBOR Tranche following any liquidation or redeployment of such funds caused by such action. Such Bank need not prove matched funding of any particular funds, and a certificate as to the amount of such loss, cost, or expense detailing the calculation thereof and certifying that such Bank customarily charges such amounts to its other customers in similar circumstances submitted by such Bank to the Borrower shall be conclusive and binding for all purposes, absent manifest error. Notwithstanding anything herein to the contrary, until the date 180 days after the Closing Date, the Borrower shall reimburse each Bank for all losses, costs and expenses of the type described in this Section 2.7 incurred by such Bank in connection with the syndication of the Facilities by the Administrative Agent in its sole discretion and the addition of new Banks.

### 2.8 Increased Costs.

a. Cost of Funds.

If due to either (i) any introduction of, change in, or change in the interpretation of, any law or regulation, in each case, after the date of this Agreement or (ii) compliance with any guideline or request from any central bank or other governmental authority having appropriate jurisdiction (whether or not having the force of law) given after the date of this Agreement, there shall be any increase in the costs of any Bank attributable to (x) committing to make any Term Loan Advance or Revolving Loan Advance or obtaining funds for the making, funding, or maintaining of such Bank's ratable share of any LIBOR Tranche in the relevant interbank market or (y) committing to make Letters of Credit or

issuing, funding, or maintaining Letters of Credit (including any increase in any applicable reserve requirement specified by the Board, including those for emergency, marginal, supplemental, or other reserves), then the Borrower shall pay to such Bank upon demand any amounts required to compensate such Bank for such increased costs, such amounts being due and payable upon demand by such Bank. A certificate as to the cause and amount of such increased cost detailing the calculation of such cost and certifying that such Bank customarily charges such amounts to its other customers in similar circumstances submitted by such Bank to the Borrower shall be conclusive and binding for all purposes, absent manifest

error. No Bank may make any claim for compensation under this Section 2.8(a) for increased costs incurred more than 90 days prior to the delivery of any such certificate.

## b. Capital Adequacy.

If, due to either (i) any introduction of, change in, or change in the interpretation of, any law or regulation, in each case, after the date of this Agreement or (ii) compliance with any guideline or request from any central bank or other governmental authority having appropriate jurisdiction (whether or not having the force of law) given after the date of this Agreement, there shall be any increase in the capital requirements of any Bank or its parent or holding company attributable to (x) committing to make Term Loan Advances or Revolving Loan Advances or making, funding, or maintaining Term Loan Advances or Revolving Loan Advances or (y) committing to make Letters of Credit or issuing, funding, or maintaining Letters of Credit, as such capital requirements are allocated by such Bank, then the Borrower shall pay to such Bank upon demand any amounts required to compensate such Bank or its parent or holding company for such increase in costs (including an amount equal to any reduction in the rate of return on assets or equity of such Bank or its parent or holding company), such amounts being due and payable upon demand by such Bank. A certificate as to the cause and amounts detailing the calculation of such amounts and certifying that such Bank customarily charges such amounts to its other customers in similar circumstances submitted by such Bank to the Borrower shall be conclusive and binding for all purposes, absent manifest error. No Bank may make any claim for compensation under this Section 2.8(b) for increased costs incurred more than 90 days prior to the delivery of any such certificate.

## 2.9 Illegality.

Notwithstanding any other provision in this Agreement, if it becomes unlawful for any Bank to obtain deposits or other funds for making or funding such Bank's ratable share of any LIBOR Tranche in the relevant interbank market, such Bank shall so notify the Borrower and the Administrative Agent, such Bank's commitment to create LIBOR Tranches shall be suspended until such condition has passed, all LIBOR Tranches applicable to such Bank shall be converted to the Prime Rate Tranche as of the end of each applicable Interest Period or earlier if necessary, and all subsequent requests for LIBOR Tranches shall be deemed to be requests for Prime Rate Borrowings or continuations and conversions of the Prime Rate Tranche, as applicable, with respect to such Bank.

#### 2.10 Market Failure.

Notwithstanding any other provision in this Agreement, if the Administrative Agent determines that: (a) quotations of interest rates for the relevant deposits referred to in the definition of "LIBOR" are not being provided in the relevant amounts, or maturities for purposes of determining the rate of interest referred to in the definition of "LIBOR" or (b) the relevant rates of interest referred to in the definition of "LIBOR" which are used as the basis to determine the rate of interest for LIBOR Tranches will not adequately cover the cost to any Bank of making or maintaining such Bank's ratable share of any LIBOR Tranche, then if the Administrative Agent so notifies the Borrower, the Banks' commitments to create LIBOR Tranches shall be suspended

until such condition has passed, all LIBOR Tranches shall be converted to the Prime Rate Tranche as of the end of each applicable Interest Period or earlier if necessary, and all subsequent requests for LIBOR Tranches shall be deemed to be requests for Prime Rate Borrowings or continuations and conversions of the Prime Rate Tranche, as applicable, with respect to such Bank.

2.11 Payment Procedures and Computations.

a. Payment Procedures.

Time is of the essence in this Agreement and the Credit Documents. All payments hereunder shall be made in Dollars. The Borrower shall make each payment under this Agreement and under the Term Loan Notes, the Revolving Loan Notes and the Swing Line Note not later than 12:00 noon (local time at the Applicable Lending Office of the Administrative Agent) on the day when due to the Administrative Agent at the Administrative Agent's Applicable Lending Office in immediately available funds. All payments by the Borrower hereunder shall be made without any offset, abatement, withholding, deduction, counterclaim, or reduction. Upon receipt of payment from the Borrower of any principal, interest, or fees due to the Banks, the Administrative Agent shall promptly after receipt thereof distribute to the Banks their ratable share of such payments for the account of their respective Applicable Lending Offices. If and to the extent that the Administrative Agent shall not have so distributed to any Bank its ratable share of such payments, the Administrative Agent agrees that it shall pay interest on such amount for each day after the day when such amount is made available to the Administrative Agent by the Borrower until the date such amount is paid to such Bank by the Administrative Agent at the Federal Funds Rate in effect from time to time, provided that if such amount is not paid by the Administrative Agent by the end of the third day after the Borrower makes such amount available to the Administrative Agent, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the fourth day and shall remain at such increased rate thereafter. Interest on such amount shall be due and payable by the Administrative Agent upon demand by such Bank. Upon receipt of other amounts due solely to the Administrative Agent, the Issuing Bank, the Swing Line Lender, or a specific Bank, the Administrative Agent shall distribute such amounts to the appropriate party to be applied in accordance with the terms of this Agreement.

b. Administrative Agent Reliance.

Unless the Administrative Agent shall have received written notice from the Borrower prior to any date on which any payment is due to the Banks that the Borrower shall not make such payment in full, the Administrative Agent may assume that the Borrower has made such payment in full to the Administrative Agent on such date, and the Administrative Agent may, in reliance upon such assumption, cause to be distributed to each Bank on such date an amount equal to the amount then due such Bank. If and to the extent the Borrower shall not have so made such payment in full to the Administrative Agent, each Bank shall repay to the Administrative Agent forthwith on demand such amount distributed to such Bank, together with interest thereon from the date such amount is distributed to such Bank until the date such Bank repays such amount to the Administrative Agent, at an interest rate equal to, the Federal Funds Rate in effect from time to time, provided that with respect to such Bank, if such amount is not repaid by such Bank by the end of the second day after the date of the Administrative Agent's

demand, the interest rates specified above shall be increased by a per annum amount equal to 2.00% on the third day after the date of the Administrative Agent's demand and shall remain at such increased rate thereafter.

c. Sharing of Payments.

Each Bank agrees that if it should receive any payment (whether by voluntary payment, by realization upon security, by the exercise of the right of setoff or banker's lien, by counterclaim or cross action, by the enforcement of any right under the Credit Documents, or otherwise) in respect of any obligation of the Borrower to pay principal, interest, fees, or any other obligation incurred under the Credit Documents in a proportion greater than the total amount of such principal, interest, fees, or other obligation then owed and due by the Borrower to such Bank bears to the total amount of principal, interest, fees, or other obligation then owed and due by the Borrower to all of the Banks immediately prior to such receipt, then such Bank receiving such excess payment shall purchase for cash without recourse from the other Banks an interest in the obligations of the Borrower to such Banks in such amount as shall result in a participation by all of the Banks, in proportion with the Banks' respective pro rata shares, in the aggregate unpaid amount of principal, interest, fees, or any such other obligation, as the case may be, owed by the Borrower to all of the Banks; provided that if all or any portion of such excess payment is thereafter recovered from such Bank, such purchase shall be rescinded and the purchase price restored to the extent of such recovery, in proportion with the Banks' respective pro rata shares, but without interest.

d. Authority to Charge Accounts.

The Administrative Agent, if and to the extent payment owed to the Administrative Agent or any Bank is not made when due, may charge from time to time against any account of the Borrower with the Administrative Agent any amount so due. The Administrative Agent agrees promptly to notify the Borrower after any such charge and application made by the Administrative Agent, provided that the failure to give such notice shall not affect the validity of such charge and application.

e. Computation Methods.

Unless expressly provided for in this Agreement, (i) all computations of interest based on the Prime Rate (including the Base Rate, when applicable) shall be made on the basis of a 365/366 day year, as the case may be, (ii) all computations of interest based on the Federal Funds Rate (including the Base Rate, when applicable) shall be made on the basis of a 360 day year, (iii) all computations of interest based upon the LIBOR shall be made on the basis of a 360 day year, and (iv) all computations of fees shall be made on the basis of a 360 day year, in each case for the actual number of days (including the first day, but excluding the last day) occurring in the period for which such interest or fees are payable. Each determination by the Administrative Agent of an interest rate or fee shall be conclusive and binding for all purposes, absent manifest error.

#### f. Payment Dates.

Whenever any payment shall be stated to be due on a day other than a Business Day, such payment shall be made on the next succeeding Business Day, and such extension of time shall in such case be included in the computation of payment of interest or fees, as the case may be. If

the time for payment for an amount payable is not specified in this Agreement or in any other Credit Document, the payment shall be due and payable on demand by the Administrative Agent or the applicable Bank.

2.12 Taxes.

a. No Deduction for Certain Taxes.

Any and all payments by the Borrower shall be made free and clear of and without deduction for any and all present or future taxes, levies, imposts, deductions, charges, or withholdings, and all liabilities with respect thereto, other than taxes imposed on the income of, and franchise taxes imposed in lieu of net income taxes on, the Administrative Agent, any Bank, or the Applicable Lending Office thereof by any jurisdiction in which any such entity is a present or former citizen or resident or any political subdivision of such jurisdiction (other than any such connection arising solely from the Administrative Agent, such Bank, or the Applicable Lending Office having executed, delivered or performed its obligations or received a payment under, or enforced, this Agreement or any other Letter of Credit Document) (all such nonexcluded taxes, levies, imposts, deductions, charges, withholdings, and liabilities being hereinafter referred to as "Taxes"). If the Borrower shall be required by law to deduct any Taxes or Other Taxes from or in respect of any sum payable to the Administrative Agent, any Bank, or the Applicable Lending Office thereof, (i) the sum payable shall be increased as may be necessary so that, after making all required deductions (including deductions applicable to additional sums payable

under this Section 2.12), such Person receives an amount equal to the sum it would have received had no such deductions been made; (ii) the Borrower shall make such deductions; and (iii) the Borrower shall pay the full amount deducted to the relevant taxation authority or other authority in accordance with applicable law.

b. Other Taxes.

The Borrower agrees to pay any present or future stamp or documentary taxes or any other excise or property taxes, charges, or similar levies ("Other Taxes") which arise from any payment made or from the execution, delivery, or registration of, or otherwise with respect to, this Agreement or the other Credit Documents (other than those which become due as a result of any Bank joining this Agreement as a result of any Assignment and Acceptance, which shall be paid by the Bank which becomes a Bank hereunder as a result of such Assignment and Acceptance).

c. Foreign Bank Withholding Exemption.

Each Bank and Issuing Bank that is not a "US Person" as defined in Section 7701(a)(30) of the Code agrees that it shall deliver to the Borrower and the Administrative Agent (i) two duly completed copies of United States Internal Revenue Service Form W-8BEN, W-8ECI or other applicable form, as the case may be, certifying in each case that such Bank is entitled to receive payments under this Agreement, the Term Loan Notes and the Revolving Loan Notes payable to it, without deduction or withholding of any United States federal income taxes, (ii) if applicable, a statement indicating that such Bank is entitled to the "portfolio interest" exception under Section 871(h) or 881(c)(3) of the Code, and (iii) any other governmental forms which are necessary or required under an applicable tax treaty or otherwise by law to reduce or eliminate any withholding tax, which have been reasonably requested by the Borrower. Each Bank which

delivers to the Borrower and the Administrative Agent a Form W-8BEN or W-8ECI pursuant to the immediately preceding sentence further undertakes to deliver to the Borrower and the Administrative Agent two further copies of Form W-8BEN or W-8ECI, or other applicable forms, or other manner of certification, as the case may be, on or before the date that any such form expires or becomes obsolete or after the occurrence of any event requiring a change in the most recent form previously delivered by it to the Borrower and the Administrative Agent, and such extensions or renewals thereof as may reasonably be requested by the Borrower and the Administrative Agent certifying in the case of a Form W-8BEN or W-8ECI that such Bank is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes. If an event (including without limitation any change in treaty, law or regulation) has occurred prior to the date on which any delivery required by the preceding sentence would otherwise be required which renders all such forms inapplicable or which would prevent any Bank from duly completing and delivering any such form with respect to it and such Bank advises the Borrower and the Administrative Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income tax, such Bank shall not be required to deliver such forms. The Borrower shall withhold tax at the

rate and in the manner required by the laws of the United States with respect to payments made to a Bank failing to provide the requisite Internal Revenue Service forms in a timely manner. Each Bank which fails to provide to the Borrower in a timely manner such forms shall reimburse the Borrower upon demand for any penalties paid by the Borrower as a result of any failure of the Borrower to withhold the required amounts that are caused by such Bank's failure to provide the required forms in a timely manner. Notwithstanding any other provision of this paragraph, a Non U.S. Bank shall not be required to deliver any form pursuant to this paragraph that such Non U.S. Bank is not legally able to deliver.

# 2.13 Change of Lending Office.

a. Each Bank agrees that if it makes any demand for payment under Section 2.8 or is required to pay additional amounts pursuant to Section 2.12(a), or if any adoption or change of the type described in Section 2.9 shall occur with respect to it, it will if requested by the Borrower use reasonable efforts (consistent with its internal policy and legal and regulatory restrictions and so long as such efforts would not be disadvantageous to it, as determined in its sole discretion) to designate a different lending office for any loan made hereunder affected if the making of such a designation would reduce or obviate the need for the Borrower to make payments under Section 2.8 or 2.12(a), or would eliminate or reduce the effect of any adoption or change described in Section 2.9.

b. If any Bank (including any participant Bank under Section 8.2) shall assert that any adoption or change of the type described in Section 2.9 hereof has occurred with respect to it, or if any Bank (including any participant Bank under Section 8.2) requests compensation under Section 2.8, or if the Borrower is required to pay any additional amount to any Bank or any authority for the account of any Bank pursuant to Section 2.12, then the Borrower may, at its expense and effort, upon notice to such Bank and the Administrative Agent, require such Bank to, and such Bank promptly shall, assign and delegate, without recourse (in accordance with and subject to the restrictions contained in Section 8.2 (provided that the Borrower shall be obligated to pay the administrative fee referred to therein) and subject to Section 2.7 (as if such assignment

were a prepayment by the Borrower)), all its interests, rights, and obligations under this Agreement to an assignee that shall assume such obligations (which assignee may be another Bank, if a Bank accepts such assignment); provided that (i) if such assignee is not a Bank or an Affiliate thereof, the Borrower shall have received the prior written consent of the Administrative Agent, the Swing Line Lender and the Issuing Bank (which consents shall not unreasonably be withheld or delayed), (ii) such Bank shall have received payment of an amount equal to the aggregate outstanding principal of such Bank's Term Loan Advances and Revolving Loan Advances and its participations in Letters of Credit, accrued interest thereon, accrued fees and all other amounts payable to it hereunder, from the assignee (at least to the extent of such outstanding principal) and the Borrower (in the case of all other amounts), (iii) in the case of any such assignment resulting from a claim for compensation under Section 2.8 or payment

required to be made pursuant to Section 2.12, such assignment will result in a reduction in such compensation or payments compared to the compensation or payments payable to the assigning Bank, (iv) such assignment does not conflict with any requirement of law, (v) no Event of Default shall have occurred and be continuing at the time of such assignment, (vi) prior to any such assignment, such Bank shall have taken no action under Section 2.13(a) so as to eliminate the continued need for payment of amounts owing pursuant to Section 2.8 or 2.12(a), (vii) until such time as such assignment shall be consummated, the Borrower shall pay all additional amounts (if any) required pursuant to Section 2.8 or 2.12(a), as the case may be, and (viii) any such assignment shall not be deemed to be a waiver of any rights that the Borrower, the Administrative Agent or any other Bank shall have against the replaced Bank. A Bank shall not be required to make any such assignment and delegation if, prior thereto, as a result of a waiver by such Bank or otherwise, the circumstances entitling the Borrower to require such assignment and delegation no longer exist or cease to apply.

# ARTICLE 3 CONDITIONS PRECEDENT

3.1 Conditions Precedent to Initial Extension of Credit. The obligation of each Bank to make the initial extension of credit under this Agreement (not including any extension of credit under the 2001 Credit Agreement), including the making of any Term Loan Advance or Revolving Loan Advance and the issuance of any Letter of Credit, and the obligation of the Swing Line Lender to make any Swing Line Loan shall be subject to the following conditions precedent:

a. Documents.

The Borrower shall have delivered or shall have caused to be delivered the documents and other items listed on Exhibit F, together with any other documents requested by the Administrative Agent to document the agreements and intent of the Credit Documents, each in form and with substance satisfactory to the Administrative Agent (provided that if some or all of the Mortgages are not delivered by the Closing Date, the delivery of such Mortgage(s) as a condition precedent shall be deemed waived but it shall be an Event of Default if substantially all Mortgages have not been executed and delivered to the Administrative Agent, together with all documents related thereto as the Administrative Agent may require, within 90 days after the Closing Date);

b. Material Adverse Change.

No Material Adverse Change shall have occurred since December 31, 2003;

c. 2001 Credit Agreement.

The Administrative Agent shall have received satisfactory evidence that all obligations owing to any Prior Bank (which is not a Bank) and the Prior Administrative Agent shall have been paid in full and each letter of credit issued pursuant to the 2001 Credit Agreement has either been terminated or cancelled or the issuer thereof has accepted a Letter of Credit as security therefor;

d. Fee Letter.

The Borrower shall have paid all amounts due under the Fee Letter by the Closing Date; and

e. Defeasance of Senior Subordinated Notes.

The Borrower shall have delivered or caused to be delivered to the Administrative Agent evidence satisfactory to the Administrative Agent that, after giving effect to the Loan Advances to be made on the Closing Date, Senior Subordinated Notes with an aggregate principal amount of \$50,000,000 will be defeased in accordance with the terms of the Senior Subordinated Note Indentures.

3.2 Conditions Precedent to Each Extension of Credit.

The obligation of each Bank to make any extension of credit under this Agreement, including the making of any Term Loan Advance or Revolving Loan Advance and the issuance, increase, or extension of any Letter of Credit, and the obligation of the Swing Line Lender to make any Swing Line Loan shall be subject to the further conditions precedent that on the date of such extension of credit:

a. Representations and Warranties.

As of the date of the making of any extension of credit hereunder, the representations and warranties contained in each Credit Document shall be (i) if qualified as to materiality, true and correct or (ii) if not so qualified, true and correct in all material respects, on and as of such date as if made on and as of such date; and

b. Default.

As of the date of the making of any extension of credit hereunder, there shall exist no Default or Event of Default, and the making of the extension of credit would not cause a Default or Event of Default.

Each borrowing by and issuance of a Letter of Credit on behalf of the Borrower hereunder shall constitute a representation and warranty by the Borrower as of the date of such extension of credit that the conditions contained in this Section 3.2 have been satisfied.

# ARTICLE 4 REPRESENTATIONS AND WARRANTIES.

The Borrower represents and warrants to the Administrative Agent and each Bank, and with each request for any extension of credit hereunder, including the making of any Term Loan Advance or Revolving Loan Advance, and the issuance, increase, or extension of any Letter of Credit, and the making of any Swing Line Loan, again represents and warrants to the Administrative Agent and each Bank, as follows:

#### 4.1 Organization.

As of the date of this Agreement, each Restricted Entity (a) is duly organized, validly existing, and in good standing (or, with respect to any entity listed on Schedule IV, shall be in good standing within 30 days after the date of this Agreement) under the laws of such Person's respective jurisdiction of organization and (b) is duly licensed, qualified to do business, and in good standing in each jurisdiction in which such Person is organized, owns property, or conducts operations to the extent that any failure to be so licensed, qualified, or in good standing in accordance with this clause (b) could reasonably be expected to cause a Material Adverse Change.

## 4.2 Authorization.

The execution, delivery, and performance by each Credit Party of the Credit Documents to which such Credit Party is a party and the consummation of the transactions contemplated thereby (a) do not contravene the organizational documents of such Credit Party, (b) have been duly authorized by all necessary organizational action of each Credit Party, and (c) are within each Credit Party's organizational powers.

# 4.3 Enforceability.

Each Credit Document to which any Credit Party is a party has been duly executed and delivered by each Credit Party which is a party to such Credit Document and constitutes the legal, valid, and binding obligation of each such Credit Party, enforceable against each such Credit Party in accordance with such Credit Document's terms, except as limited by applicable bankruptcy, insolvency, reorganization, moratorium, or similar laws at the time in effect affecting the rights of creditors generally and subject to the availability of equitable remedies.

## 4.4 Absence of Conflicts and Approvals.

The execution, delivery, and performance by each Credit Party of the Credit Documents to which such Credit Party is a party and the consummation of the transactions contemplated thereby, (a) do not result in any violation or breach of any provisions of, or constitute a default under, any note, indenture, credit agreement, security agreement, credit support agreement, or other similar agreement to which such Credit Party is a party or any other material contract or agreement to which such Credit Party is a party, (b) do not violate any law or regulation binding on or affecting such Credit Party, (c) do not require any authorization, approval, or other action by, or any notice to or filing with, any governmental authority or any other Person, and (d) do not result in or require the creation or imposition of any Lien prohibited by this Agreement. 4.5 Investment Companies.

No Restricted Entity or Affiliate thereof is an "investment company" or a company "controlled" by an "investment company" within the meaning of the Investment Company Act of 1940, as amended.

### 4.6 Public Utilities.

No Restricted Entity or Affiliate thereof is a "holding company," or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company" or of a "subsidiary company" of a "holding company," within the meaning of the Public Utility Holding Company Act of 1935, as amended. No Restricted Entity or Affiliate thereof is a regulated public utility.

### 4.7 Financial Condition.

a. The Borrower has delivered to the Administrative Agent the Financial Statements, including therein the consolidated balance sheet of the Borrower and consolidated statement of income for the periods shown therein, which accurately and completely, in all material respects, present fairly the financial condition of Borrower as of such date and for such periods. The Borrower has further delivered to the Administrative Agent the Interim Financial Statements, and the Interim Financial Statements are accurate and complete in all material respects and present fairly the financial condition of Borrower as of their date and for their period in accordance with generally accepted accounting principles, as applicable to interim financial reports.

b. As of the date of the Financial Statements, there were no material contingent obligations, liabilities for taxes, unusual forward or long term commitments, or unrealized or anticipated losses of the Borrower or any of the Borrower's Subsidiaries, except as disclosed in the Financial Statements and the Interim Financial Statements, and adequate reserves for such items have been made in accordance with generally accepted accounting principles. No Material Adverse Change has occurred since the date of the Financial Statements. No Default exists.

# 4.8 Condition of Assets.

Each Restricted Entity has good and indefeasible title to substantially all of its owned property and valid leasehold rights in all of its leased property, as reflected in the financial statements most recently provided to the Administrative Agent free and clear of all Liens except Permitted Liens. Each Restricted Entity possesses and has properly approved, recorded, and filed, where applicable, all permits, licenses, patents, patent rights or licenses, trademarks, trademark rights, trade names rights, and copyrights which are useful in the conduct of its business and which the failure to possess could reasonably be expected to cause a Material Adverse Change. The material properties used in the operations of each Restricted Entity are in good repair, working order, and condition, normal wear and tear excepted. The properties of each Restricted Entity have not been adversely affected as a result of any fire, explosion, earthquake, flood, drought, windstorm, accident, strike or other labor disturbance, embargo, requisition or taking of property or cancellation of contracts, permits, or concessions by a governmental authority, riot, activities of armed forces, or acts of God or of any public enemy in any manner which (after giving effect to any insurance proceeds) could reasonably be expected to cause a Material Adverse Change.

### 4.9 Litigation.

There are no actions, suits, or proceedings pending or, to the knowledge of the Borrower, threatened against any Restricted Entity at law, in equity, or in admiralty, or by or before any governmental department, commission, board, bureau, agency, instrumentality, domestic or foreign, or any arbitrator which could reasonably be expected to cause a Material Adverse Change.

### 4.10 Subsidiaries.

As of the date of this Agreement, (a) the Borrower has no Subsidiaries except as disclosed in Schedule II, which sets forth the name, organizational number (if any) and jurisdiction of organization of each Subsidiary and, as to each such Subsidiary, the number of shares, units or interests (if any) and percentage of each class of equity owned by any Credit Party, (b) the Borrower has no Subsidiaries which have not been disclosed in writing to the Administrative Agent and (c) there are no outstanding subscriptions, options, warrants, calls, rights or other agreements or commitments (other than stock options granted to employees or directors and directors' qualifying shares) of any nature relating to any capital stock of the Borrower or any Subsidiary, except as created by the Credit Documents.

## 4.11 Laws and Regulations.

Each Restricted Entity is in compliance with all federal, state, and local laws and regulations which are applicable to the operations and property of such Person where the failure to comply with the same could reasonably be expected to cause a Material Adverse Change. No part of the proceeds of any Loan Advances, and no other extensions of credit hereunder, will be used for "buying" or "carrying" any "margin stock" within the respective meanings of each of the quoted terms under Regulation U of the Board as now and from time to time hereafter in effect or for any purpose that violates the provisions of the Regulations of the Board.

# 4.12 Environmental Compliance.

Each Restricted Entity has been and is in compliance with all Environmental Laws and has obtained and is in compliance with all related permits necessary for the ownership and operation of any such Person's properties, in each case, where the failure to be in compliance with the same could reasonably be expected to cause a Material Adverse Change. Each Restricted Entity has not received notice of and has not been investigated for any violation or alleged violation of any Environmental Law in connection with any such Person's presently or previously owned properties which currently threaten action or suggest liabilities which could reasonably be expected to cause a Material Adverse Change. Each Restricted Entity (a) does not and has not created, handled, transported, used, or disposed of any Hazardous Materials on or about any such Person's properties (nor has any such Person's properties been used for those purposes); (b) has never been responsible for the release of any Hazardous Materials into the environment in connection with any such Person's operations and has not contaminated any properties with Hazardous Materials, and (c) does not and has not owned any properties contaminated by any Hazardous Materials, in each case in any manner which could reasonably be expected to cause a Material Adverse Change.

#### 4.13 ERISA.

Each Restricted Entity and each of their respective Commonly Controlled Entities are in compliance with all provisions of ERISA, the Code and other federal, state or local law with respect to any Plan to the extent that the failure to be in compliance could reasonably be expected to cause a Material Adverse Change. No Restricted Entity nor any of their respective Commonly Controlled Entities participates in or during the past five years has participated in any employee pension benefit plan covered by Title IV of ERISA or any multiemployer plan under Section 4001(a)(3) of ERISA. No Restricted Entity nor any of their respective Commonly Controlled Entities has any liability with respect to any Plan. With respect to the Plans of the Restricted Entities, no Reportable Event or Prohibited Transaction has occurred and exists that could reasonably be expected to cause a Material Adverse Change.

## 4.14 Taxes.

Each Restricted Entity has filed all United States federal, state, and local income tax returns and all other domestic and foreign tax returns which are required to be filed by such Person and has paid, or provided for the payment before the same became delinquent of, all taxes due pursuant to such returns or pursuant to any assessment received by such Person except for tax payments being contested in good faith, for which adequate reserves have been established and reported in

accordance with generally accepted accounting principals, and which could not reasonably be expected to cause a Material Adverse Change. The charges, accruals, and reserves on the books of the Restricted Entities in respect of taxes are adequate in accordance with generally accepted accounting principles.

#### 4.15 True and Complete Disclosure.

All factual information furnished by or on behalf of any Credit Party in writing to the Administrative Agent or any Bank in connection with the Credit Documents and the transactions contemplated thereby is true and accurate in all material respects on the date as of which such information was dated or certified and does not contain any untrue statement of material fact or omit to state any material fact necessary to make the statements contained therein not misleading. All projections, estimates, and pro forma financial information furnished by any Credit Party were prepared on the basis of assumptions, data, information, tests, or conditions believed to be reasonable at the time such projections, estimates, and pro forma financial informatio. 4.16 Senior Indebtedness.

The Credit Obligations constitute "Senior Indebtedness" (or any other defined term having a similar purpose) of the Borrower under and as defined in the Senior Subordinated Note Indentures and any Subordinated Debt Indenture. The obligations of each Guarantor under the Guaranty constitute "Guarantor Senior Indebtedness" (or any other defined term having a similar purpose) of such Guarantor under and as defined in the Senior Subordinated Note Indentures and any Subordinated Debt Indenture.

# 4.17 Certain Documents.

The Borrower has delivered to the Administrative Agent a complete and correct copy of the Senior Subordinated Note Indentures and any Subordinated Debt Indenture, in each case including any amendments, supplements or modifications with respect to any of the foregoing.

## ARTICLE 5 COVENANTS

Until the Administrative Agent and the Banks receive irrevocable payment of the Credit Obligations and have terminated this Agreement and each other Credit Document, and all Letters of Credit have expired or been cancelled, the Borrower shall comply with and cause compliance with the following covenants:

## 5.1 Organization.

The Borrower shall cause each Restricted Entity to (a) maintain itself as an entity duly organized, validly existing, and in good standing under the laws of such Person's respective jurisdiction of organization and (b) be duly licensed, qualified to do business, and in good standing in each jurisdiction in which such Person is organized, owns property, or conducts operations and which requires such licensing or qualification where failure to be so licensed, qualified, or in good standing as required by this clause (b) could reasonably be expected to cause a Material Adverse

Change; provided, however, that nothing in this Section 5.1 shall be interpreted to be violated as a result of a transaction permitted by Section 5.9.

Reporting. The Borrower shall furnish to the Administrative Agent (and the Administrative Agent shall promptly thereafter furnish to each Bank) all of the following:

### a. Annual Reports.

As soon as available and in any event not later than 90 days after the end of each fiscal year of the Borrower, (i) a copy of the annual financial statements for such fiscal year for the Borrower, including therein the consolidated balance sheets of the Borrower as of the end of such fiscal year and the consolidated statements of income, stockholders' equity, and cash flows for the Borrower for such fiscal year and all notes thereto, setting forth the consolidated financial position and results of the Borrower for such fiscal year, accompanied by the unqualified opinion of a nationally recognized certified public accounting firm to the effect that such financial statements fairly present the consolidated financial condition of the Borrower as of the date thereof and the result of its consolidated operations for the period covered thereby in accordance with generally accepted accounting principles, consistently applied, and without expressing any doubt as to the ability of the Borrower or any of its Subsidiaries to continue as a going concern, (ii) a completed Compliance Certificate duly certified by a Responsible Officer of the Borrower, and (iii) a completed Contract Status Report duly certified by a Responsible Officer of the Borrower;

b. Quarterly Reports.

As soon as available and in any event not later than 45 days after the end of each of the first three fiscal quarters of the Borrower of each year, and in each case in form and substance acceptable to the Administrative Agent, (i) a copy of the internally prepared consolidated financial statements of the Borrower for such fiscal quarter and for the fiscal year to date period ending on the last day of such fiscal quarter, including therein the consolidated balance sheets of the Borrower as of the end of such fiscal quarter and the consolidated statements of income, and cash flows for such fiscal quarter and for such fiscal year to date period, setting forth the consolidated financial position and results of the Borrower for such fiscal quarter and fiscal year to date period, all in reasonable detail and duly certified by a Responsible Officer of the Borrower as having been prepared in accordance with generally accepted accounting principles, including those applicable to interim financial reports which permit normal year end adjustments and do not require complete financial notes, (ii) a completed Compliance Certificate duly certified by a Responsible Officer of the Borrower, and (iii) a completed Contract Status Report duly certified by a Responsible Officer of the Borrower;

c. Semi-annual Subsidiary Update.

As soon as available and in any event not later than 45 days after the end of the second fiscal quarter of the Borrower of each year and 90 days after the end of the fourth fiscal quarter of the Borrower of each year, an update of the information included in Schedule II, including all Subsidiaries acquired by the Borrower since the previous update of such scheduled information.

d. Acquisition Information.

As soon as available prior to the closing of any Acquisition requiring approval of the Majority Banks, and on or prior to the closing of any Acquisition not requiring such approval, a completed

Acquisition Certificate duly certified by a Responsible Officer of the Borrower, which the Administrative Agent shall forward to the Banks for any Acquisition requiring approval of the Majority Banks (and prior to the consummation of the Acquisition, the Borrower shall, upon request by the Administrative Agent, make available to the Administrative Agent at the Borrower's offices in Houston, Texas, the acquisition documents regarding the acquired assets, including schedules reflecting litigation liabilities, environmental liabilities, and other assumed liabilities, and any other information regarding the acquired assets as the Administrative Agent may reasonably request);

e. SEC Filings.

As soon as available and in any event not later than 30 days after the filing or delivery thereof, copies of all financial statements, reports, and proxy statements which the Borrower shall have sent to its stockholders generally and copies of all regular and periodic reports, if any, which any Restricted Entity shall have filed with the Securities and Exchange Commission;

f. Consolidated Budget and Projections.

As soon as available, and in any event no later than 45 days after the end of each fiscal year of the Borrower, a detailed consolidated budget for the following fiscal year (including a projected consolidated balance sheet of the Borrower and its Subsidiaries as of the end of the following fiscal year, the related consolidated statements of projected cash flow, projected changes in financial position and projected income and a description of the underlying assumptions applicable thereto), and, within 5 days after the availability thereof, significant revisions, if any, of such budget and projections with respect to such fiscal year (collectively, the "Projections"), which Projections shall in each case be accompanied by a certificate of a Responsible Officer of the Borrower stating that such Projections are based on reasonable estimates, information and assumptions and made in good faith;

## g. Defaults.

Promptly, but in any event within 5 Business Days after the discovery thereof, a notice of any facts known to a Responsible Officer of the Borrower which constitute a Default, together with a statement of a Responsible Officer of the Borrower setting forth the details of such facts and the actions which the Borrower has taken and proposes to take with respect thereto (and the Administrative Agent shall, promptly upon receipt from the Borrower of a notice pursuant to this Section 5.2(g), forward a copy of such notice to each Bank);

#### h. Litigation.

Promptly, but in any event within 10 Business Days after notice of commencement thereof, notice of all actions, suits, and proceedings before or brought by any court or governmental department, commission, board, bureau, agency, or instrumentality, domestic or foreign, affecting any Restricted Entity which, if determined adversely, could reasonably be expected to cause a Material Adverse Change;

## i. Material Agreement Default.

Promptly, but in any event within 10 Business Days after a Responsible Officer of the Borrower obtains knowledge thereof, notice of any breach by any

Restricted Entity of any contract or agreement which breach could reasonably be expected to cause a Material Adverse Change;

#### j. Material Changes.

Prompt written notice of any other condition or event of which a Responsible Officer of the Borrower has knowledge, which condition or event has resulted in or could reasonably be expected to cause a Material Adverse Change;

k. Indentures; Senior Subordinated Notes.

(i) As soon as available, a complete and correct copy of each of the Senior Subordinated Note Indentures and any Subordinated Debt Indenture; and (ii) no later than 5 Business Days prior to the effectiveness thereof, copies of substantially final drafts of any proposed amendment, supplement, waiver or other modification with respect to the Senior Subordinated Note Indentures or any Subordinated Debt Indenture;

1. Asset Coverage Computation.

Within 25 days after the last day of each calendar month, a computation of the "Asset Coverage Ratio" in accordance with Section 5.5(e) as of such last day, duly certified by a Responsible Officer of the Borrower; and

m. Other Information.

Such other information respecting the business operations or property of any Restricted Entity, financial or otherwise, as the Administrative Agent or the Majority Banks may from time to time reasonably request.

## 5.3 Inspection.

The Borrower shall cause each Restricted Entity to permit the Administrative Agent and the Banks to visit and inspect any of the properties of such Restricted Entity, to examine all of such Person's books of account, records, reports, and other papers, to make copies and extracts therefrom, and to discuss their respective affairs, finances, and accounts with their respective officers, employees, and independent public accountants all at such reasonable times and as often as may be reasonably requested, provided that the Borrower is given at least 3 Business Days' advance notice thereof and reasonable opportunity to be present when independent public accountants or other third parties are contacted, and provided further that so long as no Default or Event of Default exists, the Administrative Agent and the Banks shall not exercise the foregoing inspection right more often than once in any calendar year.

## 5.4 Use of Proceeds.

The proceeds of the Revolving Loan Borrowings shall be used by the Borrower for Acquisitions, working capital needs, Capital Expenditures, and for other lawful corporate purposes. The proceeds of the Term Loan Borrowing shall be used by the Borrower in connection with the defeasance of \$50,000,000 in principal amount of

Senior Subordinated Notes.

## 5.5 Financial Covenants.

The Administrative Agent shall determine compliance with the following financial covenants based upon the applicable Schedule of the most recent Compliance Certificate delivered to the Administrative Agent pursuant to Section 5.2(a) or 5.2(b).

#### a. Net Worth.

The Borrower shall not permit the consolidated Net Worth of the Borrower as of the last day of each fiscal quarter to be less than the sum of (i) \$214,045,000; plus (ii) 50% of the cumulative quarterly consolidated net income of the Borrower after September 30, 2003, for each fiscal quarter of the Borrower during which the Borrower has positive consolidated net earnings (disregarding impairment to goodwill calculated in accordance with FASB Statement No. 142, Goodwill and Other Intangibles); plus (iii) 100% of the net proceeds received by Borrower after September 30, 2003, from any sale or issuance of any equity securities of, or any other additions to capital by, the Borrower or its Subsidiaries; plus (iv) to the extent that the required consolidated Net Worth under this Section 5.5(a) was not increased in clauses (i) through (iii) above as a result of any Acquisition, 100% of any increase in the consolidated Net Worth of the Borrower resulting from any Acquisition. Compliance with this paragraph (a) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule A of such Compliance Certificate.

b. Maximum Leverage Ratios.

i. Maximum Senior Debt to EBITDA Ratio. As of the last day of each fiscal quarter of the Borrower, the Borrower shall not permit the ratio of (a) the consolidated Senior Debt of the Borrower as of end of such fiscal quarter to (b) the consolidated EBITDA of the Borrower for the preceding four fiscal quarters then ended, to be greater than 2.25 to 1.00.

ii. Maximum Total Debt to EBITDA Ratio. As of the last day of each fiscal quarter of the Borrower, the Borrower shall not permit the ratio of (a) the consolidated Total Debt of the Borrower as of end of such fiscal quarter to (b) the consolidated EBITDA of the Borrower for the preceding four fiscal quarters then ended, to be greater than (x) 3.85 to 1.00 at any time on or before December 31, 2006 and (y) 3.60 to 1.00 at any time after December 31, 2006.

Compliance with this paragraph (b) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

c. Minimum Fixed Charge Coverage Ratio.

As of the last day of each fiscal quarter, the Borrower shall not permit the

ratio of (i) the consolidated Net EBITDA of the Borrower for the preceding four fiscal quarters then ended to (ii) the consolidated Fixed Charges of the Borrower for the preceding four fiscal quarters then ended to be less than 1.35 to 1.00. Compliance with this paragraph (c) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

#### d. Capital Expenditures.

The Borrower shall not permit the consolidated Capital Expenditures (other than Capital Expenditures that are deemed to occur because of the making of an Acquisition) of the Borrower during any fiscal year to exceed \$20,000,000. Compliance with this paragraph (d) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

e. Asset Coverage Ratio.

As of the last day of each calendar month, the Borrower shall not permit the ratio of (1) the sum of (i) 85% of consolidated Unbonded Accounts Receivable plus (ii) 50% of consolidated inventories plus (iii) 50% of consolidated equipment (net of depreciation and excluding real property interests and equipment attached to real property) to (2) the sum of (i) the Revolving Loan plus (ii) the Letter of Credit Exposure plus (iii) the Term Loan plus (iv) the Swing Line Loan to be less than 1.00 to 1.00. Compliance with this paragraph (e) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

f. Rent and Lease Expense. As of the last day of each fiscal quarter, the Borrower shall not permit its consolidated rent and lease expense for the preceding four fiscal quarters then ended, to be greater than \$30,000,000. Compliance with this paragraph (f) shall be determined in the applicable Compliance Certificate based upon the adjusted financial reports contained in Schedule B of such Compliance Certificate.

5.6 Debt.

a. The Borrower shall not permit any Restricted Entity to create, assume, incur, suffer to exist, or in any manner become liable, directly, indirectly, or contingently in respect of, any Debt other than Permitted Debt.

b. The Borrower shall, as soon as available but in any event not less than 10 Business Days prior to the issuance of any preferred stock or subordinated indebtedness, deliver to the Administrative Agent a copy of the certificate of designation or subordinated debt documents, as applicable, together with a certificate, signed by a Responsible Officer of the Borrower, certifying that such preferred stock or subordinated indebtedness constitutes Qualified Preferred Stock or Subordinated Debt pursuant to the terms of this Agreement.

5.7 Liens.

The Borrower shall not permit any Restricted Entity to create, assume, incur, or suffer to exist any Lien on any of its real or personal property whether now owned or hereafter acquired, or assign any right to receive its income, except for Permitted Liens.

#### 5.8 Other Obligations.

The Borrower shall not permit any Restricted Entity to create, incur, assume, or suffer to exist any obligations in respect of unfunded vested benefits under any Plan or deferred compensation agreement in an aggregate outstanding amount in excess of \$5,000,000.

The Borrower shall not permit any Restricted Entity to create, incur, assume, or suffer to exist any obligations in respect of Derivatives, other than Derivatives used by any Restricted Entity in such Restricted Entity's respective business operations in aggregate notional quantities not to exceed the reasonably anticipated consumption of such Restricted Entity of the underlying commodity for the relevant period, but no Derivatives which are speculative in nature.

## 5.9 Corporate Transactions and Asset Sales.

a. The Borrower shall not, without the Majority Banks' consent, permit any Restricted Entity to (1) merge, consolidate, or amalgamate with another Person, or liquidate, wind up, or dissolve itself (or take any action towards any of the foregoing) or (2) make any Acquisition except that:

i. Any Subsidiary of the Borrower may merge, consolidate, or amalgamate into the Borrower or any wholly owned Subsidiary of the Borrower, or convey, sell, lease, assign, transfer, or otherwise dispose of any of its assets to the Borrower or any wholly owned Subsidiary of the Borrower (and if such disposition transfers all or substantially all of the assets of the transferring Subsidiary, such Subsidiary may then liquidate, wind up, or dissolve itself); provided that the Borrower or the wholly owned Subsidiary, as applicable, is the surviving or acquiring entity; and

ii. The Borrower or any Subsidiary of the Borrower may make any Acquisition (by purchase or merger) provided that (A) the Borrower or such Subsidiary of the Borrower is the acquiring or surviving entity, (B) the aggregate of all consideration (other than common stock of the Borrower) paid by the Restricted Entities in connection with all Acquisitions during any fiscal year does not exceed \$25,000,000, (C) no Default or Event of Default exists and the Acquisition would not reasonably be expected to cause a Default or Event of Default after giving pro forma effect thereto (including any default under Section 5.5 with respect to historical and future pro forma financial status and results), and (D) the acquired assets are in substantially the same business as the Borrower. b. The Borrower shall not permit any Restricted Entity to use any Collateral in an unlawful manner or contrary to the terms and conditions of any policy of insurance thereon. Until the occurrence of an Event of Default, (1) each Restricted Entity may sell or lease its Inventory (as such term is defined in the Security Agreement) in the ordinary course of its business; (2) each Restricted Entity may also use and consume any raw materials or supplies, the use and consumption of which are necessary to carry on such Restricted Entity's business; (3) each Restricted Entity may sell or transfer any Collateral in connection with the acquisition of similar property with a fair market value at least equal to the fair market value of the disposed property (and the Banks hereby authorize and direct the Administrative Agent to release its Lien on such disposed Collateral if such Restricted Entity takes all steps reasonably required by the Administrative Agent to grant a Lien on such acquired property), and (4) each Restricted Entity may sell or transfer any Collateral, subject to the provisions of Sections 2.1(c)(ii) and 2.2(c)(ii) (and the Banks hereby authorize and direct the Administrative Agent to release its Lien on such Collateral). A sale in the ordinary course of business does not include a transfer in partial or total satisfaction of a Debt.

## 5.10 Distributions.

The Borrower shall not (a) declare or pay any dividends; (b) purchase, redeem, retire, or otherwise acquire for value any of its capital stock now or hereafter outstanding; (c) make any distribution of assets to its stockholders as such, whether in cash, assets, or in obligations of it; (d) allocate or otherwise set apart any sum for the payment of any dividend or distribution on, or for the purchase, redemption, or retirement of, any shares of its capital stock; or (e) make any other distribution by reduction of capital or otherwise in respect of any shares of its capital stock, except that the Borrower may make payments of dividends on Qualified Preferred Stock; provided, however, that (i) the Borrower may declare or pay dividends if (x) no Default has then occurred and is continuing and (y) assuming such dividend had been paid on the last day of the immediately preceding fiscal quarter, (1) the Total Debt to EBITDA ratio, as measured in accordance with Section 5.5(b)(ii), would have been no greater than 2.50 to 1 and (2) no Default (including but not limited to any Default under Section 5.5) would have occurred as of the last day of such quarter and (ii) the Borrower may purchase, redeem, retire or otherwise acquire for value any of its capital stock if (x) no Default has then occurred and is continuing and (y) assuming such transaction had been consummated on the last day of the immediately preceding fiscal quarter, no Default (including but not limited to any Default under Section 5.5) would have occurred as of the last day of such quarter.

# 5.11 Transactions with Affiliates.

The Borrower shall not permit any Restricted Entity to enter into any transaction directly or indirectly with or for the benefit of an Affiliate except transactions with an Affiliate for the leasing of property, the rendering or receipt of services, or the purchase or sale of inventory or other assets in the ordinary course of business if the monetary or business consideration arising from such a transaction would be substantially as advantageous to such Restricted Entity as the monetary or business consideration which such Restricted Entity would obtain in a comparable arm's length transaction; provided that the Restricted Entities may make the investments described in item (g) of the definition of "Permitted Investments".

# 5.12 Insurance.

a. The Borrower shall cause each Restricted Entity to maintain insurance with responsible and reputable insurance companies or associations reasonably acceptable to the Administrative Agent in such amounts and covering such risks as are usually carried by companies engaged in similar businesses and owning similar properties in the same general areas in which such Persons operate. The Borrower shall deliver to the Administrative Agent certificates evidencing such policies or copies of such policies at the Administrative Agent's request following a reasonable period to obtain such certificates taking into account the jurisdiction where the insurance is maintained.

b. All policies representing insurance of the Restricted Entities shall name the Administrative Agent and the Banks as loss payee (with respect to casualty insurance covering Collateral) or additional named insureds (in all other cases) in a form satisfactory to the Administrative Agent. With respect to any insured liability suffered by the Administrative Agent or any Bank, all proceeds of such liability insurance coverage for the Administrative Agent and the Banks shall be paid as directed by the Administrative Agent to indemnify the Administrative Agent or the applicable Bank for the liability covered. In the event that proceeds of property or liability insurance are paid to any Restricted Entity in violation of the foregoing, the Restricted Entity shall hold the proceeds in trust for the Administrative Agent, segregate the proceeds from the other funds of such Restricted Entity, and promptly pay the proceeds to the Administrative Agent with any necessary endorsement. The Administrative Agent shall have the right, but not the obligation, during the existence of an Event of Default, to make proof of loss under, settle and adjust any claim under, and receive the proceeds under the insurance, and the reasonable expenses incurred by the Administrative Agent in the adjustment and collection of such proceeds shall be paid by the Borrower. The Borrower irrevocably appoints the Administrative Agent as its attorney in fact to take such actions in its name. If the Administrative Agent does not take such actions, the Borrower may take such actions subject to the approval of any final action by the Administrative Agent. The Administrative Agent shall not be liable or responsible for failure to collect or exercise diligence in the collection of any proceeds.

# 5.13 Investments.

The Borrower shall not permit any Restricted Entity to make or hold any direct or indirect investment in any Person, including capital contributions to the Person, investments in the debt or equity securities of the Person, and loans, guaranties, trade credit, or other extensions of credit to the Person, except for Permitted Investments.

#### 5.14 Lines of Business.

The Borrower shall not permit the Restricted Entities to engage in any type of

business not reasonably related to their business as presently and normally conducted. The parties agree that business entailing the construction, operation, or ownership of electrical infrastructure, electrical generation, electrical distribution, distributed generation, communications contracting, communications servicing, e-commerce ventures and other related ventures, together with associated financial and risk management activities conducted for the benefit of the Restricted

Entities in connection with such business, are reasonably related to the character of the Restricted Entities' business as conducted on the date hereof.

# 5.15 Compliance with Laws.

The Borrower shall cause each Restricted Entity to comply with all federal, state, and local laws and regulations which are applicable to the operations and property of such Persons, in each case, where the failure to comply could reasonably be expected to cause a Material Adverse Change.

## 5.16 Environmental Compliance.

The Borrower shall cause each Restricted Entity to comply with all Environmental Laws and obtain and comply with all related permits necessary for the ownership and operation of any such Person's properties, in each case, where the failure to comply could reasonably be expected to cause a Material Adverse Change. The Borrower shall cause each Restricted Entity to disclose promptly to the Administrative Agent any notice to or investigation of such Persons for any violation or alleged violation of any Environmental Law in connection with any such Person's presently or previously owned properties which represent liabilities which, if determined adversely to such Restricted Entity, could reasonably be expected to cause a Material Adverse Change. The Borrower shall not permit any Restricted Entity to create, handle, transport, use, or dispose of any Hazardous Materials on or about any such Person's properties; release any Hazardous Materials into the environment in connection with any such Person's operations or contaminate any properties with Hazardous Materials; or own properties contaminated by any Hazardous Materials, in each case in any manner that could reasonably be expected to cause a Material Adverse Change.

### 5.17 ERISA Compliance.

The Borrower shall cause each Restricted Entity to (i) comply in all material respects with all applicable provisions of ERISA, the Code and other federal, state or local laws and prevent the occurrence of any Reportable Event or Prohibited Transaction with respect to, or the termination of, any of their respective Plans, in each case, where the failure to do so could reasonably be expected to cause a Material Adverse Change and (ii) not create or participate in any employee pension benefit plan covered by Title IV of ERISA or any multiemployer plan under Section 4001(a) (3) of ERISA.

#### 5.18 Payment of Certain Claims.

The Borrower shall cause each Restricted Entity to pay and discharge, before the

same shall become delinquent, (a) all taxes, assessments, levies, and like charges imposed upon any such Person or upon any such Person's income, profits, or property by authorities having competent jurisdiction prior to the date on which penalties attach thereto except for tax payments being contested in good faith for which adequate reserves have been established and reported in accordance with generally accepted accounting principals which could not reasonably be expected to cause a Material Adverse Change and (b) all trade payables and current operating

liabilities, unless the same are less than 90 days past due or are being contested in good faith, have adequate reserves established and reported in accordance with generally accepted accounting principals, and could not reasonably be expected to cause a Material Adverse Change.

#### 5.19 Subsidiaries.

Upon the formation or acquisition of any new Subsidiary, the Borrower shall and shall cause such Subsidiary to promptly, but in any event within 30 days after the formation or acquisition of such new Subsidiary (a) execute and deliver to the Administrative Agent a Joinder Agreement (each, a "Joinder Agreement") in substantially the form of Exhibit G for the purpose of joining such Subsidiary as a party to the Guaranty, the Security Agreement, and, if applicable, the Stock Pledge Agreement or the Partnership Pledge Agreement, and (b) provide to the Administrative Agent the other certificates, documents, and items listed on Exhibit I.

5.20 Changes in Fiscal Periods.

The Borrower shall not permit the fiscal year of the Borrower to end on a day other than September 30 or change the Borrower's method of determining fiscal quarters.

5.21 Negative Pledge Clauses.

The Borrower shall not enter into or suffer to exist or become effective any agreement that prohibits or limits the ability of any Restricted Entity to create, incur, assume or suffer to exist any Lien upon any of its property or revenues, whether now owned or hereafter acquired, to secure its obligations under the Credit Documents to which it is a party other than (a) this Agreement and the other Credit Documents and (b) any agreements governing any purchase money Liens or Capital Lease Obligations otherwise permitted hereby (in which case, any prohibition or limitation shall only be effective against the assets financed thereby).

#### 5.22 Subordinated Debt Payments.

The Borrower shall not make any payment of principal, interest or premium on the Senior Subordinated Notes or any Subordinated Debt (not including the defeasance of the Senior Subordinated Notes as provided in Section 3.1(e)) unless no Default has then occurred and is continuing or would result therefrom.

# ARTICLE 6 DEFAULT AND REMEDIES

#### 6.1 Events of Default.

The failure to deliver the Mortgages as set forth in the proviso of Section 3.1(a) and each of the following shall be an "Event of Default" for the purposes of this Agreement and for each of the Credit Documents:

#### a. Payment Failure.

The Borrower (i) fails to pay when due any principal amounts due under this Agreement or any other Credit Document or (ii) fails to pay when due any interest, fees, reimbursements, indemnifications, or other amounts due under this Agreement or any other Credit Document and such failure has not been cured within five Business Days;

b. False Representation.

Any written representation or warranty made by any Credit Party or any Responsible Officer thereof in this Agreement or in any other Credit Document proves to have been false or erroneous in any material respect at the time it was made or deemed made;

c. Breach of Covenant.

(i) Any breach by the Borrower of any of the covenants contained in Sections 5.1(a) (with respect to the Borrower), 5.2, 5.3, 5.4, 5.5, 5.6, 5.7, 5.8, 5.9, 5.10, 5.13, 5.19 or 5.22 or (ii) any breach by any Credit Party of any other covenants contained in this Agreement, or any other Credit Document and such breach is not cured within 30 days following the earlier of knowledge of such breach by a Responsible Officer of such Credit Party or the receipt of written notice thereof from the Administrative Agent;

d. Security and Support Documents.

Any Security Document shall at any time and for any reason (other than one within the reasonable control of any Bank) cease to create the Lien on the property purported to be subject to such agreement in accordance with the terms of such agreement, or cease to be in full force and effect, or shall be contested by any party thereto;

e. Guaranty.

(i) The Guaranty shall at any time and for any reason cease to be in full force and effect with respect to any Guarantor (except as permitted under Section 5.9 hereof) or shall be contested by any Guarantor, or any Guarantor shall deny it has any further liability or obligation thereunder, or (ii) any breach by any Guarantor of any of the covenants contained in Section 1 of the Guaranty;

f. Material Debt Default.

(i) Any principal, interest, fees, or other amounts due on any Debt of any Restricted Entity (other than the Credit Obligations) is not paid when due, whether by scheduled maturity, required prepayment, acceleration, demand, or otherwise, and such failure is not cured within the applicable grace period, if any, and the aggregate amount of all Debt of such Persons so in default exceeds \$2,000,000; (ii) any other event shall occur or condition shall exist under any agreement or instrument relating to any Debt of any such Person (other than the Credit Obligations) the effect of which is to accelerate or to permit the acceleration of the maturity of any such Debt, whether or not any such Debt is actually accelerated, and such event or condition shall not be cured within the applicable grace period, if any, and the aggregate amount of all Debt of such Persons so in default exceeds \$2,000,000; or (iii) any Debt of any such Person shall be declared to be due and payable, or required to be prepaid (other than by a regularly scheduled prepayment) prior to the stated maturity thereof, and the aggregate amount of all Debt of such Persons so accelerated exceeds \$2,000,000;

g. Material Agreement Default; Material Plan Event.

(i) There shall occur any breach by any Restricted Entity of any contract or agreement which breach could reasonably be expected to cause a Material Adverse Change and such breach is not cured within the applicable grace period, if any; or (ii) there shall occur or exist an event with respect to any Plan which could in the sole judgment of the Administrative Agent reasonably be expected to have a Material Adverse Change;

h. Bankruptcy and Insolvency.

(i) There shall have been filed against any Restricted Entity or any such Person's properties, without such Person's consent, any petition or other request for relief seeking an arrangement, receivership, reorganization, liquidation, or similar relief under bankruptcy or other laws for the relief of debtors, and such request for relief (A) remains in effect for 60 or more days, whether or not consecutive, or (B) is approved by a final nonappealable order, or (ii) any such Person consents to or files any petition or other request for relief of the type described in clause (i) above seeking relief from creditors, makes any assignment for the benefit of creditors or other arrangement with creditors, or admits in writing such Person's inability to pay such Person's debts as they become due (the occurrence of any Event of Default under clause (i) or (ii) of this paragraph being a "Bankruptcy Event of Default");

i. Adverse Judgment.

The aggregate outstanding amount of judgments against the Restricted Entities not discharged or stayed pending appeal or other court action within 30 days following entry is greater than \$2,000,000;

j. Change of Control.

There shall occur any Change of Control; or

#### k. Senior Subordinated Notes Default.

(i) There shall occur any default or event of default (and such event or condition is not cured within the applicable grace period, if any), however denominated, under the Senior Subordinated Notes or either Senior Subordinated Note Indenture or any Subordinated Debt or any Subordinated Debt Indenture; (ii) any modification shall be made to the subordination provisions or economic terms of the Senior Subordinated Notes or either Senior Subordinated Note Indenture or any Subordinated Debt or any Subordinated Debt Indenture without the prior written consent of the Majority Banks; or (iii) any "Change of Control Offer" (or any other defined term having a similar purpose) as defined in either Senior Subordinated Note Indenture or any Subordinated Debt Indenture shall occur.

## 6.2 Termination of Commitments.

Upon the occurrence of any Bankruptcy Event of Default, all of the Commitments of the Administrative Agent and the Banks hereunder shall terminate. During the existence of any Event of Default other than a Bankruptcy Event of Default, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower all of the Commitments of the Administrative Agent and the Banks hereunder terminated, whereupon the same shall immediately terminate.

## 6.3 Acceleration of Credit Obligations.

Upon the occurrence of any Bankruptcy Event of Default, the aggregate outstanding principal amount of all loans made hereunder, all accrued interest thereon, and all other Credit Obligations shall immediately and automatically become due and payable. During the existence of any Event of Default other than a Bankruptcy Event of Default, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower the aggregate outstanding principal amount of all loans made hereunder, all accrued interest thereon, and all other Credit Obligations to be immediately due and payable, whereupon the same shall immediately become due and payable. In connection with the foregoing, except for the notice provided for above in this Article 6, the Borrower waives notice of any Default or Event of Default, grace, notice of intent to accelerate, notice of acceleration, presentment, demand, notice of nonpayment, protest, and all other notices.

# 6.4 Cash Collateralization of Letters of Credit.

Upon the occurrence of any Bankruptcy Event of Default, the Borrower shall pay to the Administrative Agent an amount equal to the Letter of Credit Exposure to be held in the Letter of Credit Collateral Account for disposition in accordance with Section 2.3(g). During the existence of any Event of Default other than a Bankruptcy Event of Default, the Administrative Agent shall at the request of the Majority Banks require by written notice to the Borrower that the Borrower pay to the Administrative Agent an amount equal to the Letter of Credit Exposure to be held in the Letter of Credit Collateral Account for disposition in accordance with Section 2.3(g), whereupon the Borrower shall pay to the Administrative Agent such amount for such purpose.

## 6.5 Default Interest.

If any Event of Default exists based upon a default in the payment of any amounts owing hereunder, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower that the Credit Obligations specified in such notice shall bear interest beginning on the date specified in such notice until paid in full at the applicable Default Rate for such Credit Obligations, whereupon the Borrower shall pay such interest to the Administrative Agent for the benefit of the Administrative Agent and the Banks, as applicable, upon demand by the Administrative Agent. If any other Event of Default exists, the Administrative Agent shall at the request of the Majority Banks declare by written notice to the Borrower that, unless such Event of Default is cured to the satisfaction of the Administrative Agent and the Majority Banks on or before the 30th day following the occurrence of such Event of Default, the Credit Obligations specified in such notice shall bear interest beginning on such 30th day until paid in full at the applicable Default Rate for such Credit Obligations, whereupon the Borrower shall, if such Event of Default is not cured by such date, pay such interest to the Administrative Agent for the benefit of the Administrative Agent and the Banks, as applicable, upon demand by the Administrative Agent after such date.

6.6 Right of Setoff; Adjustments.

a. During the existence of an Event of Default, each of the Administrative Agent and the Banks is hereby authorized at any time, to the fullest extent permitted by law, to set off and apply any indebtedness owed by the Administrative Agent or such Bank to the Borrower against any and all of the obligations of the Borrower under this Agreement and the other Credit Documents, irrespective of whether or not the Administrative Agent or such Bank shall have made any demand under this Agreement or the other Credit Documents and although such obligations may be contingent and unmatured. Each of the Administrative Agent and the Banks agrees promptly to notify the Borrower after any such setoff and application made by such party provided that the failure to give such notice shall not affect the validity of such setoff and application.

b. Each Bank agrees that if any Bank (a "benefited Bank") shall at any time receive any payment of all or part of its Loan Advances, or interest thereon, or receive any collateral in respect thereof (whether voluntarily or involuntarily, by set-off or otherwise) in a greater proportion than any such payment to or collateral received by any other Bank, if any, in respect of such other Loan Advances, or interest thereon, such benefited Bank shall purchase for cash from the other Banks a participating interest in such portion of each such other Bank's Loan Advances, or shall provide such other Banks with the benefits of any such collateral, or the proceeds thereof, as shall be necessary to cause such benefited Bank to share the excess payment or benefits of such collateral or proceeds ratably with each of the Banks; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such benefited Bank, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. The Borrower agrees that each Bank so purchasing a portion of another Bank's Loan Advances may exercise all rights of payment (including, without limitation, rights of set-off) with respect to such portion as fully as if such Bank were the direct holder of such portion.

6.7 Actions Under Credit Documents.

Following an Event of Default, the Administrative Agent shall at the request of the Majority Banks take any and all actions permitted under the other Credit Documents, including the Guaranty and the Security Documents.

6.8 Remedies Cumulative.

No right, power, or remedy conferred to the Administrative Agent or the Banks in this Agreement and the other Credit Documents, or now or hereafter existing at law, in equity, by statute, or otherwise, shall be exclusive, and each such right, power, or remedy shall to the full extent permitted by law be cumulative and in addition to every other such right, power, or remedy. No course of dealing and no delay in exercising any right, power, or remedy conferred to the Administrative Agent or the Banks in this Agreement and the other Credit Documents, or now or hereafter existing at law, in equity, by statute, or otherwise, shall operate as a waiver of or otherwise prejudice any such right, power, or remedy.

6.9 Application of Payments.

Prior to the Revolving Loan Maturity Date or any acceleration of the Credit Obligations, all payments made hereunder shall be applied to the Credit Obligations as directed by the Borrower, subject to the rules regarding the application of payments to certain Credit Obligations provided for hereunder and in the other Credit Documents. Following the Revolving Loan Maturity Date or any acceleration of the Credit Obligations, all payments and collections shall be applied to the Credit Obligations in the following order:

First, to the payment of the costs, expenses, reimbursements (other than reimbursement obligations with respect to draws under Letters of Credit), and indemnifications of the Administrative Agent that are due and payable under the Credit Documents;

Then, ratably to the payment of the costs, expenses, reimbursements (other than reimbursement obligations with respect to draws under Letters of Credit), and indemnifications of the Banks that are due and payable under the Credit Documents;

Then, ratably to the payment of all outstanding principal of all loans made hereunder, all reimbursement obligations for draws under Letters of Credit, all accrued but unpaid interest and fees under this Agreement and all obligations then due and payable under Interest Hedge Agreements;

Then, ratably to the payment of any other amounts due and owing with respect to the Credit Obligations; and

Finally, any surplus held by the Administrative Agent and remaining after payment in full of all the Credit Obligations and reserve for Credit Obligations not yet due and payable shall be promptly paid over to the Borrower or to whomever may be lawfully entitled to receive such surplus. All applications shall be distributed in accordance with Section 2.11(a).

# ARTICLE 7 THE ADMINISTRATIVE AGENT

# 7.1 Appointment; Nature of Relationship.

Bank One, NA is hereby appointed by each of the Banks as its contractual representative hereunder and under each other Credit Document, and each of the Banks irrevocably authorizes the Administrative Agent to act as the contractual representative of such Bank with the rights and duties expressly set forth herein and in the other Credit Documents. The Administrative Agent agrees to act as such contractual representative upon the express conditions contained in this Article. Notwithstanding the use of the defined term "Administrative Agent," it is expressly understood and agreed that the Administrative Agent shall not have any fiduciary responsibilities to any Bank by reason of this Agreement or any other Credit Document and that the Administrative Agent is merely acting as the contractual representative of the Banks with only those duties as are expressly set forth in this Agreement and the other Credit Documents. In its capacity as the Banks' contractual representative, the Administrative Agent (i) does not hereby assume any fiduciary duties to any of the Banks, (ii) is a "representative" of the Banks within the meaning of the term "secured party" as defined in the Texas Uniform Commercial Code and (iii) is acting as an independent contractor, the rights and duties of which are limited to those expressly set forth in this Agreement and the other Credit Documents. Each of the Banks hereby

agrees to assert no claim against the Administrative Agent on any agency theory or any other theory of liability for breach of fiduciary duty, all of which claims each Bank hereby waives.

# 7.2 Powers.

The Administrative Agent shall have and may exercise such powers under the Credit Documents as are specifically delegated to the Administrative Agent by the terms of each thereof, together with such powers as are reasonably incidental thereto. The Administrative Agent shall have no implied duties to the Banks, or any obligation to the Banks to take any action thereunder except any action specifically provided by the Credit Documents to be taken by the Administrative Agent.

## 7.3 General Immunity.

Neither the Administrative Agent nor any of its directors, officers, agents or employees shall be liable to the Borrower, the Banks or any Bank for any action taken or omitted to be taken by it or them hereunder or under any other Credit Document or in connection herewith or therewith except to the extent such action or inaction is determined in a final non-appealable judgment by a court of competent jurisdiction to have arisen from the gross negligence or willful misconduct of such Person (IT BEING THE INTENTION OF THIS PROVISION THAT THERE SHALL BE NO LIABILITY ARISING SOLELY OUT OF OR BY REASON OF THE NEGLIGENCE OF ANY OF THE PARTIES BEING REIMBURSED).

7.4 No Responsibility for Loans, Recitals, etc.

Neither the Administrative Agent nor any of its directors, officers, agents or employees shall be responsible for or have any duty to ascertain, inquire into, or verify (a) any statement, warranty or representation made in connection with any Credit Document or any borrowing hereunder; (b) the performance or observance of any of the covenants or agreements of any obligor under any Credit Document, including, without limitation, any agreement by an obligor to furnish information directly to each Bank; (c) the satisfaction of any condition specified in Article 3, except receipt of items required to be delivered solely to the Administrative Agent; (d) the existence or possible existence of any Default or Event of Default; (e) the validity, enforceability, effectiveness, sufficiency or genuineness of any Credit Document or any other instrument or writing furnished in connection therewith; (f) the value, sufficiency, creation, perfection or priority of any Lien in any Collateral; or (q) the financial condition of the Borrower or any guarantor of any of the Credit Obligations or of any of the Borrower's or any such quarantor's respective Subsidiaries. The Administrative Agent shall have no duty to disclose to the Banks information that is not required to be furnished by the Borrower to the Administrative Agent at such time, but is voluntarily furnished by the Borrower to the Administrative Agent (either in its capacity as Administrative Agent or in its individual capacity).

7.5 Acting on Instructions of Banks.

The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder and under any other Credit Document in accordance with written instructions

signed by the Majority Banks, and such instructions and any action taken or failure to act pursuant thereto shall be binding on all of the Banks. The Banks hereby acknowledge that the Administrative Agent shall be under no duty to take any discretionary action permitted to be taken by it pursuant to the provisions of this Agreement or any other Credit Document unless it shall be requested in writing to do so by the Majority Banks. The Administrative Agent shall be fully justified in failing or refusing to take any action hereunder and under any other Credit Document unless it shall first be indemnified to its satisfaction by the Banks pro rata against any and all liability, cost and expense that it may incur by reason of taking or continuing to take any such action.

7.6 Employment of Agents and Counsel.

The Administrative Agent may execute any of its duties as Administrative Agent hereunder and under any other Credit Document by or through employees, agents,

and attorneys in fact and shall not be answerable to the Banks, except as to money or securities received by it or its authorized agents, for the default or misconduct of any such agents or attorneys in fact selected by it with reasonable care. The Administrative Agent shall be entitled to advice of counsel concerning the contractual arrangement between the Administrative Agent and the Banks and all matters pertaining to the Administrative Agent's duties hereunder and under any other Credit Document.

# 7.7 Reliance on Documents; Counsel.

The Administrative Agent shall be entitled to rely upon any notice, consent, certificate, affidavit, letter, telegram, facsimile, telex, electronic mail message, statement, paper or document believed by it to be genuine and correct and to have been signed or sent by the proper person or persons, and, in respect to legal matters, upon the opinion of counsel selected by the Administrative Agent, which counsel may be employees of the Administrative Agent. For purposes of determining compliance with the conditions specified in Article 3, each Bank that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Bank unless the Administrative Agent shall have received notice from such Bank prior to the applicable date specifying its objection thereto.

7.8 Administrative Agent's Reimbursement and Indemnification.

Each Bank agrees to reimburse and indemnify the Administrative Agent for such Bank's ratable share of (i) any amounts not reimbursed by the Borrower for which the Administrative Agent is entitled to reimbursement by the Borrower under the Credit Documents, (ii) any other expenses incurred by the Administrative Agent on behalf of the Banks, in connection with the preparation, execution, delivery, administration and enforcement of the Credit Documents (including, without limitation, for any expenses incurred by the Administrative Agent in connection with any dispute between the Administrative Agent and any Bank or between two or more of the Banks) and (iii) any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever which may be imposed on, incurred by or asserted against the Administrative Agent in any way relating to or arising out of the Credit

Documents or any other document delivered in connection therewith or the transactions contemplated thereby (including, without limitation, for any such amounts incurred by or asserted against the Administrative Agent in connection with any dispute between the Administrative Agent and any Bank or between two or more of the Banks), or the enforcement of any of the terms of the Credit Documents or of any such other documents, provided that (i) no Bank shall be liable for any of the foregoing to the extent any of the foregoing is found in a final non-appealable judgment by a court of competent jurisdiction to have resulted from the gross negligence or willful misconduct of the Administrative Agent (IT BEING THE INTENTION OF THIS PROVISION THAT SUCH INDEMNIFICATION OBLIGATION WILL BE APPLICABLE REGARDLESS OF WHETHER CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE OF ANY OF THE PARTIES BEING INDEMNIFIED) and (ii) any

indemnification required pursuant to Section 2.11(b) shall, notwithstanding the provisions of this Section, be paid by the relevant Bank in accordance with the provisions thereof. The obligations of the Banks under this Section shall survive payment of the Credit Obligations and termination of this Agreement.

#### 7.9 Notice of Default.

The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder unless the Administrative Agent has received written notice from a Bank or the Borrower referring to this Agreement describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that the Administrative Agent receives such a notice, the Administrative Agent shall give prompt notice thereof to the Banks.

# 7.10 Rights as a Bank.

In the event the Administrative Agent is a Bank, the Administrative Agent shall have the same rights and powers hereunder and under any other Credit Document with respect to its Commitment and its Loan Advances as any Bank and may exercise the same as though it were not the Administrative Agent, and the term "Bank" or "Banks" shall, at any time when the Administrative Agent is a Bank, unless the context otherwise indicates, include the Administrative Agent in its individual capacity. The Administrative Agent and its Affiliates may accept deposits from, lend money to, and generally engage in any kind of trust, debt, equity or other transaction, in addition to those contemplated by this Agreement or any other Credit Document, with the Borrower or any of its Subsidiaries in which the Borrower or such Subsidiary is not restricted hereby from engaging with any other Person. The Administrative Agent, in its individual capacity, is not obligated to remain a Bank.

#### 7.11 Bank Credit Decision.

Each Bank acknowledges that it has, independently and without reliance upon the Administrative Agent, the Arranger or any other Bank and based on the financial statements prepared by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement and the other Credit Documents. Each Bank also acknowledges that it will, independently and without reliance upon the Administrative

Agent, the Arranger or any other Bank and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement and the other Credit Documents.

# 7.12 Successor Administrative Agent.

The Administrative Agent may resign at any time after a Default has occurred and remained uncured for 60 days by giving written notice thereof to the Banks and the Borrower, such resignation to be effective upon the appointment of a successor Administrative Agent or, if no successor Administrative Agent has been

appointed, forty-five days after the retiring Administrative Agent gives notice of its intention to resign. The Administrative Agent may be removed at any time with or without cause by written notice received by the Administrative Agent from the Majority Banks, such removal to be effective on the date specified by the Majority Banks. Upon any such resignation or removal, the Majority Banks shall have the right to appoint, on behalf of the Borrower and the Banks, a successor Administrative Agent. If no successor Administrative Agent shall have been so appointed by the Majority Banks within thirty days after the resigning Administrative Agent's giving notice of its intention to resign, then the resigning Administrative Agent may appoint, on behalf of the Borrower and the Banks, a successor Administrative Agent. Notwithstanding the previous sentence, the Administrative Agent may at any time without the consent of the Borrower or any Bank, appoint any of its Affiliates which is a commercial bank as a successor Administrative Agent hereunder. If the Administrative Agent has resigned or been removed and no successor Administrative Agent has been appointed, the Banks may perform all the duties of the Administrative Agent hereunder and the Borrower shall make all payments in respect of the Credit Obligations to the applicable Bank and for all other purposes shall deal directly with the Banks. No successor Administrative Agent shall be deemed to be appointed hereunder until such successor Administrative Agent has accepted the appointment. Any such successor Administrative Agent shall be a commercial bank having capital and retained earnings of at least \$100,000,000. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the resigning or removed Administrative Agent. Upon the effectiveness of the resignation or removal of the Administrative Agent, the resigning or removed Administrative Agent shall be discharged from its duties and obligations hereunder and under the Credit Documents. After the effectiveness of the resignation or removal of an Administrative Agent, the provisions of this Article shall continue in effect for the benefit of such Administrative Agent in respect of any actions taken or omitted to be taken by it while it was acting as the Administrative Agent hereunder and under the other Credit Documents. In the event that there is a successor to the Administrative Agent by merger, or the Administrative Agent assigns its duties and obligations to an Affiliate pursuant to this Section, then the term "Prime Rate" as used in this Agreement shall mean the prime rate, base rate or other analogous rate of the new Administrative Agent.

7.13 Administrative Agent and Arranger Fees.

The Borrower agrees to pay to the Administrative Agent and the Arranger, for their respective accounts, the fees agreed to by the Borrower, the Administrative Agent and the Arranger pursuant to the Fee Letter or as otherwise agreed from time to time.

## 7.14 Delegation to Affiliates.

The Borrower and the Banks agree that the Administrative Agent may delegate any of its duties under this Agreement to any of its Affiliates. Any such Affiliate (and such Affiliate's directors, officers, agents and employees) which performs

duties in connection with this Agreement shall be entitled to the same benefits of the indemnification, waiver and other protective provisions to which the Administrative Agent is entitled under this Agreement.

## 7.15 Execution of Collateral Documents.

The Banks hereby empower and authorize the Administrative Agent to execute and deliver to the Borrower on their behalf the Security Documents and all related agreements, documents or instruments as shall be necessary or appropriate to effect the purposes of the Security Documents.

# 7.16 Collateral Releases.

The Banks hereby empower and authorize the Administrative Agent to execute and deliver to the Restricted Entities on behalf of the Banks any agreements, documents or instruments as shall be necessary or appropriate to effect any releases of Collateral which shall be permitted by the terms hereof or of any other Credit Document or which shall otherwise have been approved by the Majority Banks (or, if required by the terms of this Agreement, all of the Banks) in writing.

#### 7.17 Syndication, Managing and Documentation Agents.

None of the Syndication, Managing and Documentation Agents shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than those applicable to all Banks as such. Without limiting the foregoing, none of the Syndication, Managing and Documentation Agents shall have or be deemed to have a fiduciary relationship with any Bank. Each Bank hereby makes the same acknowledgments with respect to such Banks as it makes with respect to the Administrative Agent in Section 7.11.

# ARTICLE 8 BENEFIT OF AGREEMENT; ASSIGNMENT; PARTICIPATIONS

8.1 Successors and Assigns.

The terms and provisions of the Credit Documents shall be binding upon and inure to the benefit of the Borrower and the Banks and their respective successors and assigns permitted hereby, except that (i) the Borrower shall not have the right to assign its rights or obligations under the Credit Documents without the prior written consent of each Bank, (ii) any assignment by any Bank must be made in compliance with Section 8.3, and (iii) any transfer by participation must

be made in compliance with Section 8.2. Any attempted assignment or transfer by any party not made in compliance with this Section 8.1 shall be null and void, unless such attempted assignment or transfer is treated as a participation in accordance with Section 8.3(b). The parties to this Agreement acknowledge that clause (ii) of this Section 8.1 relates only to absolute assignments and this Section 8.1 does not prohibit assignments creating security interests, including, without limitation, any pledge or assignment by any Bank of all or any portion of its rights under this Agreement and any note to a Federal Reserve Bank; provided, however, that no such pledge or assignment creating a security interest shall release the transferor Bank from its obligations hereunder unless and until the parties thereto have complied with the provisions of Section 8.3. The Administrative Agent may treat the Person which made any Loan Advance or which holds any note as the owner thereof for all purposes hereof unless and until such Person complies with Section 8.3; provided, however, that the Administrative Agent may in its discretion (but shall not be required to) follow instructions from the Person which made any Loan Advance or which holds any note to direct payments relating to such Loan Advance or note to another Person. Any assignee of the rights to any Loan Advance or any note agrees by acceptance of such assignment to be bound by all the terms and provisions of the Credit Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the owner of the rights to any Loan Advance (whether or not a note has been issued in evidence thereof), shall be conclusive and binding on any subsequent holder or assignee of the rights to such Loan Advance.

#### 8.2 Participations.

a. Any Bank may at any time sell to one or more banks or other entities ("Participants") participating interests in any Loan Advance owing to such Bank, any note held by such Bank, any Commitment of such Bank or any other interest of such Bank under the Credit Documents. In the event of any such sale by a Bank of participating interests to a participant, such Bank's obligations under the Credit Documents shall remain unchanged, such Bank shall remain solely responsible to the other parties hereto for the performance of such obligations, such Bank shall remain the owner of its Loan Advances and the holder of any note issued to it in evidence thereof for all purposes under the Credit Documents, all amounts payable by the Borrower under this Agreement shall be determined as if such Bank had not sold such participating interests, and the Borrower and the Administrative Agent shall continue to deal solely and directly with such Bank in connection with such Bank's rights and obligations under the Credit Documents.

Each Bank shall retain the sole right to approve, without the consent of any participant, any amendment, modification or waiver of any provision of the Credit Documents other than any amendment, modification or waiver with respect to any Loan Advance or Commitment in which such participant has an interest which would require consent of all of the Banks pursuant to the terms of Section 9.3 or of any other Credit Document.

b. The Borrower agrees that each participant shall be deemed to have the right of setoff provided in Section 6.6 in respect of its participating interest in amounts owing under the Credit Documents to the same extent as if the amount of its participating interest were owing directly to it as a Bank under the Credit Documents, provided that each Bank shall retain the right of setoff provided in Section 6.6 with respect to the amount of participating interests sold to each participant. The Banks agree to share with each participant, and each participant, by exercising

the right of setoff provided in Section 6.6, agrees to share with each Bank, any amount received pursuant to the exercise of its right of setoff, such amounts to

be shared in accordance with Section 6.2 as if each participant were a Bank. The Borrower further agrees that each participant shall be entitled to the benefits of Sections 2.7, 2.9 and 2.12 to the same extent as if it were a Bank and had acquired its interest by assignment pursuant to Section 8.3, provided that (i) a participant shall not be entitled to receive any greater payment under Sections 2.7, 2.9 and 2.12 than the Bank who sold the participating interest to such Participant would have received had it retained such interest for its own account, unless the sale of such interest to such participant is made with the prior written consent of the Borrower, and (ii) any participant not incorporated under the laws of the United States of America or any State thereof agrees to comply with the provisions of Section 2.17 to the same extent as if it were a Bank.

# 8.3 Assignments.

a. Any Bank may at any time assign to one or more banks or other entities ("Purchasers") all or any part of its rights and obligations under the Credit Documents. Such assignment shall be pursuant to an Assignment and Acceptance or in such other form as may be agreed to by the parties thereto. Each such assignment with respect to a Purchaser which is not a Bank or an Affiliate of a Bank or an Approved Fund shall either be in an amount equal to the entire applicable Commitment and Loan Advances of the assigning Bank or (unless each of the Borrower and the Administrative Agent otherwise consents) be in an aggregate amount (of Revolving Commitment (or, if the Revolving Commitment has been terminated, outstanding Revolving Loan Advances and Letter of Credit Exposure) and/or Term Loan Advances) not less than \$5,000,000, and if the assignee Bank retains any interest in the rights and obligations under the Credit Documents, it shall retain an interest (measured on the same basis) of at least \$5,000,000. The amount of the assignment shall be based on the Commitment or outstanding Loan Advances (if the Commitment has been terminated) subject to the assignment, determined as of the date of such assignment or as of the "Trade Date," if the "Trade Date" is specified in the assignment.

b. The consent of the Borrower shall be required prior to an assignment becoming effective unless the Purchaser is a Bank or an Affiliate of a Bank or an Approved Fund, provided that the consent of the Borrower shall not be required if a Default has occurred and is continuing. The consent of the Administrative Agent and the Issuing Bank shall be required prior to an assignment becoming effective unless the Purchaser is a Bank. Any consent required under this Section 8.3(b) shall not be unreasonably withheld or delayed.

c. Upon (i) delivery to the Administrative Agent of an Assignment and Assumption Agreement, together with any consents required by Sections 8.2(a) and 8.2(b), and (ii) payment of a \$3,500 fee to the Administrative Agent for processing such assignment (unless such fee is waived by the Administrative Agent), such assignment shall become effective on the effective date specified in such assignment. The assignment shall contain a representation by the Purchaser to the effect that none of the consideration used to make the purchase of the Commitment and Loans under the applicable assignment agreement constitutes "plan assets" as defined under ERISA and that the rights and interests of the Purchaser in and under the Credit Documents will not be "plan assets" under ERISA. On and after the effective date of such

assignment, such Purchaser shall for all purposes be a Bank party to this Agreement and any other Credit Document executed by or on behalf of the Banks and shall have all the rights and obligations of a Bank under the Credit Documents, to the same extent as if it were an original party thereto, and the transferor Bank shall be released with respect to the Commitment and Loan Advances assigned to such Purchaser without any further consent or action by the Borrower, the Banks or the Administrative Agent. In the case of an assignment covering all of the assigning Bank's rights and obligations under this Agreement, such Bank shall cease to be a Bank hereunder but shall continue to be entitled to the benefits of, and subject to, those provisions of this Agreement and the other Credit Documents which survive payment of the Credit Obligations and termination of the applicable agreement. Any assignment or transfer by a Bank of rights or obligations under this Agreement that does not comply with this Section 8.3 shall be treated for purposes of this Agreement as a sale by such Bank of a participation in such rights and obligations in accordance with Section 8.2. Upon the consummation of any assignment to a Purchaser pursuant to this Section 8.3(c) the transferor Bank, the Administrative Agent and the Borrower shall, if the transferor Bank or the Purchaser desires that its Loans be evidenced by notes, make appropriate arrangements so that new notes or, as appropriate, replacement notes are issued to such transferor Bank and new notes or, as appropriate, replacement notes, are issued to such Purchaser, in each case in principal amounts reflecting their respective Commitments, as adjusted pursuant to such assignment.

d. The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain at one of its offices in Houston, Texas a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Banks, and the Commitments of, and principal amounts of the Loans owing to, each Bank pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Banks may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Bank hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower at any reasonable time and from time to time upon reasonable prior notice.

### 8.4 Dissemination of Information.

The Borrower authorizes each Bank to disclose to any participant or Purchaser or any other Person acquiring an interest in the Credit Documents by operation of law (each a "Transferee") and any prospective Transferee any and all information in such Bank's possession concerning the creditworthiness of the Borrower and its Subsidiaries.

Tax Treatment. If any interest in any Credit Document is transferred to any Transferee which is not incorporated under the laws of the United States or any State thereof, the transferor Bank shall cause such Transferee, concurrently with the effectiveness of such transfer, to comply with the provisions of Section 2.12. The Borrower shall pay on demand of the applicable party specified herein (a) all reasonable out of pocket costs and expenses of the Administrative Agent and the Issuing Bank in connection with the preparation, execution, delivery, administration, modification, and amendment of this Agreement and the other Credit Documents, including the reasonable fees and expenses of outside counsel for the Administrative Agent and the Issuing Bank, and (b) all out-of-pocket costs and expenses of the Administrative Agent, the Issuing Bank, and each Bank in connection with the preservation or enforcement of their respective rights under this Agreement and the other Credit Documents, legal proceedings, or otherwise, including fees and expenses of counsel for the Administrative Agent, the Issuing Bank, and each Bank. The provisions of this Section shall survive the repayment and termination of the credit provided for under this Agreement and any purported termination of this Agreement which does not expressly refer to this Section.

## 9.2 Indemnification.

The Borrower agrees to protect, defend, indemnify, and hold harmless the Administrative Agent, the Issuing Bank, each Bank, and each of their respective Related Parties (for the purposes of this Section 8.2, collectively, the "Indemnified Parties"), from and against all demands, claims, actions, suits, damages, judgments, fines, penalties, liabilities, and out of pocket costs and expenses, including reasonable costs of attorneys and related costs of experts such as accountants (collectively, the "Indemnified Liabilities"), actually incurred by any Indemnified Party which are related to any litigation or proceeding relating to this Agreement, the other Credit Documents, or the transactions contemplated thereunder, INCLUDING ANY INDEMNIFIED LIABILITIES CAUSED BY ANY INDEMNIFIED PARTY'S OWN NEGLIGENCE, but not Indemnified Liabilities which are a result of any Indemnified Party's gross negligence or willful misconduct. The provisions of this Section shall survive the repayment and termination of the credit provided for under this Agreement and any purported termination of this Agreement which does not expressly refer to this Section.

### 9.3 Modifications, Waivers, and Consents.

No provision of this Agreement, the Term Loan Notes or the Revolving Loan Notes may be waived, amended, or modified, nor shall any consent required under this Agreement, the Term Loan Notes or the Revolving Loan Notes be effective, except pursuant to an agreement or agreements in writing entered into by the Borrower and the Majority Banks or by the Borrower and the Administrative Agent with the consent of the Majority Banks; provided that no such agreement shall (a) waive any of the conditions specified in Section 3.1 or 3.2 without the written consent of the Administrative Agent, all the Banks, and the Borrower, (b) increase the Commitment of any Bank without the written consent of such Bank, (c) forgive or reduce the amount or rate of any principal, interest, or fees payable under the Credit Documents, or postpone or extend the time for payment thereof without the written consent of the Banks, all the Banks, and the Borrower, (d) extend the scheduled maturities or times of payment without the written consent of the Administrative Agent, all the Banks, and the Borrower, (e) release any Guarantor or all or substantially all of the Collateral (except as

otherwise permitted or required as provided in the definition of "Asset Sale" or Section 5.9(b)) without the written consent of the Administrative Agent, all the Banks, and the Borrower, or (f) change the percentage of Banks required to take any action under this Agreement, the Term Loan Notes, the Revolving Loan Notes, or the Security Documents, including any amendment of the definition of "Majority Banks" or this Section 9.3 without the written consent of the Administrative Agent, all the Banks, and the Borrower. No modification, waiver, or consent shall, unless in writing and signed by the Administrative Agent, the Issuing Bank or the Swing Line Lender affect the rights or obligations of the Administrative Agent, the Issuing Bank or the Swing Line Lender, as the case may be, under the Credit Documents. The Administrative Agent shall not modify or waive or grant any consent under any other Credit Document if such action would be prohibited under this Section 9.3 with respect to the Credit Agreement, the Term Loan Notes or the Revolving Loan Notes.

#### 9.4 Survival of Agreements.

All representations, warranties, and covenants of the Borrower in this Agreement and the other Credit Documents shall survive the execution of this Agreement and the other Credit Documents and any other document or agreement.

## 9.5 Notice.

All notices and other communications under this Agreement, the Term Loan Notes and the Revolving Loan Notes shall be in writing and mailed by certified mail (return receipt requested), telecopied, telexed, hand delivered, or delivered by a nationally recognized overnight courier, to the address for the appropriate party specified in Schedule I-B or at such other address as shall be designated by such party in a written notice to the other parties. Mailed notices shall be effective when received. Telecopied or telexed notices shall be effective when transmission is completed or confirmed by telex answerback. Delivered notices shall be effective when delivered by messenger or courier. Notwithstanding the foregoing, notices and communications to the Administrative Agent pursuant to Article 2 or 7 shall not be effective until received by the Administrative Agent.

# 9.6 CHOICE OF LAW.

THIS AGREEMENT AND THE RIGHTS AND OBLIGATIONS OF THE PARTIES UNDER THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, APPLICABLE FEDERAL LAW AND THE LAW OF THE STATE OF TEXAS.

# 9.7 Submission To Jurisdiction; Waivers.

The Borrower hereby irrevocably and unconditionally:

a. Submits for itself and its property in any legal action or proceeding relating to this Agreement and the other Credit Documents to which it is a party, or for recognition and enforcement of any judgment in respect thereof, to the non-exclusive general jurisdiction of the courts of the State of Texas, the courts of the United States for the Southern District of Texas, and appellate courts from any thereof;

b. consents that any such action or proceeding may be brought in such courts and waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same;

c. agrees that service of process in any such action or proceeding may be effected by mailing a copy thereof by registered or certified mail (or any substantially similar form of mail), postage prepaid, to the Borrower, as the case may be at its address set forth in Schedule I-B or at such other address of which the Administrative Agent shall have been notified pursuant thereto; agrees that nothing herein shall affect the right to effect service of process in any other manner permitted by law or shall limit the right to sue in any other jurisdiction; and waives, to the maximum extent not prohibited by law, any right it may have to claim or recover in any legal action or proceeding referred to in this Section any special, exemplary, punitive or consequential damages.

9.8 WAIVER OF JURY TRIAL.

THE BORROWER IRREVOCABLY WAIVES ANY RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATING TO THE CREDIT DOCUMENTS OR ANY TRANSACTIONS RELATING THERETO.

### 9.9 Counterparts.

This Agreement may be executed in multiple counterparts, and by the parties hereto on separate counterparts which together shall constitute one and the same instrument.

9.10 No Further Agreements.

THIS WRITTEN AGREEMENT AND THE CREDIT DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

9.11 2001 Credit Agreement.

a. The Revolving Loan Notes, the Term Loan Notes and the Swing Line Note have been given in renewal, extension and modification, and not in extinguishment, of the credit facilities previously provided to the Borrower pursuant to the 2001 Credit Agreement.

b. From and after the Closing Date, (i) the Prior Banks (other than any Bank

party hereto, and then only to the extent set forth in this Agreement) shall have no further commitment or obligation to the Borrower as a Bank under this Agreement and (ii) the Prior Administrative Agent shall cease to be the Administrative Agent and shall have no further obligation to the Borrower, the Prior Banks or the Banks in the capacity of "Administrative Agent" under the 2001 Credit Agreement or this Agreement; provided that (x) the Prior Administrative Agent shall continue to enjoy all of its rights and remedies (including but not limited to indemnities) as set forth in the 2001 Credit Agreement with respect to events occurring before the Closing Date and each of the Prior Banks shall remain obligated to the Prior Administrative Agent with respect to such rights and remedies with respect to such events, all as if the 2001 Credit Agreement had not been amended and restated by this Agreement but had instead been terminated effective as of the Closing Date and (y) nothing in this Section 9.11(b) shall affect any claim, cause of action or other remedy which any Restricted Entity may have against any Prior Bank, the Prior Administrative Agent or any other Person under the 2001 Credit Agreement.

c. No letter of credit (each, a "Prior Letter of Credit") issued pursuant to the 2001 Credit Agreement shall be deemed a Letter of Credit, the bank (the "Prior Issuing Bank") issuing such letters of credit shall not be deemed to be an Issuing Bank and neither the Prior Letters of Credit or the Prior Issuing Bank shall be entitled to any status under this Agreement.

d. The Prior Administrative Agent, in its representative capacity and as a Prior Bank, has executed and delivered that certain Assignment of Debt and Liens and Resignation of Administrative Agent (the "Assignment") of even date herewith; to the extent necessary or appropriate, the Borrower hereby consents to the Assignment.

e. This Agreement is an amendment and restatement of the 2001 Credit Agreement.

EXECUTED as of the date first above written.

BORROWER:

INTEGRATED ELECTRICAL SERVICES, INC.

Ву:	
Name:	
Title:	

ADMINISTRATIVE AGENT:

BANK ONE, NA, as Administrative Agent

By:	 	
Name:		
Title:		

BANKS:

Copyright © 2012 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

BANK ONE, NA
Bv
By: Name:
Title:
U.S. BANK NATIONAL ASSOCIATION
By:
Name:
Title:
BANK OF SCOTLAND
By:
Name:
Title:
LASALLE BANK NATIONAL ASSOCIATION
By:
Name:
Title:
WELLS FARGO BANK, N.A.
By:
Name:
Title:
SOUTHWEST BANK OF TEXAS, N.A.
SOUTHWEST DANK OF TEARS, N.A.
Ву:
Name:
Title:
FIRST AMERICAN BANK, SSB HOUSTON
Dv7 •
By:
Name:
Title:
ALLIED IRISH BANK, P.L.C.
By:
Name:
Title:

REGIONS BANK

Зу:	
Name:	
Title:	

FIRST BANK & TRUST

Ву:		 	
Name:			
Title:			

HIBERNIA NATIONAL BANK

By:	 	
Namo .		

Name:			
Title:			

RZB FINANCE LLC

By:	
Name:	
Title:	

#### CERTIFICATION

- I, Herbert R. Allen, certify that:
  - I have reviewed this quarterly report on Form 10-Q of Integrated Electrical Services, Inc.;
  - 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
  - 3. Based on my knowledge, the financial statements and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
  - 4. I am responsible for establishing and maintaining disclosure controls and procedures (as defined in the Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and we have:
    - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
    - b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
    - c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
    - d) disclosed in this report any change in the registrant's internal

control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an quarterly report) that has materially affected, or is reasonable likely to materially affect, the registrant's internal control over financial reporting; and;

- 5. I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: April 29, 2004

/s/ Herbert R. Allen

Herbert R. Allen Chief Executive Officer and Interim Chief Financial Officer CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with this Quarterly Report of Integrated Electrical Services, Inc. (the "Company") on Form 10-Q for the period ending March 31, 2004 (the "Report"), I, Herbert R. Allen, Chief Executive Officer and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. ss. 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and result of operations of the Company.

/s/ Herbert R. Allen

Herbert R. Allen Chief Executive Officer and Interim Chief Financial Officer April 29, 2004

\_\_\_\_\_